

RECEIVED-FPSC

BEFORE THE FLORIDA PUBLIC SERVICE COMMISSION

In re: Petition by Chesapeake Utilities Corporation for Approval to Issue Common Stock and Secured and/or Unsecured Debt and to Exceed Limitation Placed on Short-Term Borrowings in 1999 Docket No. 981213-GU

PETITION BY CHESAPEAKE UTILITIES CORPORATION FOR MODIFICATION OF AUTHORITY TO ISSUE COMMON STOCK DURING THE TWELVE MONTHS ENDED DECEMBER 31, 1999

)

))

Chesapeake Utilities Corporation (Chesapeake, the Company or Petitioner) respectfully files this Petition, pursuant to Section 366.04 (1), Florida Statutes, seeking a modification in its authority to issue common stock during the twelve months ended December 31, 1999. By Order No. PSC-98-1678-FOF-GU, issued December 10, 1998, the FPSC granted Chesapeake's petition and authorized Chesapeake to issue, among other securities, 45,082 shares of its common stock pursuant to the Company's Retirement Savings Plan ("RSP") in 1999. In this instant Petition, Chesapeake is now seeking FPSC approval to issue up to 50,000 additional shares of its common stock for said RSP in 1999.

- 1. Name and principal business offices of Petitioner: IVED & FI
 - (a)
- Chesapeake Utilities Corporation P.O. Box 615 909 Silver Lake Boulevard Dover, Delaware 19904
- Chesapeake Utilities Corporation Florida Division P.O. Box 960 1015 6th Street N.W. Winter Haven, Florida 33881
- (c) Chesapeake Utilities Corporation Florida Division 1514 Alexander Street, Suite 107 Plant City, Florida 33566

FAU OF RECORDS

DOCUMENT NUMBER-DATE

07731 JUN 25 8

FPSC-RECORDS/REPORTING



and

- (d) Chesapeake Utilities Corporation Florida Division
 1639 West Gulf to Lake Highway Lecanto, Florida 33461
- 2. Incorporated:

- m. 🖻

Chesapeake Utilities Corporation - Incorporated under the Laws of the State of Delaware on November 12, 1947 and qualified to do business in Florida, Maryland, and Pennsylvania.

3. Person authorized to receive notices and communications in this respect:

Wayne L. Schiefelbein P.O. Box 13688 Tallahassee, Florida 32317 (850) 297-9890

Attomey for Chesapeake Utilities Corporation

4. <u>Major Changes in Financial Position:</u>

There have been no major changes in Chesapeake's financial position since the filing of its petition in this docket on September 23, 1998 for the 1999 fiscal year. Except for the modification requested in this Petition, the common stock, debt and short-term borrowing requirements of the Company have been met pursuant to the conditions and limitations authorized by the FPSC in Order No. PSC-98-1678-FOF-GU, issued December 10, 1998.

5. <u>Proposed transaction:</u>

Chesapeake is proposing to issue up to 50,000 additional shares of Chesapeake common stock for the Company's RSP in 1999. The additional 50,000 shares, for which approval is hereby requested, are in addition to the 45,082 shares of common stock approved by the FPSC in Order No. PSC-98-1678-FOF-GU, issued December 10, 1998.

6. <u>Purpose for which Securities are to be issued:</u>

Chesapeake's RSP was implemented on February 1, 1977. As of March 31, 1999, the RSP had 417 participants; a total market valuation of \$12,491,403; and 237,869 shares of the Company's common stock. True and correct copies of the current RSP Plan Document and Adoption Agreement are filed with the FPSC herewith as Exhibits A and B, respectively. Pursuant to the RSP, all employer matching contributions made on behalf of participants are invested in common stock of Chesapeake. The RSP was amended at the end of 1998 to provide for a larger employer matching amount, and at the same time the Company's Pension Plan was closed off to new employees. Accordingly, as the employer matching amount has increased, so has the number of shares being issued under the RSP. In addition, the number of participating employees has increased as a result of two acquisitions completed in 1998.

On June 23, 1992, the Delaware Public Service Commission by Order No. 3425 approved the issuance of 100,000 new shares of common stock for the purpose of administering the RSP. Since January 5, 1996, the RSP's trustee has issued 94,362 of these shares to fund employer matching contributions. To continue to balance the composition of capital between debt and equity, Chesapeake wants to maintain flexibility in how the RSP is funded, i.e., with new shares of its stock, buying shares on the open market, and/or a combination of both funding methods. On June 23, 1999, Chesapeake filed an application with the Delaware Public Service Commission seeking approval to issue up to 100,000 additional shares of common stock for the purpose of administering Chesapeake's RSP. Approval of such application is anticipated in July of 1999. As with Order No. 3245, the Delaware Public Service Commission's anticipated order would be "open-ended" in the sense that there is no time limit by which the approved securities would need to be issued. Chesapeake now seeks FPSC approval to issue 50,000 of these new shares of Chesapeake common stock for the purpose of administering Chesapeake's RSP during the remainder of 1999. The 45,082 shares approved for issuance by the FPSC in Order No. PSC-98-1678-FOF-GU, issued December 10, 1998, are not adequate to meet the funding requirements of the RSP for 1999.

7. Lawful object and purpose:

The common stock issued will be used for the purpose of administering Chesapeake's RSP. This is for a lawful object within the corporate purposes of Chesapeake and compatible with the public interest and is reasonably necessary or appropriate for such purposes.

8. <u>Counsel:</u>

The legality of the common stock issuance will be passed upon by William A. Denman, Esquire, Schmittinger & Rodriguez, 414 South State Street, P.O. Box 497, Dover, Delaware 19903, who will rely on Wayne L. Schiefelbein, Esquire, P.O. Box 13688, Tallahassee, Florida 32317, as to matters of Florida law.

9. Other Regulatory Agencies:

Under 26 Del. C Section 215 of the Delaware statutes, Chesapeake is regulated by the Delaware Public Service Commission and, therefore, must file a Prefiling Notice, a Notice, and an Application to obtain approval of the Delaware Commission before issuing new securities which mature more than one (1) year from the date of issuance. All necessary applications or registration statements have been or will be made as required and will be made a part of the final consummation report to the FPSC as required by Rule 25-8.009, Florida Administrative Code.

The address of the Delaware Commission is as follows:

Delaware Public Service Commission 861 Silver Lake Boulevard Cannon Building Dover, Delaware 19904 Attention: Bruce H. Burcat, Executive Director

10. <u>Control or ownership:</u>

Petitioner is not owned by any other company nor is Petitioner a member of any holding company system.

11. Exhibits:

. .

Filed herewith:

- Exhibit A: Chesapeake Utilities Corporation Retirement Savings Plan-Plan Document
- Exhibit B: Chesapeake Utilities Corporation Retirement Savings Plan-Adoption Agreement

12. Constitutionality of Statute:

The statutory requirement of FPSC approval of the issuance and sale of securities by a public utility, under Section 366.04 (1), Florida Statutes, as applied to Chesapeake, a Delaware corporation engaged in interstate commerce, is unconstitutional, in that it creates an unreasonable burden on interstate commerce. Support for this position is set out in Chesapeake's petition for declaratory statement disclaiming jurisdiction, as filed in FPSC Docket No. 930705-GU.

By FPSC Order No. PSC-93-1548-FOF-GU, issued on October 21, 1993, the FPSC denied the petition for declaratory statement, while approving the alternative petition for approval of the issuance of up to 100,000 new shares of common stock for the purpose of administering a Retirement Savings Plan. The FPSC found that "the facial constitutionality of a statute cannot be decided in an administrative proceeding," and that since the stock issuance was approved, "the question of constitutionality appears to be academic at this time."

Chesapeake continues to maintain that the assertion of jurisdiction by the FPSC over its securities unconstitutionally burdens interstate commerce, particularly where the Public Service Commission of the State of Delaware has approved their issuance and sale, and/or where the securities do not create a lien or encumbrance on assets of Chesapeake's public utility operations in the State of Florida.

. . *

Florida law provides for severe penalties for any willful violation of a statute administered by the FPSC or any of its rules or orders. Secs. 350.127 (1) and 366.095, Florida Statutes. Accordingly, Chesapeake believes it must submit to FPSC jurisdiction over its securities if it is to avoid assessment of such penalties and to otherwise remain in good standing before the FPSC. It therefore files the instant application, under protest, and without waiver of its position regarding the unconstitutionality of the statute. Based on the foregoing, Chesapeake Utilities Corporation requests that the FPSC issue an Order approving the proposed issuance of 50,000 additional shares of common stock for Chesapeake's Retirement Savings Plan.

Respectfully submitted,

Date: 1014 25,1999

5 · • 5

you've & Schiefellbein

Wayne L. Schiefelbein P.O. Box 13688 Tallahassee, Florida 32317 (850) 297-9890

Attorney for Chesapeake Utilities Corporation

whe By:

Michael P. McMasters Vice President, Treasurer, & CFO

Date: June 24, 1999

STATE OF DELAWARE * COUNTY OF KENT * SS

. . 1

BE IT REMEMBERED that on this <u>24th</u> day of June, 1999, personally appeared before me, a Notary Public for the State of Delaware, Michael P. McMasters, who being by me duly sworn, did depose and say that he is Vice President, Treasurer and CFO of Chesapeake Utilities Corporation, a Delaware corporation, and that insofar as the application of Chesapeake Utilities Corporation states facts, and insofar as those facts are within his personal knowledge, they are true; and insofar as those facts that are not within his personal knowledge, he believes them to be true, and that the exhibits accompanying this application and attached hereto are true and correct copies of the originals of the aforesaid exhibits, and that he has executed this application on behalf of the Company and pursuant to the authorization of its Board of Directors.

NPN. N. K.

Michael P. McMasters Vice President, Treasurer & CFO

SWORN TO AND SUBSCRIBED before me the day and year first above written.

uidi W. Wattlin

Notary Public My Commission Expires: 10/21/0 (

EXHIBIT A

- · --

RSP PLAN DOCUMENT

4/16/27

Chesapeake Utilities Corporation Retirement Savings Plan

/-/-

Plan Document

••

·...

January, 1999

TABLE OF CONTENTS

PARAGRAPH

.

. .

ARTICLE I DEFINITIONS

1.1	Actual Contribution Percentage	1
1.2	Actual Deferral Percentage	1
1.3	Adoption Agreement	2
1.4	Aggregate Limit	2
1.5	Allocation Date(s)	2
1.6	Annual Additions	2
1.7	Annuity Starting Date	2
1.8	Applicable Calendar Year	2
1.9	Applicable Life Expectancy	2
1.10	Average Contribution Percentage (ACP)	3 3
1.11	Average Deferral Percentage (ADP)	3
1.12	Break in Service	3
1.13	Code	3 3
1.14	Compensation	
1.15	Defined Benefit Plan	5
1.16	Defined Benefit (Plan) Fraction	5
1.17	Defined Contribution Dollar Limitation	5
1.18	Defined Contribution Plan	5
1.19	Defined Contribution (Plan) Fraction	5
1.20	Designated Beneficiary	6
1.21	Direct Rollover	6
1.22	Disability	6
1.23	Distribution Calendar Year	6
1.24	Early Retirement Age	6
1.25	Early Retirement Date	6
1.26	Earned Income	6
1.27	Effective Date	6
1.28	Election Period	6
1.29	Elapsed Time	6
1.30	Elective Deferral	6
1.31 • •	Eligible Participant	7
1.32	Eligible Retirement Plan	7
1.33	Eligible Rollover Distribution	7
1.34	Employee	7
1.35	Employee Contractor	7
1.36	Employer	8
1.37	Entry Date	8
1.38	ERISA	8
1.39	Excess Aggregate Contributions	8
1.40	Excess Amount	8
1.41	Excess Contribution	8
1.42	Excess Elective Deferrals	8
1.43	First Distribution Calendar Year	8
1.44	Hardship	8
1.45	Highest Average Compensation	8
1.45	Highly Compensated Employee	8
		•

PAGE

1.47	Hour of Service	9.
1.48	Integration Level	10
1.49	Key Employee	10
1.50	Leased Employee	10
1.51	Limitation Year	10
1.52	Matching Contribution	10
1.53	Maximum Permissible Amount	11
1.54	Normal Retirement Age	11
1.55	Normal Retirement Date	11
1.56	Participant	11
1.57	Participant's Benefit	11
1.58	Participating Employee Contractor	11
1.59	Permissive Aggregation Group	11
1.60	Plan	11
1.61	Plan Administrator	11
1,62	Plan Year	11
1,63	Present Value	11
1.64	Prior Plan Year	11
1.65	Projected Annual Benefit	12
1.66	Qualified Domestic Relations Order	12
1.67	Qualified Early Retirement Age	12
1.68	Qualified Joint and Survivor Annuity	12
1.69	Qualified Matching Contributions	12
1.70	Qualified Non-Elective Contributions	12
1.71	Qualified Plan	12
1.72	Required Aggregation Group	12
1.73	Required Beginning Date	13
1.74	Rollover Contribution	13
1.75	Salary Deferral Agreement	13
1.76	Savings Incentive Match Plan for Employees	13
1.77	Self-Employed Individual	13
1.78	Service	13
1.79	Service Provider	13
1.80	Shareholder Employee	13
1.81	Simplified Employee Pension Plan	13
1.82	Spouse	13
1.83	Super Top-Heavy Plan	13
1.84	Top-Heavy Determination Date	13
1.85	Top-Heavy Plan	13
1.86	Top-Heavy Ration	14.
1.87	Top-Paid Group	15
1.88	Transfer Contribution	15
1.89	Trust	15
1.90	Trustee	15
1.91	USERRA	15
1.92	Valuation Date	15
1.93	Vested Account Balance	15
1.94	Welfare Benefit Fund	15
1.95	Year of Service	15

••

.*

. •

•

ARTICLE II ELIGIBILITY REQUIREMENTS

2.1	Participation	17
2.2	Change in Classification of Employment	17
2.3	Computation Period	17
2.4	Employment Rights	17
2.5	Service with Controlled Groups	17
2.6	Leased Employees	18
2.7	Waiver of Participation	18

1

د

ARTICLE III

EMPLOYER CONTRIBUTIONS

3.1	Contribution Amount	19
3.2	Expenses and Fees	19
3.3	Responsibility for Contributions	19
3.4	Return of Contributions	19
3.5	Merger of Assets from Another Plan	19

ARTICLE IV

EMPLOYEE CONTRIBUTIONS

4.1	Rollover Contributions	21
4.2	Elective Transfer Contributions	21
4.3	Trustee-to-Trustee Transfer Contribution	22
4.4	Elective Deferrals	22
4.5	Automatic Enrollment	22
4.6	Make-Up Contributions under USERRA	23

ARTICLE V

PARTICIPANT ACCOUNTS

5.1	Separate Accounts	24
5.2 ·	Allocation Date	24
5.3	Allocations to Participant Accounts	24
5.4	Allocating Employer Contributions	25
5.5	Allocating Investment and Losses	. 25
5.6	Allocation Adjustments	27
5.7	Participant Statements	27
5.8	Changes in Method of Valuing Participants' Accounts	27

ARTICLE VI

RETIREMENT BENEFITS AND DISTRIBUTIONS

6.1	Normal Retirement Benefits	28
6.2	Early Retirement Benefits	28
6.3	Benefits on Termination of Employment	28
6.4	Restrictions on Immediate Distributions	29

6.5	Normal and Optional Forms of Payment	30
6.6	Commencement of Benefits	30
6.7	Claims Procedures	31
6.8	In-Service Withdrawals	31
6.9	Hardship Withdrawals	32
6.10	Direct Rollover of Benefits	34
6.11	Participant's Notice	34
6.12	Assets Transferred from Money Purchase Pension Plans	34
6.13	Assets Transferred from a 401(k) Plan	34

7

э

ARTICLE VII DISTRIBUTION REQUIREMENTS

7.1	Joint and Survivor Annuity Requirements	35
7.2	Minimum Distribution Requirements	35
7.3	Limits on Distribution Periods	35
7.4	Required Distributions on or after the	
	Required Beginning Date	35
7.5	Required Beginning Date	36
7.6	Election Under Tax Equity and Fiscal Responsibility Act	
	Of 1982	36
7.7	Designation of Beneficiary	37
7.8	No Beneficiary	37
7.9	Distribution Beginning before Death	38
7.10	Distribution Beginning after Death	38
7.11	Distribution of Excess Elective Deferrals	38
7.12	Distributions of Excess Contributions	39
7.13	Distribution of Excess Aggregate Contributions	39
7.14	Distributions to Minors and Individuals	
	who are Legally Incompetent	40
7.15	Unclaimed Benefits	40

• •

ARTICLE VIII JOINT AND SURVIVOR ANNUITY REQUIREMENTS

81 ·	Applicability of Provisions	41
8.2	Payment of Qualified Joint and Survivor Annuity	41
8.3	Payment of Qualified Pre-Retirement Survivor Annuity	41
8.4	Qualified Election	41
8.5	Notice Requirements for Qualified	
	Joint and Survivor Annuity	42
8.6	Notice Requirements for Qualified	
	Pre-Retirement Survivor Annuity	42
8.7	Automatic Joint and Survivor Annuity and	
	Early Survivor Annuity	42
8.8	Annuity Contracts	43
8.6 8.7	Joint and Survivor Annuity Notice Requirements for Qualified Pre-Retirement Survivor Annuity Automatic Joint and Survivor Annuity and Early Survivor Annuity	42

ARTICLE IX VESTING

ş

.

. •

. .

9.1	Employee Contributions	44
9.2	Employer Contributions	44
9.3	Computation Period	44
9.4	Regualification Prior to Five Consecutive	
	One-Year Breaks In Service	44
9.5	Regualification after Five Consecutive	
	One-Year Breaks in Service	44
9.6	Calculating Vested Interest	44
9.7	Forfeitures	44
9.8	Amendment of Vesting Schedule	48
9.9	Service with Controlled Groups	48
9.10	Compliance with Uniformed Service Employment and	
	Reemployment Rights Act of 1994	48

ARTICLE X

LIMITATIONS ON ALLOCATIONS

10.1	Participation in this Plan Only	46
10.2	Disposition of Excess Annual Additions	46
10.3	Participation in Multiple Defined Contribution Plans	46
10.4	Disposition of Excess Annual Additions under Two Plans	47
10.5	Limitations on Benefits	47
10.6	Participation in this Plan and a Defined	
	Benefit Plan	47

ARTICLE XI

ANTIDISCRIMINATION TESTING

11.1		General Testing Requirements	48
11.2	<u>.</u> .	Testing Limitations	48
11.3		Calculation and Distribution of Excess Contributions	
		and Excess Aggregate Contributions	49
11.4 - [.]		Qualified Non-Elective and/or Matching Contributions	49
11.5		Special Rules Relating to Application of the ADP Test	50
11.6		Recharacterization	50
11.7		Special Rules Relating to Application of the ACP Test	51

ARTICLE XII

ADMINISTRATION

12.1	Plan Administrator	53
12.2	Тпизтее	53
12.3	Administrative Fees and Expenses	54
12.4	Limitation on Liability and Indemnification	55
12.5	Persons Serving as Plan Administrator	57

ARTICLE XIII TRUST FUND/CUSTODIAL ACCOUNT

13.1	The Trust	58
13.2	Control of Plan Assets	58
13.3	Exclusive Benefit Rules	58
13.4	Assignment and Alienation of Benefits	58
13.5	Determination of Qualified Domestic	
	Relations Order (QDRO)	58

.)ŧ

. •

1

ARTICLE XIV INVESTMENTS

14.1	Fiduciary Standards	60
14.2	Funding Arrangement	60
14.3	Investment Alternatives of the Trustee	60
14.4	Participant Loans	61
14.5	Insurance Policies	63
14.6	Allocation of Investment Responsibility	64
14.7	Prohibited Transactions	66
14.8	Participant Investment Direction	66

ARTICLE XV

TOP-HEAVY PROVISIONS

15.1	Applicability of Rules	68
15.2	Minimum Contribution	68
15.3	Minimum Vesting	68
15.4	Limitations on Allocations	69

ARTICLE XVI AMENDMENT AND TERMINATION

16.1	Amendment	70
16.2	Protected Benefits	70
16.3 -	Plan Termination	70
16.4	Distribution Restrictions	70
16.5	Mergers and Consolidations	70
16.6	Resignation and Removal	71

ARTICLE XVII GOVERNING LAW

17.1	Governing Law	
------	---------------	--

72

Chesapeake Utilities Corporation Retirement Savings Plan

ARTICLE I

DEFINITIONS

1.1 Actual Contribution Percentage The ratio (expressed as a percentage and calculated separately for each Participant) of:

- (a) the Participant's Contribution Percentage Amounts [as defined at (c)-(f)] for the Current Plan Year, or for non-Highly Compensated Employees the Prior Plan Year, unless otherwise elected in the Adoption Agreement, to
- (b) the Participant's Compensation for such Plan Year. Unless otherwise specified in the Adoption Agreement, Compensation will only include amounts for the period during which the Employee was eligible to participate.

Contribution amounts on behalf of any Participant shall include:

- (c) the amount of Matching Contributions and Qualified Matching Contributions (to the extent not taken into account for purposes of the ADP test) made under the Plan on behalf of the Participant for the Plan Year,
- (d) forfeitures of Excess Aggregate Contributions or Matching Contributions allocated to the Participant's account which shall be taken into account in the year in which such forfeiture is allocated,
- (e) at the election of the Employer, Qualified Non-Elective Contributions, and
- (f) the Employer may elect to use Elective Deferrals in the Contribution Percentage Amounts as long as the ADP test is met before the Elective Deferrals are used in the ACP test and continues to be met following the exclusion of those Elective Deferrals that are used to meet the ACP test.

Contribution amounts shall not include Matching Contributions, whether or not Qualified, that are forfeited either to correct Excess Aggregate Contributions, or because the contributions to which they relate are Excess Deferrals, Excess Contributions, or Excess Aggregate Contributions.

1.2 Actual Deferral Percentage The ratio (expressed as a percentage and calculated separately for each Participant) of:

- (a) the amount of Employer contributions actually contributed to the Trust on behalf of each Participant for the Current Plan Year or for non-Highly Compensated Employees for the Prior Plan Year, unless otherwise specified in the Adoption Agreement, to
- (b) the Participant's Compensation for such Plan Year. Unless otherwise specified in the Adoption Agreement, Compensation will only include amounts received for the period during which the Employee was eligible to participate.

Employer contributions on behalf of any Participant shall include:

(c) any Elective Deferrals made pursuant to the Participant's deferral election, including Excess Elective Deferrals, but excluding Elective Deferrals that are either taken into account in the Contribution Percentage test (provided the ADP test is satisfied both with and without exclusion of these Elective Deferrals) or are returned as excess Annual Additions, and

(d) at the election of the Employer, Qualified Non-Elective Contributions and Qualified Matching Contributions.

For purposes of computing Actual Deferral Percentages, an eligible Employee who fails to make Elective Deferrals shall be treated as a Participant on whose behalf no Elective Deferrals are made.

1.3 Adoption Agreement The document attached to this Plan by which an Employer elects the terms and conditions of a qualified retirement plan and trust/custodial account under this Plan and Trust/Custodial Account.

1.4 Aggregate Limit The sum of:

- (a) 125% of the greater of the ADP of the non-Highly Compensated Employees or the ACP of non-Highly Compensated Employees under the Plan subject to Code Section 401(m) for the Current Plan Year or the Prior Plan Year, as elected by the Employer in the Adoption Agreement, and
- (b) the lesser of 200% or two percent plus the lesser of such ADP or ACP.

For purposes of determining the ADP and ACP for the non-Highly Compensated Employees, the Prior Plan Year will be used in calculating the Aggregate Limit, unless the Employer has elected in the Adoption Agreement to use the Current Plan Year.

Alternatively, the Aggregate Limit can be determined by substituting "the lesser of 200% or two percent plus" for "125% of" in (a) above, and substituting "125% of" for "the lesser of 200% or two percent plus" in (b) above.

1.5 Allocation Date(s) The date or dates on which Participant recordkeeping accounts are adjusted to reflect account activity including but not limited to contributions, loans, distributions, Hardship withdrawals as well as earnings activity including but not limited to income, capital gains or market fluctuations in accordance with Article V hereof. Unless specified otherwise in the Adoption Agreement, all allocations for a particular Plan Year will be made as of the Valuation Date of that Plan Year.

1.6 Annual Additions The sum of the following amounts credited to a Participant's account for the Limitation Year:

- (a) Employer contributions
- (b) Employee contributions
- (c) forfeitures

Employee and Employer make-up contributions under USERRA received during the current Limitation Year shall be treated as Annual Additions with respect to the Limitation Year to which the make-up contributions are attributable. Excess Amounts applied in a Limitation Year to reduce Employer contributions will be considered Annual Additions for such Limitation Year, pursuant to the provisions of Article X.

1.7 Annuity Starting Date The first day of the first period for which an amount is paid as an annuity or in any other form.

1.8 Applicable Calendar Year The First Distribution Calendar Year, and in the event of the recalculation of life expectancy, such succeeding calendar year. If payments commence in accordance with paragraph 7.4(d) before the Required Beginning Date, the Applicable Calendar Year is the year such payments commence. If distribution is in the form of an immediate annuity purchased after the Participant's death with the Participant's remaining interest, the Applicable Calendar Year is the year of purchase.

1.9 Applicable Life Expectancy The life expectancy or joint and last survivor expectancy calculated using the attained age of the Participant or Designated Beneficiary as of the Participant's or Designated Beneficiary's birthday in the Applicable Calendar Year, reduced by one for each calendar year which has elapsed since the date life expectancy was first calculated. If life expectancy is being recalculated, the Applicable Life Expectancy shall be the life expectancy as so recalculated. The life expectancy of a non-Spouse Beneficiary may not be recalculated.

1.10 Average Contribution Percentage (ACP) The average of the Contribution Percentages for each Highly Compensated Employee determined for the Current Plan Year and for each non-Highly Compensated Employee determined for the Prior Plan Year, unless the Current Plan Year is selected in the Adoption Agreement.

1.11 Average Deferral Percentage (ADP) The average of the Actual Deferral Percentages for each Highly Compensated Employee determined for the Current Plan Year and for each non-Highly Compensated Employee determined for the Prior Plan Year, unless the Current Plan Year is selected in the Adoption Agreement.

1.12 Break in Service A 12-consecutive month period during which an Employee fails to complete more than 500 Hours of Service, if the Hours of Service method is used to determine a Year of Service or a period of severance of at least 12 consecutive months if the Elapsed Time Method is being used to determine a Year of Service.

1.13 Code The Internal Revenue Code of 1986, including any amendments thereto.

1.14 *Compensation* The Employer may select one of the following three safe-harbor definitions of Compensation in the Adoption Agreement. Compensation will also include Compensation by the Employer through another individual or entity under the provisions of Code Sections 3121 and 3306.

- (a) Code Section 3401(a) Wages All remuneration received by an Employee for services performed for the Employer which are subject to Federal income tax withholding at the source. Unless elected otherwise in the Adoption Agreement, Compensation shall include any amount deferred under a salary reduction agreement which is not includible in the gross income of a Participant under Code Section 125 in connection with a cafeteria plan, Code Section 402(e)(3) in connection with a cash or deferred plan, Code Section 402(h)(1)(B) in connection with a Simplified Employee Pension Plan, and Code Section 403(b) in connection with a tax-sheltered annuity plan. Wages are determined without regard to any rules that limit the remuneration included in wages based on the nature or location of the employment or the services performed [such as the exception for agricultural labor in Code Section 3401(a)(2)].
- (b) Code Sections 6041, 6051 and 6052 Reportable Wages All remuneration received by an Employee for services performed for the Employer which are required to be reported on Form W-2. Unless otherwise elected in the Adoption Agreement, Compensation shall include any amount deferred under a salary reduction agreement which is not includible in the gross income of a Participant under Code Section 125 in connection with a cafeteria plan, Code Section 402(e)(3) in connection with a cash or deferred plan, Code Section 402(h)(1)(B) in connection with a Simplified Employee Pension Plan, and Code Section 403(b) in connection with a tax-sheltered annuity plan. A Participant's wages includes remuneration defined at subparagraph (a) above and all other remuneration paid to an Employee by the Employer (in the course of the Employer's trade or business) for which the Employer is required to furnish the Employee a written statement under Code Sections 6041(d), 6051(a)(3) and 6052. Such amount must be determined without regard to any rules that limit the remuneration included in wages based on the nature or location of the employment or the services performed [such as the exception for agricultural labor in Code Section 3401(a)(2)].
- (c) Code Section 415 Compensation A Participant's Earned Income, wages, salaries, and fees for professional services and other amounts received, without regard to whether or not an amount is paid in cash, for personal services actually rendered in the course of employment with the Employer maintaining the Plan. Compensation includes, but is not limited to, commissions paid salesmen, Compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, bonuses, fringe benefits and reimbursements or other expense allowances under a nonaccountable plan [as described in Regulation Section 1.62-2(c)]. Compensation excludes the following:

- (1) for Plan Years beginning before January 1, 1998, Employer contributions, made under the terms of a salary deferral agreement between an Employee and the Employer, to a plan of deferred compensation which are not includible in the Employee's gross income for the taxable year in which contributed. Such contributions shall include any amount deferred under Code Section 125 in connection with a cafeteria plan, Code Section 402(e)(3) in connection with a cash or deferred plan, Code Section 402(h)(1)(B) in connection with a Simplified Employee Pension Plan, Code Section 457 Plan and Code Section 403(b) in connection with a tax-sheltered annuity plan;
- (2) distributions received from a plan of deferred compensation;
- (3) amounts realized from the exercise of a non-qualified stock option, or when restricted stock (or property) held by the Employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;
- (4) amounts realized from the sale, exchange or other disposition of stock acquired under a qualified stock option; and
- (5) amounts deferred by an Employee under the terms of a non-qualified deferred compensation plan;

Exclusions from Compensation A Participant's Compensation shall be determined in accordance with paragraph (a), (b) or (c) above and shall not exclude any item of income unless provided in the basic definition or elected by the Employer in the Adoption Agreement.

Annual Additions and Top-Heavy Rules For purposes of Article X and XV, Compensation shall be Code Section 415 Compensation as described in paragraph 1.14(c). For Plan Years beginning before January 1, 1998, Compensation excludes amounts deferred under a plan of deferred Compensation as described at paragraph 1.14(c)(1). For Plan Years beginning after December 31, 1997, Compensation includes amounts deferred under a plan of deferred compensation includes amounts deferred under a plan of deferred compensation as described at paragraph 1.14(c)(1). For Plan Years beginning after December 31, 1997, Compensation includes amounts deferred under a plan of deferred compensation as described at paragraph 1.14(c)(1). For purposes of applying the limitations of Article IX, Compensation for a Limitation Year is the Compensation actually paid or made available during such Limitation Year. Notwithstanding the preceding sentence, Compensation with respect to a Participant in a Defined Contribution Plan who is permanently and totally disabled [as defined in Code Section 22(e)(3)] is the Compensation such Participant would have received for the Limitation Year if the Participant had been paid at the rate of Compensation paid immediately before becoming permanently and totally disabled; for Limitation Years beginning before January 1, 1997, but not for Limitation Years beginning after December 31, 1996, such imputed Compensation for the disabled Participant may be taken into account only if the Participant is not a Highly Compensated Employee (defined at paragraph 1.46 of the Plan) and contributions made on behalf of such Participant are nonforfeitable when made.

Highly Compensated and Key Employees For purposes of paragraphs 1.46 and 1.49, Compensation shall be Code Section 415 Compensation as described in paragraph 1.14(c). Such definition shall include any amount deferred under Code Section 125 in connection with a cafeteria plan, Code Section 402(e)(3) in connection with a cash or deferred plan, Code Section 402(h)(1)(B) in connection with a Simplified Employee Pension Plan, Code Section 402(k) in connection with a Savings Incentive Match Plan for Employees and Code Section 403(b) in connection with a tax-sheltered annuity plan. The Employer if elected in the Adoption Agreement, may limit Compensation considered for purposes of the Plan.

Computation Period The Plan Year while eligible to participate shall be the computation period for purposes of determining a Participant's Compensation, unless the Employer selects a different computation period in the Adoption Agreement.

Limitation on Compensation The annual Compensation of each Participant which may be taken into account for determining all benefits provided under the Plan for any year shall not exceed the limitation as imposed by Code Section 401(a)(17), as adjusted under Code Section 401(a)(17)(B). If a Plan has a Plan Year that contains fewer than 12 calendar months, the annual Compensation limit for that period is an amount equal to the limitation as

imposed by Code Section 401(a)(17) as adjusted for the calendar year in which the Compensation period begins, multiplied by a fraction, the numerator of which is the number of full months in the short Plan Year and the denominator of which is 12.

USERRA For purposes of Employee and Employer make-up contributions, Compensation during the period of military service shall be deemed to be the Compensation the Employee would have received during such period if the Employee were not in qualified military service, based on the rate of pay the Employee would have received from the Employer but for the absence due to military leave. If the Compensation the Employee would have received during the leave is not reasonably certain, Compensation will be equal to the Employee's average Compensation from the Employer during the 12-month period immediately preceding the military leave or, if shorter, the Employee's actual period of employment with the Employer.

1.15 **Defined Benefit Plan** A plan under which a Participant's benefit is determined by a formula contained in the plan and no individual accounts are maintained for Participants.

1.16 Defined Benefit (Plan) Fraction For Limitation Years beginning before January 1, 2000, a fraction, the numerator of which is the sum of the Participant's Projected Annual Benefits under all the Defined Benefit Plans (whether or not terminated) maintained by the Employer, and the denominator of which is the lesser of 125% of the dollar limitation determined for the Limitation Year under Code Sections 415(b) and (d) or 140% of the Highest Average Compensation, including any adjustments under Code Section 415(b).

Transitional Rule If an Employee was a Participant as of the first day of the first Limitation Year beginning after 1986, in one or more Defined Benefit Plans maintained by the Employer which were in existence on May 6, 1986, the denominator of this fraction will not be less than 125% of the sum of the annual benefits under such plans which the Participant had accrued as of the close of the last Limitation Year beginning before 1987, disregarding any changes in the terms and conditions of the plan after May 5, 1986. The preceding sentence applies only if the Defined Benefit Plans individually and in the aggregate satisfied the requirements of Code Section 415 for all Limitation Years beginning before 1987.

1.17 Defined Contribution Dollar Limitation Thirty thousand dollars (\$30,000) as adjusted by the Secretary of the Treasury for increases in the cost-of-living. Such increases will be in multiples of five thousand dollars (\$5,000).

1.18 **Defined Contribution Plan** A plan under which individual accounts are maintained for each Participant to which all contributions, forfeitures, investment income and gains or losses, and expenses are credited or deducted. A Participant's benefit under such plan is based solely on the fair market value of his or her account balance.

1.19 Defined Contribution (Plan) Fraction For Limitation Years beginning before January 1, 2000, a fraction, the numerator of which is the sum of the Annual Additions to the Participant's account under all the Defined Contribution Plans (whether or not terminated) maintained by the Employer for the current and all prior Limitation Years (including the Annual Additions attributable to the Participant's nondeductible Employee contributions to all Defined Benefit Plans, whether or not terminated, maintained by the Employer, and the Annual Additions attributable to all Welfare Benefit Funds as defined in paragraph 1.105, individual medical accounts as defined in Code Section 415(1)(2) and Simplified Employee Pension Plans as defined in paragraph 1.89, maintained by the Employer), and the denominator of which is the sum of the maximum aggregate amounts for the current and all prior Limitation Plan was maintained by the Employer). The maximum aggregate amount in the Limitation Year is the lesser of 125% of the dollar limitation determined under Code Sections 415(b) and (d) in effect under Code Section 415(c)(1)(A) or 35% of the Participant's Compensation for such year.

Transitional Rule If an Employee was a Participant as of the end of the first day of the first Limitation Year beginning after 1986, in one or more Defined Contribution Plans maintained by the Employer which were in existence on May 6, 1986, the numerator of this fraction will be adjusted if the sum of this fraction and the Defined Benefit Fraction would otherwise exceed 1.0 under the terms of this Plan. Under the adjustment, an amount equal to the product of the excess of the sum of the fractions over 1.0 multiplied by the denominator of this fraction, will be permanently subtracted from the numerator of this fraction. The adjustment is calculated using the fractions as

they would be computed as of the end of the last Limitation Year beginning before 1987, and disregarding any changes in the terms and conditions of the Plan made after May 6, 1986, but using the Code Section 415 limitation applicable to the first Limitation Year beginning on or after January 1, 1987. The Annual Addition for any Limitation Year beginning before 1987, shall not be re-computed to treat all Employee contributions as Annual Additions.

1.20 Designated Beneficiary The individual, trust or other entity who is designated as the beneficiary of a Participant's account under the Plan in accordance with Code Section 401(a)(9) and the regulations thereunder.

1.21 Direct Rollover A payment by the Plan to an Eligible Retirement Plan specified by the Participant.

1.22 Disability An illness or injury of a potentially permanent nature, expected to last for a continuous period of not less than 12 months, certified by a physician selected by or satisfactory to the Employer, which prevents the Employee from engaging in any occupation for wage or profit for which the Employee is reasonably fit by training, education or experience.

1.23 Distribution Calendar Year A calendar year for which a minimum distribution is required.

1.24 *Early Retirement Age* The age set by the Employer in the Adoption Agreement (but not less than 55), which is the earliest age at which a Participant may retire and receive his or her benefits under the Plan.

1.25 *Early Retirement Date* The date on which a Participant satisfies the age and Service requirements specified in the Adoption Agreements.

1.26 *Earned Income* Net earnings from self-employment in the trade or business with respect to which the Plan is established, determined without regard to items not included in gross income and the deductions allocable to such items, provided that personal services of the individual are a material income-producing factor. Earned Income shall be reduced by contributions made by an Employer to a qualified plan to the extent deductible under Code Section 404. Net earnings shall be determined taking into account the deduction for one-half of self-employment taxes allowed to the taxpayer under Code Section 164(f), to the extent deductible.

1.27 *Effective Date* The date on which the Employer's Plan or amendment to such Plan becomes effective. For amendments reflecting statutory and regulatory changes contained in the Uruguay Round Agreements Act of the General Agreement on Tariffs and Trade ("GATT"), the Uniform Services Employment and Reemployment Rights Act of 1994 (USERRA) effective December 12, 1996, the Small Business Job Protection Act of 1996 (SBJPA), the Taxpayer Relief Act of 1997 (TRA'97), the Effective Date will be the earlier of the date upon which such amendment is first administratively applied or the first day of the Plan Year following the date of adoption of such amendment.

1.28 *Election Period* The period which begins on the first day of the Plan Year in which the Participant attains age 35 and ends on the date of the Participant's death. If a Participant separates from Service prior to the first day of the Plan Year in which age 35 is attained, the Election Period shall begin on the date of separation, with respect to the account balance as of the date of separation.

1.29 *Elapsed Time* A method of determining an Employee's entitlement under the Plan with respect to eligibility to participate, as well as vesting which is not based on the Employee's completion of a specified number of Hours of Service during a consecutive twelve month period, but rather with reference to the total period of time which elapses during which the Employee is employed by the Employer maintaining the Plan.

1.30 Elective Deferral Employer contributions made to the Plan at the election of the Participant, in lieu of cash Compensation pursuant to a Salary Deferral Agreement or other deferral mechanism. With respect to any taxable year, a Participant's Elective Deferral is the sum of all Employer contributions made on behalf of such Participant pursuant to an election to defer under any qualified cash or deferred arrangement as described in Code Section 401(k), any Simplified Employee Pension Plan with a cash or deferred arrangement as described in Code Section 402(h)(1)(B), any SIMPLE as described in Code Section 401(k)(11) or Individual Retirement Account arrangement as described in Code Sections 401(k)(11)(A) or 408(p), any eligible deferred compensation plan under Code Section 457, any plan as described under Code Section 501(c)(18), and any Employer contributions made on behalf of a Participant for the purchase of an annuity contract under Code Section 403(b) pursuant to a Salary Deferral Agreement. Elective Deferrals shall not include any deferrals properly distributed as excess Annual Additions.

1.31 Eligible Participant An Employee who is eligible to make a Voluntary Contribution or an Elective Deferral (if the Employer takes such contributions into account in the calculation of the Contribution Percentage), or to receive a Matching Contribution (including forfeitures) or a Qualified Matching Contribution. If an Employee contribution or Elective Deferral is required as a condition of participation in the Plan, any Employee who would be a Participant in the Plan if such Employee made such a contribution shall be treated as an Eligible Participant even though no Employee contributions or Elective Deferrals are made.

1.32 Eligible Retirement Plan An individual retirement account (IRA) as described in Code Section 408(a), an individual retirement annuity (IRA) as described in Code Section 408(b), an annuity plan as described in Code Section 403(a), or a qualified trust as described in Code Section 401(a), which accepts Eligible Rollover Distributions. However, in the case of an Eligible Rollover Distribution to a surviving Spouse, an Eligible Retirement Plan is an individual retirement account or individual retirement annuity.

1.33 Eligible Rollover Distribution An Eligible Rollover Distribution is any distribution of all or any portion of the balance to the credit of the Participant except that an Eligible Rollover Distribution does not include:

- (a) any distribution that is one of a series of substantially equal periodic payments made not less frequently than annually for the life (or life expectancy) of the Participant or the joint lives (or joint life expectancies) of the Participant and the Participant's Designated Beneficiary, or for a specified period of ten years or more,
- (b) any distribution to the extent such distribution is required under Code Section 401(a)(9),
- (c) the portion of any distribution that would not be includible in gross income if paid to the Participant (determined without regard to the exclusion for net unrealized appreciation with respect to Employer securities), and
- (d) Hardship Withdrawals under Code Section 401(k)(2)(B)(i)(IV) or excess amounts which are returned to a Participant in accordance with paragraph 10.2.

1.34 Employee A person employed by the Employer (including Self-Employed Individuals and partners), all Employees of a member of an affiliated service group [as defined in Code Section 414(m)], all Employees of a controlled group of corporations [as defined in Code Section 414(b)], all Employees of any incorporated or unincorporated trade or business which is under common control [as defined in Code Section 414(c)], leased Employees [as defined in Code Section 414(n)], and any Employee required to be aggregated by Code Section 414(o). All such Employees shall be treated as employed by a single Employer. No Employee Contractor shall be considered an Employee, however, for purposes of administering and construing this Plan, any reference to Employee will also mean an Employee Contractor, with the exception of any references to Employer Matching, Discretionary, Qualified non-Elective, and Qualified Matching Contributions being made on any Employee's behalf.

Leased Employees shall not be Employees for purposes of participation in the Plan. The exclusion is only available if Leased Employees do not constitute more than 20% of the recipient Employer's non-highly compensated work force, and the Employer complies with the requirements as outlined in paragraph 2.6, and so elects in the Adoption Agreement.

1.35 *Employee Contractor* An Employee hired for a specific job or jobs to be supervised directly or indirectly by the Employer. Hours worked and duration of employment are at the discretion of the Employer. The Employee Contractor will work on an "as needed" basis and is not eligible for any employee benefits offered by the Employer unless specified at the time of employment. The status of an Employee Contractor is subject to change at the discretion of the Employer.

1.36 Employer Chesapeake Utilities Corporation or another entity that succeeds the Employer and adopts this Plan and all members of a controlled group of corporations [as defined in Code Section 414(b) as modified by Code Section 415(h)], all commonly controlled trades or businesses [as defined in Code Section 414(c) as modified by Code Section 415(h)] or affiliated service groups [as defined in Code Section 414(m)] of which the adopting Employer is a part, and any other entity required to be aggregated with the Employer pursuant to regulations under Code Section 414(o).

1.37 *Entry Date* The date as of which an Employee who has satisfied the Plan's eligibility requirements enters or reenters the Plan, as defined in the Adoption Agreement.

- 1.38 ERISA The Employee Retirement Income Security Act of 1974, as amended.
- 1.39 Excess Aggregate Contribution The excess, with respect to any Plan Year, of:
 - (a) the aggregate Contribution Percentage Amounts taken into account in computing the numerator of the Contribution Percentage actually made on behalf of Highly Compensated Employees for such Plan Year, over
 - (b) the maximum Contribution Percentage Amounts permitted by the ACP test determined in accordance with Article XI hereof.

Such determination shall be made after first determining Excess Elective Deferrals pursuant to paragraph 1.42 and then determining Excess Contributions pursuant to paragraph 1.41.

1.40 Excess Amount The excess of the Participant's Annual Additions for the Limitation Year over the Maximum Permissible Amount.

- 1.41 Excess Contribution With respect to any Plan Year, the excess of:
 - (a) the aggregate amount of Employer contributions actually taken into account in computing the ADP of Highly Compensated Employees for such Plan Year, over
 - (b) the maximum amount of such contributions permitted by the ADP test determined in accordance with Article XI hereof.

1.42 Excess Elective Deferrals Those Elective Deferrals that are includible in a Participant's gross income under Code Section 402(g) to the extent such Participant's Elective Deferrals for a taxable year exceed the dollar limitation under such Code Section. Excess Elective Deferrals shall be treated as Annual Additions under the Plan, unless such amounts are distributed no later than the first April 15th following the close of the Participant's taxable year.

1.43 First Distribution Calendar Year For distributions beginning before the Participant's death, the First Distribution Calendar Year is the calendar year immediately preceding the calendar year which contains the Participant's Required Beginning Date. For distributions beginning after the Participant's death, the First Distribution Calendar Year is the calendar year in which distributions are required to begin.

1.44 *Hardship* An immediate and heavy financial need of the Employee where such Employee lacks other available resources to satisfy such financial need.

1.45 Highest Average Compensation For Limitation Years beginning before January 1, 2000, the average Compensation for the three consecutive Years of Service with the Employer that produces the highest average. A Year of Service with the Employer is the 12-consecutive month period defined in the Adoption Agreement.

1.46 *Highly Compensated Employee* Effective for years after December 31, 1996, the term Highly Compensated Employee means any Employee who: (1) is a 5% owner at any time during the year or preceding year, or (2) for the preceding year had Compensation from the Employer in excess of \$80,000 and if the Employer

so elects in the Adoption Agreement, is in the Top-Paid Group for the preceding year. The \$80,000 amount is adjusted at the same time and in the same manner as under Code Section 415(d), except that the base period is the calendar quarter ending September 30, 1996.

For the determination of Highly Compensated Employee the applicable year of the Plan for which a determination is being made is called a determination year and the preceding 12-month period is called a look-back year.

A Highly Compensated former Employee is based on the rules applicable to determining Highly Compensated Employee status in effect for that determination year, in accordance with Section 1.414(q)-1T, A-4 of the temporary Income Tax Regulations and Notice 97-75.

In determining whether an Employee is a Highly Compensated Employee for years beginning in 1997, the amendments to Code Section 414(q) stated above are treated as having been in effect for years beginning in 1996. In order to be effective, a Top-Paid Group election or calendar year data election must apply consistently to all plans of the Employer that begin with or within the same calendar year.

1.47 Hour of Service

. .

- (a) Unless otherwise specified in the Adoption Agreement, each hour for which an Employee is paid, or entitled to payment, for the performance of duties for the Employer. These hours shall be credited to the Employee for the computation period in which the duties are performed, and
- (b) each hour for which an Employee is paid, or entitled to payment, by the Employer on account of a period of time during which no duties are performed (irrespective of whether the employment relationship has terminated) due to vacation, holiday, illness, incapacity (including disability), layoff, jury duty, military duty or leave of absence. No more than 501 Hours of Service shall be credited under this paragraph for any single continuous period (whether or not such period occurs in a single computation period). Hours under this paragraph shall be calculated and credited pursuant to Section 2530.200b-2 of the Department of Labor Regulations which are incorporated herein by this reference, and
- (c) each hour for which back pay, irrespective of mitigation of damages, is either awarded or agreed to by the Employer. The same Hours of Service shall not be credited both under paragraph (a) or paragraph (b), as the case may be, and under this paragraph (c). These hours shall be credited to the Employee for the computation period or periods to which the award or agreement pertains rather than the computation period in which the award, agreement or payment is made.
- (d) Hours of Service shall be credited for employment with the Employer and with other members of an affiliated service group [as defined in Code Section 414(m)], a controlled group of corporations [as defined in Code Section 414(b)], or a group of trades or businesses under common control [as defined in Code Section 414(c)] of which the adopting Employer is a member, and any other entity required to be aggregated with the Employer pursuant to Code Section 414(o) and the regulations thereunder. Hours of Service shall also be credited for any individual considered an Employee for purposes of this Plan under Code Section 414(n) or Code Section 414(o) and the regulations thereunder.
- (e) Solely for purposes of determining whether a Break in Service, as defined in paragraph 1.12, for participation and vesting purposes has occurred in a computation period, an individual who is absent from work for maternity or paternity reasons shall receive credit for the Hours of Service which would otherwise have been credited to such individual but for such absence, or in any case in which such hours cannot be determined, 8 Hours of Service per day of such absence. For purposes of this paragraph, an absence from work for maternity or paternity reasons means an absence by reason of the pregnancy of the individual, by reason of a birth of a child of the individual, by reason of such child by such individual, or for purposes of caring for such child for a period beginning immediately following such birth or placement. The Hours of Service credited under

this paragraph shall be credited in the computation period in which the absence begins if the crediting is necessary to prevent a Break in Service in that period, or in all other cases, in the following computation period. No more than 501 hours will be credited under this paragraph.

(f) Hours of Service shall be determined under an Hours Counting Method or an Elapsed Time Method as selected by the Employer on the Adoption Agreement. If no selection is made actual Hours will be used.

1.48 Integration Level The amount of Compensation specified in the Adoption Agreement at or below which the rate of contributions or benefits (expressed in each case as a percentage of such Compensation) provided under the Plan is less than the rate of contributions or benefits (expressed in each case as a percentage of such Compensation) provided under the Plan with respect to Compensation above such level. The Adoption Agreement must specify an Integration Level in effect for the Plan Year for each Participant and no Integration Level in effect for a particular year may exceed the contribution and benefit base ("Taxable Wage Base") under Section 230 [Code Section 3121(a)(1)] of the Social Security Act in effect on the first day of the Plan Year. If the Plan uses Compensation based on the period of the Participant's participation, then the integration level must be prorated for a Participant's short Plan Year.

1.49 Key Employee Any Employee or former Employee (and the beneficiaries of such Employee) who at any time during the determination period was:

- (a) an officer of the Employer if such individual's annual Compensation exceeds 50% of the dollar limitation under Code Section 415(b)(1)(A) the defined benefit maximum annual benefit,
- (b) an owner or an individual considered an owner under Code Section 318 of one of the ten largest interests in the Employer if such individual's Compensation exceeds 100% of the dollar limitation under Code Section 415(c)(1)(A),
- (c) a more than 5% owner of the Employer, or
- (d) a one-percent owner of the Employer who has an annual Compensation of more than \$150,000.

The determination period is the Plan Year containing the Top-Heavy Determination Date and the four preceding Plan Years. The determination of Key Employee status will be made in accordance with Code Section 416(i)(1) and the regulations thereunder.

1.50 Leased Employee Effective December 12, 1994, any person (other than an Employee of the recipient) who, pursuant to an agreement between the recipient and any other person ("leasing organization"), has performed services for the recipient [or for the recipient and related persons determined in accordance with Code Section 414(n)(6)] on a substantially full-time basis for a period of at least one year and such services are performed under the primary direction or control of the recipient Employer. If a Leased Employee is treated as an Employee by reason of this paragraph 1.50, "Compensation" includes Compensation from the leasing organization which is attributable to services performed for the Employer.

1.51 Limitation Year The calendar year or such other 12-consecutive month period designated by the Employer in the Adoption Agreement for purposes of determining the maximum Annual Additions to a Participant's account. All qualified plans maintained by the Employer must use the same Limitation Year. If the Limitation Year is amended to a different 12-consecutive month period, the new Limitation Year must begin on a date within the Limitation Year in which the amendment is made. If no designation is made on the Adoption Agreement, the Limitation Year will automatically default to the Plan Year.

1.52 *Matching Contribution* An Employer contribution made to this or any other Defined Contribution Plan on behalf of a Participant on account of an Employee Voluntary Contribution made by such Participant, or on account of a Participant's Elective Deferral or required after-tax contribution made by such Participant under a plan maintained by the Employer.

1.53 Maximum Permissible Amount The maximum Annual Additions that may be contributed or allocated to a Participant's account under the Plan for any Limitation Year shall not exceed the lesser of:

- (a) the Defined Contribution Dollar Limitation, or
- (b) 25% of the Participant's Compensation for the Limitation Year.

The Compensation limitation referred to in (b) shall not apply to any contribution for medical benefits [within the meaning of Code Section 401(h) or Code Section 419A(f)(2)] which is otherwise treated as an Annual Addition under Code Section 415(1)(1) or 419(d)(2). If a short Limitation Year is created because of an amendment changing the Limitation Year to a different 12-consecutive month period, the Maximum Permissible Amount will not exceed the Defined Contribution Dollar Limitation multiplied by a fraction, the numerator of which is the number of months in the short Limitation Year and the denominator of which is 12.

1.54 Normal Retirement Age The age, set by the Employer in the Adoption Agreement, not to exceed age 65, at which a Participant may retire and is eligible to receive his or her benefits under the Plan.

1.55 Normal Retirement Date The date on which a Participant attains Normal Retirement Age.

1.56 **Participant** An Employee who has met the eligibility requirements and is participating in the Plan. For purposes of administering and construing this Plan, any reference to Participant will also mean a Participating Employee Contractor, with the exception of any references to Employer Matching, Discretionary, Qualified Non-Elective, and Qualified matching Contributions being made on any Participant's behalf.

1.57 Participant's Benefit With respect to required distributions pursuant to paragraph 7.4, the account balance as of the last Valuation Date in the calendar year immediately preceding the Distribution Calendar Year increased by the amount of any contributions or forfeitures allocated to the account balance as of the dates in the calendar year after the Valuation Date and decreased by distributions made in the calendar year after the Valuation Date. A special exception exists for the second Distribution Calendar Year. For purposes of this paragraph, if any portion of the minimum distribution for the First Distribution Calendar Year is made in the second Distribution Calendar Year on or before the Required Beginning Date, the amount of the minimum distribution made in the second Distribution Calendar Year shall be treated as if it had been made in the immediately preceding Distribution Calendar Year.

1.58 *Participating Employee Contractor* An Employee Contractor who has met the eligibility requirements of the Plan and has submitted a Salary Deferral Agreement.

1.59 *Permissive Aggregation Group* The Required Aggregation Group of plans plus any other plan or plans of the Employer which, when considered as a group with the Required Aggregation Group, would continue to satisfy the requirements of Code Sections 401(a)(4) and 410.

1.60 *Plan* The Employer's retirement plan as embodied herein and in the Adoption Agreement, as may be amended from time to time.

. 1.61 *Plan Administrator* The Employer or individual or entity appointed by the Employer to administer the Plan as provided at paragraph 12.1 hereof.

1.62 Plan Year The 12-consecutive month period designated by the Employer in the Adoption Agreement.

1.63 **Present Value** The actuarial equivalent of a Participant's accrued benefit under a Defined Benefit Plan maintained by the Employer expressed in the form of a lump sum. Actuarial equivalence shall be based on reasonable interest and mortality assumptions determined in accordance with the Top-Heavy provisions of the respective plan. Present Value is used for the purposes of the Top-Heavy test and the determination with respect thereto.

1.64 Prior Plan Year The Plan Year immediately preceding the Current Plan Year. If elected in the Adoption Agreement, the Average Deferral Percentage and Average Contribution Percentage determined for non-Highly Compensated Employees for the Prior Plan Year will be used in the antidiscrimination test for the Current Plan Year

to determine the maximum Average Deferral Percentage and Average Contribution Percentage for Highly Compensated Employees in the Current Plan Year.

1.65 **Projected Annual Benefit** For Limitation Years beginning before January 1, 2000, the annual retirement benefit (adjusted to an actuarial equivalent straight life annuity if such benefit is expressed in a form other than a straight life annuity or Qualified Joint and Survivor Annuity) to which the Participant would be entitled under the terms of a Defined Benefit Plan or plans, assuming:

- (a) the Participant will continue employment until Normal Retirement Age under the Plan (or current age, if later), and
- (b) the Participant's Compensation for the current Limitation Year and all other relevant factors used to determine benefits under the Plan will remain constant for all future Limitation Years.

1.66 Qualified Domestic Relations Order A Qualified Domestic Relations Order (QDRO) is a signed domestic relations order issued by a state court or agency which creates, recognizes or assigns to an alternate payee(s) the right to receive all or part of a Participant's plan benefit and which meets the requirements of Code Section 414(p). An alternate payee is a Spouse, former Spouse, child, or other dependent who is treated as a beneficiary under the Plan as a result of the QDRO. Unless elected otherwise by the Employer in the Adoption Agreement, the earliest date for payment of a QDRO to an alternate payee, is the date upon which the order is deemed qualified.

1.67 Qualified Early Retirement Age For purposes of paragraph 8.9, Qualified Early Retirement Age is the latest of:

- (a) the earliest date under the Plan on which the Participant may elect to receive retirement benefits, or
- (b) the first day of the 120th month beginning before the Participant reaches Normal Retirement Age, or
- (c) the date the Participant begins participation.

1.68 Qualified Joint and Survivor Annuity An immediate annuity for the life of the Participant with a survivor annuity for the life of the Participant's Spouse which is at least one-half of but not more than the amount of the annuity payable during the joint lives of the Participant and the Participant's Spouse. The exact amount of the survivor annuity is to be specified by the Employer in the Adoption Agreement. If not designated by the Employer, the survivor annuity.will be one-half of the amount paid to the Participant during his or her lifetime. The Qualified Joint and Survivor Annuity will be the amount of benefit which can be provided by the Participant's Vested Account Balance.

1.69 Qualified Matching Contributions ("QMACs") Matching Contributions which, when made are subject to the distribution and nonforfeitability requirements under Code Section 401(k).

1.70 Qualified Non-Elective Contributions ("QNECs") Contributions (other than Matching Contributions or Qualified Matching Contributions) made by the Employer and allocated to Participants' accounts that the Participants may not elect to receive in cash until distributed from the Plan, that are nonforfeitable when made, and that are distributable only in accordance with the distribution provisions that are applicable to Elective Deferrals and Qualified Matching Contributions.

1.71 Qualified Plan Any pension, profit-sharing, stock bonus, or other plan which meets the requirements of Code Section 401 and includes a trust exempt from tax under Code Section 501(a) or any annuity plan described in Code Section 403(a).

1.72 Required Aggregation Group A group of plans including:

- (a) each Qualified Plan of the Employer in which at least one Key Employee participates or participated at any time during the determination period (regardless of whether the plan has terminated), and
- (b) any other Qualified Plan of the Employer which enables a plan described in (a) to meet the requirements of Code Sections 401(a)(4) or 410.

1.73 *Required Beginning Date* The date on which a Participant is required to take his or her first minimum distribution under the Plan. The rules are set forth at paragraph 7.5.

1.74 Rollover Contribution A contribution made by a Participant of an amount distributed to such Participant from another Qualified Plan in accordance with Code Sections 402(a)(5), (6), and (7).

1.75 Salary Deferral Agreement An agreement between the Employer and a participating Employee or Participating Employee Contractor where such individuals authorize the Employer to withhold a specified percentage or dollar amount of Compensation for deposit to the Plan on behalf of such Employee or Employee Contractor.

1.76 Savings Incentive Match Plan for Employees (SIMPLE) A plan adopted under Code Section 401(k)(11) or 408(p) under which eligible Employees are permitted to make Elective Deferrals to a qualified plan or Individual Retirement Account.

1.77 Self-Employed Individual An individual who has Earned Income for the taxable year from the trade or business for which the Plan is established including an individual who would have had Earned Income but for the fact that the trade or business had no Net Profit for the taxable year.

1.78 Service The period of current or prior employment with the Employer including any imputed period of employment which must be counted under USERRA. If the Employer maintains a plan of a predecessor employer, service for such predecessor shall be treated as Service for the Employer. Service is determined under an Hours Counting Method or Elapsed Time Method as selected by the Employer in the Adoption Agreement.

1.79 Service Provider The individual or entity who may be retained by the Plan Administrator on behalf of the Plan to provide specified administrative services to the Plan.

1.80 Shareholder Employee An Employee or officer who owns [or is considered as owning within the meaning of Code Section 318(a)(1)], on any day during the taxable year of an electing small business corporation (S Corporation), more than 5% of such corporation's outstanding stock.

1.81 Simplified Employee Pension Plan A plan under which the Employer makes contributions for eligible Employees pursuant to a written formula. Contributions are made to an Individual Retirement Account which meets the requirements of Code Section 408(k).

1.82 Spouse The individual to whom a Participant is married, or was married in the case of a deceased Participant who was married at the time of his or her death. A former Spouse will be treated in the same manner as a Spouse to the extent provided under a Qualified Domestic Relations Order as described in Code Section 414(p).

1.83 Super Top-Heavy Plan A Plan described at paragraph 1.95 under which the Top-Heavy Ratio exceeds 90%.

1.84 *Top-Heavy Determination Date* For the first Plan Year of the Plan, the last day of that year. For any Plan Year subsequent to the first Plan Year, the last day of the preceding Plan Year.

1.85 Top-Heavy Plan For any Plan Year, the Employer's Plan is Top-Heavy if any of the following conditions exist:

- (a) if the Top-Heavy Ratio for the Employer's Plan exceeds 60% and this Plan is not part of any Required Aggregation Group or Permissive Aggregation Group of plans.
- (b) if the Employer's Plan is a part of a Required Aggregation Group of plans but not part of a Permissive Aggregation Group and the Top-Heavy Ratio for the group of plans exceeds 60%.
- (c) if the Employer's Plan is a part of a Required Aggregation Group and part of a Permissive Aggregation Group of plans and the Top-Heavy Ratio for the Permissive Aggregation Group exceeds 60%.

1.86 Top-Heavy Ratio

- (a) If the Employer maintains one or more Defined Contribution plans (including any Simplified Employee Pension Plan) and the Employer has not maintained any Defined Benefit Plan which during the five-year period ending on the Determination Date(s) has or has had accrued benefits, the Top-Heavy Ratio for this Plan alone, or for the Required or Permissive Aggregation Group as appropriate, is a fraction,
 - (1) the numerator of which is the sum of the account balances of all Key Employees as of the Determination Date(s) [including any part of any account balance distributed in the 5year period ending on the Determination Date(s)], and
 - (2) the denominator of which is the sum of all account balances [including any part of any account balance distributed in the five-year period ending on the Determination Date(s)], both computed in accordance with Code Section 416 and the regulations thereunder.

Both the numerator and denominator of the Top-Heavy Ratio are increased to reflect any contribution not actually made as of the Determination Date but which is required to be taken into account on that date under Code Section 416 and the regulations thereunder.

- If the Employer maintains one or more Defined Contribution Plans (including any Simplified **(b)** Employee Pension Plan) and the Employer maintains or has maintained one or more Defined Benefit Plans which during the five-year period ending on the Determination Date(s) has or has had any accrued benefits, the Top-Heavy Ratio for any Required or Permissive Aggregation Group, as appropriate, is a fraction, the numerator of which is the sum of account balances under the aggregated Defined Contribution Plan or Plans for all Key Employees, determined in accordance with (a) above, and the Present Value of accrued benefits under the aggregated Defined Benefit Plan or Plans for all Key Employees as of the Determination Date(s), and the denominator of which is the sum of the account balances under the aggregated Defined Contribution Plan or Plans for all Participants, determined in accordance with (a) above, and the Present Value of accrued benefits under the Defined Benefit Plan or Plans for all Participants as of the Determination Date(s), all determined in accordance with Code Section 416 and the regulations thereunder. The accrued benefits under a Defined Benefit Plan in both the numerator and denominator of the Top-Heavy Ratio are increased for any distribution of an accrued benefit made in the five-year period ending on the Determination Date.
- (c) For purposes of (a) and (b) above, the value of account balances and the Present Value of accrued benefits will be determined as of the most recent Valuation Date that falls within or ends with the 12-month period ending on the Determination Date, except as provided in Code Section 416 and the regulations thereunder for the first and second Plan Years of a Defined Benefit Plan. The account balances and accrued benefits of a Participant who is not a Key Employee but who was a Key Employee in a prior year, or who has not been credited with at least one Hour of Service with any Employer maintaining the Plan at any time during the five-year period ending on the Determination Date, will be disregarded. The calculation of the Top-Heavy Ratio, and the extent to which distributions, rollovers, and transfers are taken into account will be made in accordance with Code Section 416 and the regulations thereunder. Qualified Voluntary Employee

Contributions will not be taken into account for purposes of computing the Top-Heavy Ratio. When aggregating plans, the value of account balances and accrued benefits will be calculated with reference to the Determination Dates that fall within the same calendar year. The accrued benefit of a Participant other than a Key Employee shall be determined under the method, if any, that uniformly applies for accrual purposes under all Defined Benefit Plans maintained by the Employer, or if there is no such method, as if such benefit accrued not more rapidly than the slowest accrual rate permitted under the fractional rule of Code Section 411(b)(1)(C).

1.87 Top-Paid Group The group consisting of the top 20% of Employees when ranked on the basis of Compensation paid during such year. For purposes of determining the number of Employees in the group (but not who is in it), the following Employees shall be excluded:

- (a) Employees who have not completed 6 months of Service by the end of the year;
- (b) Employees who normally work less than 17¹/₂ hours per week by the end of the year;
- (c) Employees who normally work not more than 6 months during any year;
- (d) Employees who have not attained age 21 by the end of the year;
- (e) Employees included in a collective bargaining unit, covered by an agreement between Employee representatives and the Employer, where retirement benefits were the subject of good faith bargaining, if they constitute at least 90% of the Employer's workforce and the Plan covers only non-union Employees; and
- (f) Employees who are nonresident aliens and who receive no Earned Income which constitutes income from sources within the United States.

1.88 *Transfer Contribution* A non-taxable transfer of a Participant's benefit directly from a Qualified Plan to this Plan. This type of transfer does not constitute constructive receipt of plan assets.

1.89 Trust The assets of the Plan held by or in the name of the Trustee.

1.90 Trustee The Trustee named in the Adoption Agreement or its successor serving from time to time.

1.91 USERRA The Uniform Services Employment and Reemployment Rights Act of 1994, as amended effective August 5, 1996.

1.92 Valuation Date The last day of the Plan Year and such other date(s) as specified in the Adoption Agreement on which the fair market value of Plan assets is determined. The Trustee must also value the Trust Fund on such other Valuation Dates as directed by the Plan Administrator or as elected by the Employer in the Adoption Agreement.

1.93 Vested Account Balance The aggregate value of the Participant's Vested Account Balances derived from Employer and Employee contributions (including Rollovers), whether vested before or upon death, including the proceeds of insurance contracts, if any, on the Participant's life. The provisions of Article VIII shall apply to a Participant who is vested in amounts attributable to Employer contributions, Employee contributions (or both) at the time of death or distribution.

1.94 Welfare Benefit Fund Any fund that is part of a plan of the Employer, or has the effect of a plan, through which the Employer provides welfare benefits to Employees or their beneficiaries. For these purposes, Welfare Benefit means any benefit other than those with respect to which Code Section 83(h) (relating to transfers of property in connection with the performance of services), Code Section 404 (relating to deductions for contributions to an Employees' trust or annuity and Compensation under a deferred payment plan), Code Section 404A (relating to certain foreign deferred compensation plans) apply. A "Fund" for purposes of this paragraph, is any social club, voluntary employee benefit association, supplemental unemployment benefit trust or qualified group legal service organization described in Code Section 501(c)(7), (9), (17) or (20); any trust, corporation, or other organization not exempt from income tax, or to the extent provided in regulations, any account held for an Employer by any person.

1.95 Year of Service

- (a) Hours of Service Method A 12-consecutive month period during which an Employee is credited with not less than 1,000 (or such lesser number as specified by the Employer in the Adoption Agreement) Hours of Service.
- (b) Elapsed Time Method For purposes of determining either an Employee's initial or continued eligibility to participate in the Plan, or the nonforfeitable interest in the Participant's account balance derived from Employer contributions, an Employee will receive credit for the aggregate of all time period(s) worked commencing with the Employee's first day of employment or reemployment and ending on the date a Break in Service begins. The first day of employment or reemployment is the first day the Employee performs an Hour of Service for the Employer. An Employee will also receive credit for any Period of Severance of less than 12 consecutive months. Fractional periods of a year will be expressed in terms of days.

Each Employee will share in Employer contributions for the period beginning on the date the Employee commences participation under the Plan and ending on the date on which such Employee terminates employment with the Employer or is no longer a member of an eligible class of Employees.

An Employer adopting the Elapsed Time Method is required to credit periods of Service and, under the Service spanning rules, certain periods of severance of 12 months or less. Under the first Service spanning rule, if an Employee severs from Service as a result of resignation, discharge or retirement and then returns to Service within 12 months, the Period of Severance is required to be taken into account. A situation may arise in which an Employee is absent from Service for any reason other than resignation, discharge, retirement or death and during the absence a resignation, discharge or retirement occurs. The second Service spanning rule provides that, under such circumstances, the Plan is required to take into account the period of time between the severance from Service date (i.e., the date of resignation, discharge or retirement) and the first anniversary of the date on which the Employee was first absent, if the Employee returns to Service on or before such first anniversary date.

ARTICLE II

ELIGIBILITY REQUIREMENTS

2.1 Participation

- (a) Employees and who meet the eligibility requirements in the Adoption Agreement on the Effective Date of the Plan shall become Participants as of the Effective Date of the Plan. If elected in the Adoption Agreement, all Employees employed on the Effective Date of the Plan may participate, even if they have not satisfied the Plan's specified eligibility requirements. Employees hired after the Effective Date of the Plan, upon meeting the eligibility requirements, shall become Participants on the Entry Date selected in the Adoption Agreement. The Employee must satisfy the eligibility requirements specified in the Adoption Agreement and be employed on the Entry Date to become a Participant in the Plan.
- In the event that an Employee has satisfied the eligibility requirements, but is not employed on the (b) next Entry Date, such Employee will become a Participant upon his or her rehire, if such individual is rehired before incurring a Break In Service or Period of Severance. If such individual incurs a Break in Service or Period of Severance before his or her rehire, such individual will be treated as a new Employee and will have to requalify under the Plan's eligibility requirements. In the event an Employee who is not a member of the eligible class of Employees becomes a member of the eligible class, such Employee shall participate immediately if such Employee has satisfied the minimum age and Service requirements and would have previously become a Participant had he or she been in the eligible class. A former Participant will be eligible to authorize Elective Deferrals and may make other Employee contributions as permitted under the Plan as of the first payroll period beginning after his or her date of rehire. A former Participant shall again become a Participant with respect to Employer related contributions on the next Entry Date following the date on which the individual is rehired. However, for purposes of Employer related contributions determined after the Participant's Entry Date, Compensation and Service shall be considered from his or her date of rehire.

2.2 Change in Classification of Employment In the event a Participant becomes ineligible to participate because he or she is no longer a member of an eligible class of Employees as elected by the Employer in the Adoption Agreement, such Employee shall participate upon his or her return to an eligible class of Employees. For this purpose, a Participant's Compensation and Service shall be considered from the date of change. A former Participant must be eligible to begin Elective Deferrals and/or other Employee contributions as of the first payroll period beginning after the date of change of classification.

2.3 Computation Period To determine Years of Service and Breaks in Service for purposes of eligibility under the Hours of Service Method, the 12-consecutive month period shall commence on the date on which an Employee first performs an Hour of Service for the Employer, and each anniversary thereof. If however, the eligibility period specified in the Adoption Agreement is one year or less and the Employee fails to complete the hours requirement during his or her first employment year, the second and succeeding 12-consecutive month periods shall commence on the first day of the Plan Year beginning prior to the anniversary of the date the Employee first performed an Hour of Service regardless of whether the Employee is entitled to be credited with 1,000 (or such lesser number as specified by the Employer in the Adoption Agreement) Hours of Service during his or her first employment year.

2.4 *Employment Rights* Participation in the Plan shall not confer upon a Participant any employment rights, nor shall it interfere with the Employer's right to terminate the employment of any Employee at any time.

2.5 Service with Controlled Groups All Years of Service with other members of a controlled group of corporations [as defined in Code Section 414(b)], trades or businesses under common control [as defined in Code Section 414(c)], or members of an affiliated service group [as defined in Code Section 414(m)] shall be credited for purposes of determining an Employee's eligibility to participate.

2.6 Leased Employees A leased Employee, with respect to a Standardized Plan, shall be treated as an Employee of the recipient Employer. Contributions or benefits provided by the leasing organization which are attributable to services performed for the recipient Employer shall be treated as provided by the recipient Employer.

However, a Leased Employee shall not be considered an Employee of the recipient if such Employee is covered by a money purchase pension plan sponsored by the leasing organization providing:

- (a) a non-integrated Employer contribution rate of at least 10% of Compensation, [as defined in Code Section 415(c)(3) but including amounts contributed by the Employer pursuant to a salary reduction agreement, which are excludable from the Employee's gross income under a cafeteria plan covered by Code Section 125, a cash or deferred profit-sharing plan under Code Section 401(k), a Simplified Employee Pension Plan under Code Section 402(h)(1)(B) and a tax-sheltered annuity under Code Section 403(b)],
- (b) immediate participation, and
- (c) full and immediate vesting.

This exclusion is only available if Leased Employees do not constitute more than 20% of the recipient's non-Highly Compensated work force.

The Plan Administrator must apply this paragraph 2.6 consistent with Code Sections 414(n) and 414(o) and the regulations issued thereunder.

The Employer must specify in the Adoption Agreement the manner in which the Plan will determine the allocation of Employer contributions and Participant forfeitures on behalf of a Participant if the Participant is a Leased Employee covered by a plan maintained by the leasing organization.

2.7 Waiver of Participation. The Employer may treat Employees who waive participation in the Plan as a nondiscriminatory class of Employees who are ineligible to participate therein by making the proper designation in the Adoption Agreement. A waiver shall not be considered a cash or deferred arrangement if it is irrevocable, applies to all Plans maintained by the Employer, and is made prior to the date on which the Employee is first eligible to participate in the Plan of the Employer. The Plan Administrator shall establish uniform and nondiscriminatory procedures as it deems necessary to carry out this provision including but not limited to, rules prescribing the timing and filing of elections not to participate.

An Employee or Participant continues to earn credit for each year of Eligibility Service or Vesting Service he or she completes and his or her account (if any) will share in the gains or losses of the Trust during the periods he or she elects not to participate.

Plan Document

ARTICLE III

EMPLOYER CONTRIBUTIONS

3.1 Contribution Amount

- (a) The Employer will make periodic contributions to the Plan in accordance with the formula or formulas selected in the Adoption Agreement.
- (b) The Employer shall also make Matching, Top-Heavy Plan minimum contributions and any other Employer contribution for the benefit of Participants who are covered by USERRA. Employer Matching Contributions under USERRA shall be made with respect to the Plan Year for which the Participant exercises his or her right to make-up Elective Deferrals and/or other Employee contributions for prior years. Top-Heavy minimum contributions and other Employer contributions for USERRA-protected Service shall be made during the Plan Year in which the individual returns to employment with the Employer. Employer contributions required under USERRA are not increased or decreased with respect to Plan investment earnings for the period to which such contributions relate. The Employer's contribution for any Plan Year shall be subject to the limitations on allocations contained in Article X.

3.2 Expenses and Fees The Employer may reimburse the Plan for all expenses and fees incurred in the administration of the Plan or Trust and paid from the assets of the Plan. Such expenses shall include, but shall not be limited to, fees for professional services, recordkeeping services, printing and postage. Brokerage commissions may not be reimbursed. If such expenses and fees are not paid from the Plan, the Employer may pay such expenses and fees directly. Reimbursement of any Plan fees will be considered Employer contributions subject to Code Sections 404 and 415.

3.3 **Responsibility for Contributions** The Trustee shall not be required to determine if the Employer has made a contribution or if the amount contributed from its general assets is in accordance with the Adoption Agreement or the Code. The Employer shall have sole responsibility in this regard. The Trustee shall be accountable solely for contributions actually received within the limits of Article X.

3.4 *Return of Contributions* Contributions made to the Plan by the Employer shall be irrevocable except as provided below:

- (a) Any contribution forwarded to the Trustee due to a mistake of fact, provided that the contribution is returned to the Employer within one year of the date of the contribution.
- (b) In the event that the Commissioner of Internal Revenue determines that the Plan is not initially qualified under the Internal Revenue Code, any contribution dependent on the initial qualification by the Employer must be returned to the Employer within one year after the date the initial qualification is denied, but only if the application for the qualification is made by the time prescribed by law for filing the Employer's return for the taxable year in which the Plan is adopted, or such later date as the Secretary of the Treasury may prescribe.
- (c) Contributions forwarded to the Trustee are presumed to be deductible and are conditioned on their deductibility. Contributions which are determined by the Internal Revenue Service to not be deductible will be returned to the Employer.

3.5 Merger of Assets from Another Plan The Employer may in its sole discretion direct the Trustee to accept assets from another Defined Contribution Plan, or to transfer assets to another Defined Contribution Plan, provided that such transfer satisfies the requirements of Code Section 414(1) and the regulations thereunder. Except with respect to elective transfers under Code Section 411(d)(6), all benefits which are protected under Code Section 411(d)(6) shall be preserved in the Plan accepting the transfer. The Employer shall have the right to refuse to accept or transfer assets for any reason. The Trustee shall have the right to refuse to accept an in-kind transfer of assets. Nothing in this paragraph 3.5 shall give the Trustee the right to refuse to make a direct transfer of an Eligible Rollover Distribution if requested to do so by a Participant in accordance with paragraph 4.4.

۲

•

•
ARTICLE IV

EMPLOYEE CONTRIBUTIONS

4.1 Rollover Contributions Unless elected otherwise in the Adoption Agreement, a Participant/Employee may make a Rollover Contribution to the Plan established hereunder of all or any part of an amount distributed or distributable to him or her from a Qualified Plan or an Individual Retirement Account (IRA) qualified under Code Section 408 where the IRA was used as a conduit from a Qualified Plan provided:

- (a) the amount distributed to the Participant/Employee is deposited to the Plan no later than the sixtieth day after such distribution was received by the Participant/Employee,
- (b) the amount distributed is not one of a series of substantially equal periodic payments made for the life (or life expectancy) of the Participant or the joint lives (or joint life expectancies) of the Participant and the Participant's Designated Beneficiary, or for a specified period of ten years or more,
- (c) the amount distributed is not a required minimum distribution required under Code Section 401(a)(9),
- (d) if the amount distributed included property, such property is rolled over only upon the Trustee/Custodian's approval, or if sold, the proceeds of such property may be rolled over,
- (e) the amount distributed would otherwise be includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to Employer securities), and
- (f) the amount rolled over does not include any amounts contributed on an after-tax basis by the Participant to the Qualified Plan.

The Plan Administrator shall be held solely responsible for determining the tax free status of any Rollover Contribution made to this Plan, and the Trustee shall have no responsibility for any such determination.

4.2 Elective Transfer Contributions

- (a) Unless elected otherwise in the Adoption Agreement, a Participant or an Employee may arrange for the direct transfer of his or her entire benefit from a Qualified Plan to this Plan. Such transfer shall be made in cash and/or in-kind. The Employer and/or the Trustee/Custodian shall have the right to refuse to accept a transfer in-kind if such assets do not comply operationally, would result in a prohibited transaction, are not readily marketable or are not compatible with the Employer's investment policy objectives. For accounting and recordkeeping purposes, Transfer Contributions shall be treated in the same manner as Rollover Contributions.
- .(b) An Employer maintaining a Safe-Harbor Profit Sharing Plan in accordance with the provisions of paragraph 8.7, acting in a nondiscriminatory manner, may in its sole discretion refuse to allow transfer contributions to its profit-sharing plan, if such contributions are directly or indirectly being transferred from a Defined Benefit Plan, a money purchase plan (including a Target Benefit Plan), a stock bonus plan, or another profit-sharing plan which would otherwise provide for a life annuity form of payment to the Participant.
- (c) In the event the Employer accepts a Transfer Contribution from a Plan in which the Participant/Employee was directing the investment of his or her account, the Employer may, if the Employer determines that it is appropriate and not in violation of the nondiscrimination rules under Regulation Section 1.401(a)(4)-4, permit the Employee to continue to direct his or her investments in accordance with paragraph 14.9 with respect only to such Transfer Contribution.

4.3 Trustee-to-Trustee Transfer Contribution

- (a) The Employer may arrange for the direct transfer of a Participant's benefit from a Qualified Plan to this Plan. For accounting and recordkeeping purposes, Transfer Contributions shall be treated in the same manner as Rollover Contributions.
- (b) In the event the Employer accepts a Transfer Contribution from a Plan in which the Participant/Employee was directing the investments of his or her account, the Employer may, if appropriate and not in violation of the nondiscrimination rules under Regulation Section 1.401(a)(4)-4, permit the Employee to continue to direct his or her investments in accordance with paragraph 14.9 with respect only to such Transfer Contribution.
- (c) Notwithstanding any provision of this Plan to the contrary, to the extent that any optional form of benefit under this Plan permits a distribution prior to the Employee's retirement, death, disability, or severance from employment, and prior to Plan termination, the optional form of benefit is not available with respect to benefits attributable to assets (including the post-transfer earnings thereon) and liabilities that are transferred, within the meaning of Code Section 414(1), to this Plan from a money purchase pension plan qualified under Code Section 401(a) (other than any portion of those assets and liabilities attributable to voluntary Employee contributions).

4.4 Elective Deferrals

- A Participant may enter into a Salary Deferral Agreement with the Employer authorizing the (a) Employer to withhold a portion of such Participant's Compensation not to exceed the dollar limit under Code Section 402(g), as adjusted under Code Section 415(d), for the Applicable Calendar Year, or the percentage of Compensation specified in the Adoption Agreement. Any such contribution shall be credited to the Employee's Elective Deferral account. A Participant may terminate deferrals at any time and may amend his or her Salary Deferral Agreement to increase or decrease the deferral percentage as provided in the Adoption Agreement. The Participant shall notify the Plan Administrator of any change in his or her deferral election in writing or in such other form as permitted by the Plan Administrator. The Employer may, notwithstanding any limit to the contrary, in the Adoption Agreement, limit the maximum deferral percentage for Highly Compensated Employees by the issuance of advance notice to such Employees. If a Participant terminates his or her agreement, such Participant shall be permitted to put a new Salary Deferral Agreement into effect as provided in the Adoption Agreement. The Employer may also amend or terminate said agreement on written notice to the Participant, if required to maintain the qualified status of the Plan.
- (b) If a Participant has not authorized the Employer to withhold the maximum annual deferral amount pursuant to Code Section 402(g) and desires to increase the total amount withheld for a Plan Year, the Participant may authorize the Employer to withhold a supplemental amount up to 100% of his or her Compensation for one or more pay periods, if allowed and approved by the Employer. In no event may the amounts withheld under the Salary Deferral Agreement plus the supplemental withholding exceed the lesser of 25% of a Participant's Compensation for such Plan Year or any other limitation elected in the Adoption Agreement by the Employer.
- (c) Elective Deferrals shall be deposited as soon as administratively feasible after being withheld from the Participant's pay, but in no event later than 15 business days after the end of the month during which they were withheld.

4.5 Automatic Enrollment If the Employer so elects in the Adoption Agreement, each Employee eligible under the Employer's Code Section 401(k) cash or deferred arrangement shall automatically become a Participant in the Plan as of the first Entry Date after satisfying the Plan's eligibility requirements. However, such Employee may elect not to defer Compensation, or to change the amount of Compensation that will be deferred by filing a written notice with the Plan Administrator stating their election. Prior to an Employee's automatic enrollment becoming effective, the Plan Administrator will provide such Employee with appropriate guidance as to the procedures then in effect, for the Employee to make the alternative elections referenced above. Each Employee deferring Compensation pursuant to this paragraph shall be deemed to have consented to an Elective Deferral contribution in the amount specified by the Employer in the Adoption Agreement unless he/she has filed an election to the contrary with the Plan Administrator pursuant to the Plan's administrative procedures.

4.6 Make-Up Contributions under USERRA A Participant who has the right to make-up Elective Deferrals, Voluntary Contributions and/or Required Contributions under USERRA shall be permitted to increase his or her deferral election with respect to a make-up year without regard to any provision limiting contributions for such Plan Year. Make-up contributions shall be limited to the maximum amount permitted under the Plan and the statutory limitations applicable with respect to the make-up year. Employee-related make-up contributions must be made within the time period beginning on the date of reemployment and continuing for the lesser of five years or three times the period of military service.

ARTICLE V

PARTICIPANT ACCOUNTS

5.1 Separate Accounts The Plan Administrator or individual or institution appointed by the Plan Administrator shall establish a separate recordkeeping account for each Participant showing the fair market value of his or her Plan benefits. Each Participant's account shall be separated for recordkeeping purposes into the following sub-accounts:

- (a) Employer contributions:
 - (1) Matching Contributions
 - (2) Qualified Matching Contributions
 - (3) Qualified Non-Elective Contributions
 - (4) Discretionary Contributions
- (b) Employee contributions:
 - (1) Elective Deferrals
 - (2) Rollover and Transfer Contributions

Allocation Date Individual Participant recordscepting accounts are updated in accordance with paragraph 5.2 5.3 hereof as of each Allocation or Valuation Date or such other date as elected by the Employer in the Adoption Agreement. Plan Administrators utilizing a daily valuation system for Participant recordkeeping purposes shall process for any contributions, distributions, investment income or loss, investment transactions (including a purchase or sale of an investment alternative) and any other transactions which affect a Participant on a daily basis or when a transaction occurs. Plan Administrators utilizing a balance forward valuation system for Participant recordkeeping purposes will process contributions, distributions, investment income or loss, investment transactions (including a purchase or sale of an investment alternative) and any other transactions at the Plan level on the Allocation or Valuation Date specified in the Adoption Agreement. Individual Participant recordkeeping accounts will be updated within the allocation period on the date or dates determined by the Plan Administrator with respect to contributions and distributions. Investment earnings will be allocated at the end of the allocation period. Any other transactions which affect Participant accounts will be posted or allocated to individual Participant accounts on the next following Allocation Date unless the Plan Administrator elects, in a uniform and nondiscriminatory manner, to allocate such transactions as they occur. The Employer may elect to utilize a daily valuation system for a portion of the Plan and a balance forward valuation system for the balance of the Plan.

5.3 Allocations to Participant Accounts As of each Allocation or Valuation Date elected by the Employer in the Adoption Agreement and/or on any date within the allocation period selected by the Employer, each Participant's account shall be adjusted to reflect:

- (a) the Participant's share of the Employer's contribution and forfeitures as determined in the Adoption Agreement,
- (b) any Employee contributions,
- (c) any repayment of amounts previously distributed to a Participant upon a separation from Service and repaid by the Participant since the last Allocation or Valuation Date, and

(d) the Participant's proportionate share of any investment earnings and increase in the fair market value of the Trust since the last Allocation Date.

The Employer shall deduct from each account:

- (e) any withdrawals or payments made from the Participant's account since the last Allocation Date,
- (f) the Participant's proportionate share of any decrease in the fair market value of the Trust since the last Allocation Date, and
- (g) the Participant's proportionate share of any fees and expenses paid from the Plan.

5.4 Allocating Employer Contributions

- (a) The Employer's contribution shall be allocated to Participants in accordance with the allocation formula selected by the Employer in the Adoption Agreement and the minimum contribution for Top-Heavy Plans. Employer contributions shall be allocated to all Participants eligible to receive a contribution as provided in the Adoption Agreement.
- (b) Notwithstanding any provision of this Plan to the contrary, Participants will accrue the right to share in allocations of Employer contributions with respect to periods of qualified military service as provided in Code Section 414(u).
- (c) Unless otherwise specified in the Adoption Agreement, at the end of each Plan Year the Plan Administrator shall redetermine any Matching Contribution for each Participant based on his or her eligible annual Compensation. Any Participant for whom any Matching Contribution has not been sufficiently made in accordance with the Matching Contribution formula elected by the Employer shall receive an additional Matching Contribution so that the total annual deferrals reflected as a percentage of eligible annual Compensation are matched in accordance with the formula elected by the Employer.

5.5 Allocating Investment and Losses

- (a) Daily Valuation System Account balances are adjusted to reflect actual income received and investment gains and losses based on the closing price of Plan assets as determined on the most recent day securities were traded on the New York Stock Exchange or any other securities market. Investment earnings are determined separately for each investment alternative offered under the Plan.
 - (1) The value of a Participant's account invested in a mutual fund will equal the value of a share in such fund multiplied by the number of shares credited to the Participant's account.
 - (2) In the case of any pooled investment vehicle, gains or losses on the pooled investment vehicle will be allocated among the Participant's accounts in proportion to the value of each Participants's account invested in that investment vehicle immediately prior to the allocation date, and the gain or loss attributed to each investment vehicle will be credited to or charged against the Participant's account. Alternatively, the Plan Administrator or his designate may establish unit values for each pooled investment vehicle offered under the Plan in accordance with uniform procedures established by the Plan Administrator for this purpose. The value of the portion of a Participant's account invested in a pooled investment vehicle will equal the value of a unit in such investment vehicle multiplied by the number of units credited to the account.
 - (3) In the case of any investment that is held specifically for a Participant's account, any gain or loss on such investment will be charged or credited to that Participant's account.

If Participant investment direction is not offered, each Participant's account shall be credited with its proportionate share of the income and investment gains or losses of the Trust for the day or period since securities were last traded on a national securities market or valued.

Any investment gain or loss of the Trust that is not directly attributable to the investment of the account of any Participant (including, for example, any "float" earned on the disbursement account established for the Plan and not treated as part of the compensation of the Trustee/Custodian or paying agent for the Plan, and any 12b-1 or similar fees paid to the Plan) will be applied to pay administrative expenses of the Plan, with any excess remaining at the close of the Plan Year being allocated among the Participant's accounts in accordance with the procedure established by the Plan Administrator for this purpose.

- Balance Forward Valuation System Account balances are adjusted to reflect actual income and (b) investment gains and losses from the period beginning on the day following the last Allocation Date and ending on the current Allocation or Valuation Date. Each Participant's account shall receive a proportionate share of the actual income and investment gains and losses during the period. The value of accounts for allocation purposes shall be based on the value of all active Participant accounts (other than accounts with segregated investments) as of the last Allocation or Valuation Date less withdrawals, distributions and expenses plus any contributions including Elective Deferrals if any, paid from the Trust since the last Allocation or Valuation Date. Investment earnings shall be credited to all Participant accounts having a balance on the Allocation or Valuation Date regardless of the vested status of such account and regardless of the Participant's employment status. The Plan Administrator shall also have the right to adopt an alternative procedure for allocating income and investment gains and losses provided that such alternative procedure is uniform and does not discriminate in favor of Highly Compensated Employees. Any change in procedure shall be effective as of the next following Allocation Date or Valuation Date or such other date as agreed to by the Employer and the Plan Administrator. Accounts with segregated investments shall receive only the income or loss on such segregated investments. Investment earnings are determined separately for each investment alternative offered under the Plan.
 - (1) The value of a Participant's account invested in a mutual fund will equal the value of a share in such fund multiplied by the number of shares credited to the Participant's account.
 - (2) In the case of any pooled investment vehicle, gains or losses on the pooled investment vehicle will be allocated among the Participant's accounts in proportion to the value of each Participants's account invested in that investment vehicle immediately prior to the Allocation or Valuation Date, and the gain or loss attributed to each investment vehicle will be credited to or charged against the Participant's account. Alternatively, the Plan Administrator or his designate may establish unit values for each pooled investment vehicle offered under the Plan in accordance with uniform procedures established by the Plan Administrator for this purpose. The value of the portion of a Participant's account investment vehicle multiplied by the number of units credited to the account.
 - (3) In the case of any investment that is held specifically for a Participant's account, any gain or loss on such investment will be charged or credited to that Participant's account.

If Participant investment direction is not offered, each Participant's account shall be credited with its proportionate share of the income and investment gains or losses of the Trust for the day or period since securities were last traded on a national securities market or valued.

Any investment gain or loss of the Trust that is not directly attributable to the investment of the Account of any Participant (including, for example, any "float" earned on the disbursement account established for the Plan and not treated as part of the compensation of the Trustee or paying agent for the Plan, and any 12b-1 or similar fees paid to the Plan) will be applied to pay administrative expenses of the Plan, with any excess remaining at the close of the Plan Year being allocated among the Participant's accounts in accordance with the procedure established by the Plan Administrator for this purpose.

5.6 Allocation Adjustments The Plan Administrator or his designate, if applicable, shall have the right to redetermine the value of Participant accounts if a previous allocation was performed incorrectly. Such redetermination shall be made without regard to the reason for the incorrect allocation. Such reasons may include, but are not limited to, incorrect contribution or Employee information provided by the Employer or representative of the Employer, incorrect valuation of Plan assets, incorrect determination of investment income and gains or losses, improper interpretation of the Plan's allocation formulas or procedures, erroneous omission of Top-Heavy minimum contributions and failure to transmit, receive or interpret amendments to the allocation formulas, methods or procedures. Subject to express limits that may be imposed under the Code, the Plan Administrator reserves the right to delay the processing of any contribution, distribution or other transaction for any legitimate business reason (including, but not limited to, failure of systems or computer programs, failure of means of transmission of data, force majeure, the failure of any service provider to timely receive values or prices, or to correct for its errors omissions of any service provider). After having made any necessary adjustments, the Plan Administrator or his designate, if applicable, shall issue either revised or adjusted statements to Participants with an explanation of the allocation adjustments.

5.7 Participant Statements The Plan Administrator shall prepare a statement for each Participant not less than annually as of the close of the Plan Year. Statements may be prepared more frequently as agreed between the Plan Administrator and the Recordkeeper or other entity responsible for the maintenance of Plan records or for valuing Plan assets. Each statement shall show the additions to and subtractions from the Participant's account for the period since the last such statement and shall show the fair market value of the Participant's account as of the current statement date.

5.8 Changes in Method of Valuing Participants' Accounts If necessary or appropriate, the Plan Administrator may establish different or additional uniform and nondiscriminatory procedures for determining the fair market value of Participant's accounts under the Plan.

ARTICLE VI

RETIREMENT BENEFITS AND DISTRIBUTIONS

6.1 Normal Retirement Benefits A Participant shall be entitled to receive the balance held in his or her account upon attaining Normal Retirement Age or at such earlier dates as the provisions of this Article VI may permit. If a Participant elects to continue working past his or her Normal Retirement Age, he or she will continue as an active Participant. Unless the Employer elects otherwise in the Adoption Agreement, distribution shall be made to such Participant at his or her request prior to his or her actual retirement. Settlement shall be made in the normal form, or if elected, in one of the optional forms of payment provided below.

6.2 Early Retirement Benefits An Early Retirement benefit may be available if elected in the Adoption Agreement to individuals who meet the age and Service requirements specified in the Adoption Agreement. An individual who meets the Early Retirement Age requirements will become fully vested, regardless of any vesting schedule which otherwise might apply. If a Participant separates from Service with a nonforfeitable benefit before satisfying the age requirements, but after having satisfied the Service requirement, the Participant will be entitled to elect an Early Retirement benefit upon satisfaction of the age requirement.

6.3 Benefits on Termination of Employment

- (a) If a Participant terminates employment prior to Normal Retirement Age, such Participant shall be entitled to receive the vested balance held in his or her account payable at Normal Retirement Age in the normal form, or if elected, in one of the other forms of payment provided hereunder. If applicable, the Early Retirement benefit provisions may be elected. Notwithstanding the preceding, a former Participant may, if allowed in the Adoption Agreement, make application to the Employer requesting early payment of any deferred vested and nonforfeitable benefit due.
- (b) If a Participant terminates employment, and the value of the Participant's Vested Account Balance derived from Employer and Employee contributions is not greater than \$5,000, the Plan Administrator may require the Participant to receive a lump sum distribution of the value of the entire vested portion of such account balance and the nonvested portion will be treated as a forfeiture. The Plan Administrator shall continue to follow a consistent and nondiscriminatory policy, as may be established, regarding immediate cash-outs of Vested Account Balances.
- (c) For purposes of this Article, if the value of a Participant's Vested Account Balance is zero, the Participant shall be deemed to have received a distribution of such Vested Account Balance immediately following termination. If the Participant is reemployed prior to incurring five consecutive one-year Breaks in Service, he or she will be deemed to have immediately repaid such distribution. Notwithstanding the above, if the Employer maintains or has maintained a policy of not distributing any amounts until the Participant's Normal Retirement Age, the Employer can continue to uniformly apply such policy.
- (d) If a Participant terminates employment with a Vested Account Balance derived from Employer and Employee contributions in excess of \$5,000, and elects (with his or her Spouse's consent, if required) to receive 100% of the value of his or her Vested Account Balance in a lump sum, the nonvested portion will be treated as a forfeiture. The Participant (and his or her Spouse, if required) must consent to any distribution when the Vested Account Balance described above exceeds \$5,000 or if at the time of any prior distribution it exceeded \$5,000.

- (e) Distribution of less than 100% of the Participant's Vested Account Balance shall be permitted if the Participant is fully vested upon termination of employment.
- (f) If a Participant who is not 100% vested receives or is deemed to receive a distribution pursuant to this paragraph and resumes employment covered under this Plan, the Participant shall have the right to repay to the Plan the full amount of the distribution attributable to both Employer contributions and Elective Deferrals on or before the earlier of the date that the Participant incurs five consecutive one-year Breaks in Service following the date of distribution or five years after the first date on which the Participant is subsequently reemployed. In such event, the Participant's account shall be restored to the value thereof at the time the distribution was made. The account may be further increased by the Plan's income and investment gains and/or losses on the undistributed amount from the date of the distribution to the date of repayment.
- (g) A Participant shall have the option to postpone payment of his or her Plan benefits until his or her Required Beginning Date if the Participant's Vested Account Balance derived from Employer and Employee contributions is not greater than \$5,000. Any balance in a Participant's account resulting from his or her Employee contributions listed at paragraph 5.1(b) hereof not previously withdrawn, if any, may be withdrawn by the Participant immediately following separation from Service.
- (h) If a Participant ceases to be an active Employee as a result of a Disability, such Participant shall have the right to make an application for a disability retirement benefit payment. The Participant's account balance will be deemed "immediately distributable" as set forth in paragraph 6.4, and will be fully vested pursuant to paragraph 9.2.

6.4 Restrictions on Immediate Distributions

- (a) An account balance is immediately distributable if any part of the account balance could be distributed to the Participant (or Surviving Spouse) before the Participant attains (or would have attained if not deceased) the later of the Normal Retirement Age or age 62.
- (b) If the value of a Participant's Vested Account Balance derived from Employer and Employee contributions exceeds (or at the time of any prior distribution exceeded) \$5,000, and the account balance is immediately distributable, the Participant and his or her Spouse (or where either the Participant or the Spouse has died, the survivor) must consent to any distribution of such account balance. The consent of the Participant and the Spouse shall be obtained in writing by the Plan Administrator within the 90-day period ending on the Annuity Starting Date, which is the first day of the first period for which an amount is paid as an annuity or any other form. The Plan Administrator shall notify the Participant and the Participant's Spouse of the right to defer any distribution until the Participant's account balance is no longer immediately distributable. Such notification shall include a general description of the material features, and an explanation of the relative values of, the optional forms of benefit available under the Plan in a manner that would satisfy the notice requirements of Code Section 417(a)(3), and shall be provided no less than 30 days and no more than 90 days prior to the Annuity Starting date.

If the distribution is one to which Code Sections 401(a)(11) and 417 do not apply, such distributions may commence less than 30 days after the notice required under Regulation Section 1.411(a)-11-(c) is given provided that:

- (1) the Plan Administrator clearly informs the Participant that the Participant has a right to a period of at least 30 days after receiving the notice to consider the decision of whether or not to elect a distribution (and, if applicable, a particular distribution option), and
- (2) the Participant after receiving the notice, affirmatively elects a distribution.

If a distribution is one to which Code Section 417 does apply, the distribution may commence less than 30 days, but not less than 7 days after the notice required under Regulation Section 1.411(a)-11-(c) is given, provided that the conditions of sub-paragraphs (1) and (2) above are satisfied with regard to both the Participant and the Participant's Spouse.

(c) Notwithstanding the foregoing, only the Participant need consent to the commencement of a distribution in the form of a Qualified Joint and Survivor Annuity while the account balance is immediately distributable. Furthermore, if payment in the form of a Qualified Joint and Survivor Annuity is not required with respect to the Participant pursuant to paragraph 8.7 of the Plan, only the Participant need consent to the distribution of an account balance that is immediately distributable. Neither the consent of the Participant nor the Participant's Spouse shall be required to the extent that a distribution is required to satisfy Code Section 401(a)(9) or Code Section 415 or constitutes Excess Deferrals, Excess Contributions or Excess Aggregate Contributions. In addition, upon termination of this Plan if the Plan does not offer an annuity option (purchased from a commercial provider), the Participant's account balance may, without the Participant's consent, be distributed to the Participant or transferred to another Defined Contribution Plan [other than an employee stock ownership plan as defined in Code Section 4975(e)(7)] within the same controlled group.

6.5 Normal and Optional Forms of Payment The normal form of payment for a profit-sharing plan satisfying the requirements of paragraph 8.7 hereof shall be a lump sum with no option for annuity payments. For all other plans, the normal form of payment hereunder shall be a Qualified Joint and Survivor Annuity as provided under Article VIII. A Participant whose Vested Account Balance derived from Employer and Employee contributions exceeds \$5,000 or if at the time of any prior distribution it exceeded \$5,000, shall (with the consent of his or her Spouse) have the right to receive his or her benefit in a lump sum or in installments if permitted in the Adoption Agreement. Installment payments need not be equal or substantially equal until such time as the individual reaches his or her Required Beginning Date. Installment payments which are intended to be equal or substantially equal can be made monthly, quarterly, semi-annually or annually based on any period not extending beyond the Joint and Survivor life expectancy of the Participant and his or her Designated Beneficiary.

Benefits payable in the form of a lump sum may be distributed in cash or in-kind at the Participant's request. Inkind distributions are only available with respect to assets which can be distributed in-kind. Benefits payable in the form of installments shall be paid in cash.

The normal form of payment shall be automatic, unless the Participant files a written request with the Employer prior to the date on which the benefit is automatically payable, electing another option available under the Plan. No amendment to the Plan may eliminate one of the optional distribution forms described above.

6.6 Commencement of Benefits

- (a) Unless the Participant elects otherwise, distribution of benefits will begin no later than the 60th day after the close of the Plan Year in which the latest of the following events occurs:
 - (1) the Participant attains age 65 (or Normal Retirement Age if earlier),
 - (2) the 10th anniversary of the year in which the Participant commenced participation in the Plan, or
 - (3) the Participant terminates Service with the Employer.
- (b) Notwithstanding the foregoing, the failure of a Participant and Spouse (if necessary) to consent to a distribution while a benefit is immediately distributable within the meaning of paragraph 6.4 hereof, shall be deemed an election to defer commencement of payment of any benefit sufficient to satisfy this paragraph.

6.7 **Claims Procedures** Upon retirement, death, severance of employment, or any and all types of claims regarding benefits under the Plan, the Participant or his or her representative may make application to the Employer requesting payment of benefits due and the manner of payment. If no application for benefits is made, the Employer shall automatically pay any vested benefit due hereunder in the normal form at the time prescribed at paragraph 6.6. If an application for benefits is made, the Employer shall accept, deny, or modify such request and shall notify the Participant in writing setting forth the response of the Employer and in the case of a denial or modification the Employer shall:

- (a) state the specific reason or reasons for the denial,
- (b) provide specific reference to pertinent Plan provisions on which the denial or modification is based,
- (c) provide a description of any additional material or information necessary for the Participant or his representative to perfect the claim and an explanation of why such material or information is necessary, and
- (d) explain the Plan's claim review procedure as contained in this Plan.

In the event the request is rejected or modified, the Participant or his or her representative may within 60 days following receipt by the Participant or representative of such rejection or modification, submit a written request for review by the Employer of its initial decision. Within 60 days following such request for review (120 days if extraordinary circumstances exist), the Employer shall render its final decision in writing to the Participant or representative stating specific reasons for such decision. If the Participant or representative is not satisfied with the Employer's final decision, the Participant or representative can institute an action in a Federal court of competent jurisdiction; for this purpose, process would be served on the Employer.

6.8 In-Service Withdrawals

- Employee Contributions An Employee may withdraw all or any part of the fair market value of (a) his or her contributions as described in Article IV, other than Elective Deferrals, upon request to the Plan Administrator. Employee Rollover and Transfer Contributions may be withdrawn at any time. Paragraph (b) below details the requirements for withdrawal of Elective Deferrals. Such request shall be in accordance with the procedures established by the Plan Administrator. A Participant may withdraw all or any part of the fair market value of his or her pre-1987 Voluntary Contributions with or without withdrawing the earnings attributable thereto. Post-1986 Voluntary Contributions may only be withdrawn along with a portion of the earnings thereon. The amount of the earnings to be withdrawn is determined by using the formula: DA [1-(V + V+E)], where DA is the distribution amount, V is the amount of Voluntary Contributions and V+E is the amount of Voluntary Contributions plus the earnings attributable thereto. The aggregate value of the Participant's Vested Account Balances derived from Employer and Employee contributions (including rollovers), whether vested before or upon death, includes the proceeds of insurance contracts, if any, on the Participant's life. The provisions of this article shall apply to a Participant who is vested in amounts attributable to Employer contributions, Employee contributions (or both) at the time of death or distribution.
- (b) Employer Contributions Money purchase pension plans and Target Benefit Plans may not allow in-service withdrawals. If specified in the Adoption Agreement, a Participant in a profitsharing plan is eligible to withdraw all or any part of the fair market value of his or her vested Employer contributions plus the investment earnings thereon as of the previous Allocation or Valuation Date. Such request shall be made in accordance with the procedures established by the Plan Administrator. An in-service withdrawal shall not be eligible for redeposit to the Trust. A withdrawal under this paragraph shall not prohibit such Participant from sharing in any future Employer contribution he or she would otherwise be eligible to share in. Subject to Article VIII, Joint and Survivor Annuity Requirements (if applicable), a Participant may withdraw any part of his or her Qualified Voluntary Contribution account by making application to the Plan

Administrator. A request to withdraw amounts pursuant to this paragraph must be consented to by the Participant's Spouse unless the Plan satisfies the safe-harbor under paragraph 8.7 hereof. Spousal consent, if required, shall comply with the requirements of paragraph 6.4 relating to immediate distributions. Elective Deferrals, Qualified Non-Elective Contributions, and Qualified Matching Contributions, and income allocable to each, are not distributable to a Participant earlier than upon separation from Service, death, or Disability. Such amounts may also be distributed upon:

- termination of the Plan without the establishment of another Defined Contribution Plan other than an employee stock ownership plan [as defined in Code Sections 4975(e)(7)] or a Simplified Employee Pension Plan [as defined in Code Section 408(k)], or SIMPLE IRA plan [as defined in Code Section 408(p)],
- (2) the disposition by a corporation to an unrelated corporation of substantially all of the assets [within the meaning of Code Section 409(d)(2)] used in a trade or business of such corporation if such corporation continues to maintain this Plan after the disposition, but only with respect to Employees who continue employment with the corporation acquiring such assets,
- (3) the disposition by a corporation to an unrelated entity of such corporation's interest in a subsidiary [within the meaning of Code Section 409(d)(3)] if such corporation continues to maintain this Plan, but only with respect to Employees who continue employment with such subsidiary, or
- (4) the attainment of age 59%.
- (c) Notwithstanding any provisions of the Plan to the contrary, to the extent that any optional form of benefit under this Plan permits a distribution prior to the Participant's retirement, death, Disability, or separation from Service, and prior to Plan termination, the optional form of benefits is not available with respect to benefits attributable to assets (including the post-transfer earnings thereon) and liabilities that are transferred within the meaning of Code Section 414(1), to this Plan from a money purchase plan qualified under Code Section 401(a) (other that any portion of those assets and liabilities attributable to Voluntary Employee Contributions).
- (d) *Partially Vested Participants* If a distribution is made at a time when a Participant has a nonforfeitable right to less than 100% of the account balance derived from Employer contributions and the Participant may increase the nonforfeitable percentage in the account:
 - (1) a separate account will be established for the Participant's interest in the Plan as of the time of the distribution, and
 - (2) at any relevant time the Participant's nonforfeitable portion of the separate account will be equal to an amount ("X") determined by the formula:

$$X = P [AB + (R \times D)] - (R \times D)$$

For purposes of applying the formula: "P" is the nonforfeitable percentage at the relevant time, "AB" is the account balance at the relevant time, "D" is the amount of the distribution and "R" is the ratio of the account balance at the relevant time to the account balance after distribution.

6.9 Hardship Withdrawals If elected in the Adoption Agreement, a Participant may request a Hardship withdrawal as provided in this paragraph. Hardship withdrawals are subject to the spousal consent requirements contained in Code Sections 401(a)(11) and 417. Such request shall be made in accordance with procedures adopted by the Plan Administrator who shall have sole authority to authorize a Hardship withdrawal pursuant to the following rules.

(a) Administrative Requirements

- (1) The Participant must have obtained all distributions, other than Hardship distributions, and all nontaxable loans under all plans maintained by the Employer.
- (2) The Participant's Elective Deferrals, Voluntary Contributions and Required Contributions will be suspended for all plans maintained by the Employer (other than benefits under Code Section 125 plans) for twelve months after the receipt of the Hardship distribution,
- (3) The distribution is not in excess of the amount of the immediate and heavy financial need described at paragraph (b) including amounts necessary to pay any Federal, state or local income taxes or penalties reasonably anticipated to result from the distribution.
- (4) All plans maintained by the Employer provide that a Participant may not make Elective Deferrals for the Participant's taxable year immediately following the taxable year of the Hardship distribution in excess of the applicable limit under Code Section 402(g) for such taxable year, less the amount of such Participant's Elective Deferrals for the taxable year during which the Hardship distribution was received.
- (5) Liquidation of a Participant's assets for the purpose of a Hardship withdrawal will be allocated on a pro-rata basis across all the investment alternatives in a Participant's account.

(b) Exclusive Reasons for Hardship Withdrawal

- (1) Expenses incurred or necessary for medical care, [described in Code Section 213(d)] of the Participant, his or her Spouse, children and other dependents,
- (2) The purchase (excluding mortgage payments) of the principal residence of the Participant,
- (3) Payment of tuition and related educational expenses for the next twelve (12) months of post-secondary education for the Participant, his or her Spouse, children or other dependents,
- (4) The need to prevent eviction of the Participant from or a foreclosure on the mortgage of, the Participant's principal residence, or
- (c) Order of Distribution If a request for Hardship withdrawal is approved by the Plan Administrator, funds shall be withdrawn from the following contribution types in the order which follows unless provided otherwise by administrative procedure:
 - (1) Elective Deferrals
 - (2) investment earnings on Elective Deferrals allocated as of the last day of the Plan Year ending before July 1, 1989

And, if elected in the Adoption Agreement:

- (3) Fully vested Employer related contributions plus the investment earnings thereon.
- (4) Qualified Matching Contributions, Qualified Non-Elective Contributions and Elective Deferrals reclassified as Voluntary Contributions plus the investment earnings thereon to the extent that they were credited to the Participant's account as of the last day of the Plan Year ending prior to July 1, 1989.

6.10 Direct Rollover of Benefits Notwithstanding any provision of the Plan to the contrary that would otherwise limit a Participant's election under this paragraph, for distributions made on or after January 1, 1993, a Participant may elect, at the time and in the manner prescribed by the Plan Administrator, to have any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan or Individual Retirement Account specified by the Participant in a Direct Rollover. Any portion of a distribution which is not paid directly to an Eligible Retirement Plan or Individual Retirement Account shall be distributed to the Participant. For purposes of this paragraph, a surviving Spouse or a Spouse or former Spouse who is an alternate payee under a Qualified Domestic Relations Order as defined in Code Section 414(p), will be permitted to elect to have any Eligible Rollover Distribution paid directly to an Individual Retirement Account (IRA) or an individual retirement annuity (IRA) or to another Oualified Plan in which the alternate payee is a participant.

6.11 Participant's Notice In the event that a Participant's benefit becomes payable under Plan terms or if a Participant requests distribution of his or her benefit, the Plan Administrator shall provide such Participant with a notice regarding distribution of such benefit. The notice shall describe any Plan related information regarding the distribution including the Joint and Survivor Annuity requirements provided at paragraph 8.5, if applicable, the normal and optional forms of payment provided at paragraph 6.5, and the information required in connection with an Eligible Rollover Distribution. Information in connection with an Eligible Rollover Distribution shall include the right to have the funds transferred directly to another Qualified Plan or Individual Retirement Account, the income tax withholding requirements, the rollover rules with respect to amounts distributed to the Participant, and the general tax rules which apply to such distributions. Such notice shall be provided to the Participant within the time period prescribed at paragraph 8.5 hereof or, if the safe-harbor provisions of paragraph 8.7 are applicable, not less than 30 days prior to the Annuity Starting Date, subject to a waiver period of a lesser number of days if elected by the Participant and if applicable, their Spouse.

6.12 Assets Transferred from Money Purchase Pension Plans Notwithstanding any provision of this Plan to the contrary, to the extent that any optional form of benefit under this Plan permits a distribution prior to the Employee's retirement, death, Disability, or severance from employment, and prior to Plan termination, the optional form of benefit is not available with respect to benefits attributable to assets (including the associated post-transfer earnings) and liabilities that are transferred, within the meaning of Code Section 414(1), to this Plan from a money purchase pension plan qualified under Code Section 401(a) (other than any portion of those assets and liabilities attributable to Voluntary Employee Contributions).

6.13 Assets Transferred from a 401(k) Plan If the Plan receives a direct transfer (by merger or otherwise) of Elective Deferrals (or amounts treated as Elective Deferrals) under a Plan with a Code Section 401(k) arrangement, the distribution restrictions of Code Sections 401(k)(2) and 401(k)(10) continue to apply to those transferred Elective Deferrals.

ARTICLE VII

DISTRIBUTION REQUIREMENTS

7.1 Joint and Survivor Annuity Requirements All distributions made under the terms of this Plan must comply with the provisions of Article VIII including, if applicable, the safe-harbor provisions thereunder.

7.2 Minimum Distribution Requirements All distributions required under this Article shall be determined and made in accordance with the minimum distribution requirements of Code Section 401(a)(9) and the regulations issued thereunder, including the minimum distribution incidental benefit rules found at Regulations Section 1.401(a)(9)-2. The entire interest of a Participant must be distributed or begin to be distributed no later than the Participant's Required Beginning Date. Life expectancy and joint and last survivor life expectancies are computed by using the expected return multiples found in Tables V and VI of Regulations Section 1.72-9.

7.3 Limits on Distribution Periods As of the First Distribution Calendar Year, distributions, if not made in a single sum, may only be made over one of the following periods (or a combination thereof):

- (a) the life of the Participant,
- (b) the life of the Participant and a Designated Beneficiary,
- (c) a period certain not extending beyond the life expectancy of the Participant, or
- (d) a period certain not extending beyond the joint and last survivor life expectancy of the Participant and a Designated Beneficiary.

7.4 Required Distributions on or after the Required Beginning Date

- (a) If a Participant's Benefit is to be distributed over (i) a period not extending beyond the life expectancy of the Participant or the joint life and last survivor expectancy of the Participant and the Participant's Designated Beneficiary or (ii) a period not extending beyond the life expectancy of the Designated Beneficiary, the amount required to be distributed for each calendar year, beginning with distributions for the First Distribution Calendar Year, must at least equal the sum obtained by dividing the Participant's Benefit by the Applicable Life Expectancy.
- (b) For calendar years beginning after 1988, the amount to be distributed each year beginning with distributions for the First Distribution Calendar Year, shall not be less than the quotient obtained by dividing the Participant's Benefit by the lesser of (i) the Applicable Life Expectancy or (ii) if the Participant's Spouse is not the Designated Beneficiary, the applicable divisor determined from the table set forth in Q&A-4 of Regulations Section 1.401(a)(9)-2. Distributions after the death of the Participant shall be distributed using the Applicable Life Expectancy as the relevant divisor without regard to Regulations Section 1.401(a)(9)-2.
- (c) The minimum distribution required for the Participant's First Distribution Calendar Year must be made on or before the Participant's Required Beginning Date. The minimum distribution for other calendar years, including the minimum distribution for the Distribution Calendar Year in which the Participant's Required Beginning Date occurs, must be made on or before December 31 of that Distribution Calendar Year.
- (d) If the Participant's Benefit is distributed in the form of an annuity, distributions thereunder shall be made in accordance with the requirements of Code Section 401(a)(9) and the Regulations thereunder.

- (e) For purposes of determining the amount of the required distribution for each Distribution Calendar Year, the account balance to be used is the account balance determined as of the last Valuation Date preceding the Distribution Calendar Year. This balance will be increased by the amount of any contributions or forfeitures allocated to the account balance after the Valuation Date in such preceding calendar year. Such balance will also be decreased by distributions made after the Valuation Date in such preceding Calendar Year.
- (f) For purposes of subparagraph 7.4(f), if any portion of the minimum distribution for the First Distribution Calendar Year is made in the second Distribution Calendar Year on or before the Required Beginning Date, the amount of the minimum distribution made in the second Distribution Calendar Year shall be treated as if it had been made in the immediately preceding Distribution Calendar Year.

7.5 Required Beginning Date

¢

.

- (a) Participants other than 5% Owners The Required Beginning Date is the later of the first day of April of the calendar year following the calendar year in which the Participant attains age 70% or the first day of April of the calendar year following the calendar year in which the Participant terminates employment with the Employer.
- (b) Participants who are 5% Owners The Required Beginning Date is the first day of April of the calendar year following the calendar year in which the Participant attains age 70½. A Participant is treated as a 5% owner for purposes of this paragraph if such Participant is a 5% owner as defined in Code Section 416(i) (determined without regard to whether the Plan is Top-Heavy) at any time during the Plan Year ending with or within the calendar year in which such owner attains age 66½ or any subsequent Plan Year. Once distributions have begun to a 5% owner under this paragraph, they must continue to be distributed even if the Participant ceases to be a 5% owner in a subsequent year.

(c) Transitional Rules:

- Any Participant, other than a 5% owner, attaining age 70½ in years after 1995 may elect by April 1 of the calendar year following the year in which the Participant attained age 70½, or by December 31, 1997 in the case of a Participant attaining age 70½ in 1996, to defer distributions until the calendar year following the calendar year in which the Participant retires. If no such election is made, the Participant will begin receiving distributions by the April 1 of the calendar year following the year in which the Participant attained age 70½, or by December 31, 1997 in the case of a Participant will begin receiving distributions by the April 1 of the calendar year following the year in which the Participant attained age 70½, or by December 31, 1997 in the case of a Participant attaining age 70½ in 1996.
- (2) Any Participant, other than a 5% owner, attaining age 70¼ in years prior to 1997 may elect to stop distributions and recommence by the April 1 of the calendar year following the year in which the Participant retires. In such event, the minimum distribution to such Participant shall be determined based upon the highest attained age of the individual and, if applicable, his or her Designated Beneficiary in the year during which distributions are to commence.

7.6 Election under Tax Equity and Fiscal Responsibility Act of 1982

(a) Notwithstanding the other requirements of this Article and subject to the requirements of Article VIII, Joint and Survivor Annuity Requirements, distribution on behalf of any Employee, including a 5% owner, may be made in accordance with all of the following requirements, regardless of when such distribution commences:

- the distribution by the Trust is one which would not have disqualified such Trust under Code Section 401(a)(9) as in effect prior to amendment by the Deficit Reduction Act of 1984,
- (2) the distribution is in accordance with a method of distribution designated by the Employee whose interest in the Trust is being distributed or, if the Employee is deceased, by a beneficiary of such Employee,
- (3) such designation was in writing, was signed by the Employee or the beneficiary, and was made before 1984,
- (4) the Employee had accrued a benefit under the Plan as of December 31, 1983,
- (5) the method of distribution designated by the Employee or the beneficiary specifies the time at which distribution will commence, the period over which distributions will be made, and in the case of any distribution upon the Employee's death, the beneficiaries of the Employee listed in order of priority.
- (b) A distribution upon death will not be covered by this transitional rule unless the information in the designation contains the required information described above with respect to the distributions to be made upon the death of the Employee.
- (c) For any distribution which commences before 1984, but continues after 1983, the Employee or the beneficiary to whom such distribution is being made, will be presumed to have designated the method of distribution under which the distribution is being made, if the method of distribution was specified in writing and the distribution satisfies the requirements in subparagraphs (a)(1) through (5) above.
- (d) If a designation is revoked, any subsequent distribution must satisfy the requirements of Code Section 401(a)(9) and the regulations thereunder. If a designation is revoked subsequent to the date distributions are required to begin, the Plan must distribute by the end of the calendar year following the calendar year in which the revocation occurs the total amount not yet distributed which would have been required to have been distributed to satisfy Code Section 401(a)(9) and the regulations thereunder, but for the Section 242(b)(2) election of the Tax Equity and Fiscal Responsibility Act of 1982. For calendar years beginning after 1988, such distributions must meet the minimum distribution incidental benefit requirements in Section 1.401(a)(9)-2 of the Income Tax Regulations. Any changes in the designation will be considered to be a revocation of the designation. However, the mere substitution or addition of another beneficiary (one not named in the designation) under the designation will not be considered to be a revocation of the designation, so long as such substitution or addition does not alter the period over which distributions are to be made under the designation, directly or indirectly (for example, by altering the relevant measuring life). In the case in which an amount is transferred or rolled over from one plan to another plan, the rules in Q&A J-2 and Q&A J-3 of the regulations shall apply.

7.7 Designation of Beneficiary Each Participant shall file a written designation of beneficiary with the Employer upon qualifying for participation in this Plan. Such designation shall remain in force until revoked by the Participant by filing a new beneficiary designation form with the Employer. The Designated Beneficiary of a married Participant shall be the Participant's Surviving Spouse, unless such Spouse has properly consented otherwise.

7.8 No Beneficiary Any portion of the amount payable hereunder which is not disposed of because of the Participant's or former Participant's failure to designate a beneficiary, or because all of the Designated Beneficiaries predeceased the Participant, shall be paid to his or her Spouse. If the Participant had no Spouse at the time of death, payment shall be made to his or her estate in a lump sum.

. .

٠

7.9 Distribution Beginning before Death This paragraph is applicable only after the Participant's Required Beginning Date as defined in paragraph 7.5. If the Participant dies after distribution of his or her interest has begun, the remaining portion of such interest will continue to be distributed at least as rapidly as under the method of distribution being used prior to the Participant's death.

7.10 Distribution Beginning after Death This paragraph is applicable before the Participant's Required Beginning Date as defined in paragraph 7.5, even if distributions have commenced from the Plan. If the Participant dies before distribution of his or her interest begins, distribution of the Participant's entire interest shall be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death, except to the extent that an election is made to receive distributions in accordance with (a) or (b) below:

- (a) if any portion of the Participant's interest is payable to a Designated Beneficiary, distributions may be made over the life or over a period certain not greater than the life expectancy of the Designated Beneficiary commencing on or before December 31 of the calendar year immediately following the calendar year in which the Participant died;
- (b) if the Designated Beneficiary is the Participant's Surviving Spouse, the date distributions are required to begin in accordance with (a) above shall not be earlier than the later of (i) December 31 of the calendar year immediately following the calendar year in which the Participant died or (ii) December 31 of the calendar year in which the Participant would have attained age 70½.

If the Participant has not made an election pursuant to this paragraph 7.10 by the time of his or her death, the Participant's Designated Beneficiary must elect the method of distribution no later than the earlier of (i) December 31 of the calendar year in which distributions would be required to begin under this section, or (ii) December 31 of the calendar year which contains the fifth anniversary of the date of death of the Participant. If the Participant has no Designated Beneficiary, or if the Designated Beneficiary does not elect a method of distribution, then distribution of the Participant's entire interest must be completed by December 31 of the calendar year containing the fifth anniversary of the Surviving Spouse dies after the Participant but before payments to such Spouse begin, the provisions of this paragraph with the exception of subparagraph (b) herein, shall be applied as if the Surviving Spouse were the Participant. For the purposes of this paragraph and paragraph 7.9, distribution of a Participant's interest is considered to begin on the Participant's Required Beginning Date (or, if the preceding sentence is applicable, the date distribution is required to begin to the Surviving Spouse). If distribution in the form of an annuity described in paragraph 7.4(e) irrevocably commences to the Participant before the Required Beginning Date, the date distribution is considered to begin is the date distribution actually commences.

7.11 Distribution of Excess Elective Deferrals

- (a) No Participant shall be permitted to defer under this Plan with respect to a calendar year more than the maximum dollar amount permitted under Code Section 402(g), as indexed, for such calendar year. If a Participant defers more than the maximum allowed due to mistake of fact, such Excess Elective Deferrals shall be distributed to the Participant no later than April 15 following the calendar year to which the excess is attributable. If an individual who participates in this Plan and in another plan which permits Elective Deferrals, defers more than the Code Section 402(g) maximum, such individual shall have the right to notify one or both plans by March 1 of the calendar year following the year to which the excess is attributable requesting a distribution of the Excess Elective Deferral. A Participant is deemed to notify the Plan Administrator of any Excess Elective Deferrals that arise by taking into account only those Elective Deferrals made to the Plan of the Employer. If distribution is requested, the applicable plan(s) shall make distribution of the excess no later than April 15 following the calendar year to which the excess is attributable. Excess Elective Deferrals which are distributed on a timely basis shall not be considered Annual Additions for the Limitation Year during which such amounts are deferred.
- (b) Excess Elective Deferrals shall be adjusted for any income or loss up to the end of the taxable year during which such excess was deferred. Income or loss will be calculated under any reasonable method consistently followed for all Participants, used to calculate investment earnings and losses.

۴.

(c) The amount a Participant receives as a distribution of his or her Excess Elective Deferrals is includible in income with respect to the taxable year to which the excess is attributable.

7.12 Distributions of Excess Contributions

- (a) Excess Contributions plus any income and minus any loss allocable thereto, shall be distributed to affected Participants no later than the last day of the Plan Year following the Plan Year to which the Excess Contributions are attributable. If such Excess Contributions are distributed more than 2½ months after the last day of the Plan Year to which the Excess Contributions are attributable, a 10% excise tax will be imposed on the Employer maintaining the Plan with respect to the principal amount of the excess.
- (b) Excess Contributions, including any amount recharacterized as a Voluntary Contribution, shall be treated as Annual Additions.
- (c) Excess Contributions shall be adjusted for any income or loss up to the end of the Plan Year to which such excess is attributable. The income or loss allocable to Excess Contributions is based on the amounts included in the ADP Test for the applicable Plan Year. Income or loss will be calculated under any reasonable method consistently followed for all Participants, used to calculate investment earnings and losses.
- (d) Excess Contributions shall be distributed from the Participant's Elective Deferral account and Qualified Matching Contribution account (if applicable) in proportion to the Participant's Elective Deferrals and Qualified Matching Contributions (to the extent used in the ADP test) for the test year. Excess Contributions shall be distributed from the Participant's Qualified Non-Elective Contribution account only to the extent that such Excess Contributions exceed the Participant's Elective Deferrals and Qualified Matching Contributions for the applicable test year.

7.13 Distribution of Excess Aggregate Contributions

- (a) Excess Aggregate Contributions including any income or loss allocable thereto, shall be forfeited, if forfeitable, or if not forfeitable, distributed to affected Participants no later than the last day of the Plan Year following the Plan Year to which the Excess Aggregate Contributions are attributable. If such Excess Aggregate Contributions are distributed more than 2½ months after the last day of the Plan Year to which the Excess Contributions are attributable, a 10% excise tax will be imposed on the Employer maintaining the Plan with respect to the principal amount of the excess.
- (b) Excess Aggregate Contributions shall be treated as Annual Additions with respect to the Plan Year to which such excess is attributable.
- (c) Excess Aggregate Contributions shall be adjusted for any income or loss up to the end of the Plan Year to which such excess is attributable. The income or loss allocable to Excess Aggregate Contributions is based on the amounts included in the ACP test for the applicable Plan Year. Income or loss will be calculated under any reasonable method used to calculate investment earnings and losses.
- (d) Excess Aggregate Contributions shall be forfeited if such amount is not vested. If vested, such excess shall be distributed on a pro-rata basis from the Participant's Voluntary Contribution account and, if applicable, the Participant's Qualified Non-Elective Contribution account, Matching Contribution account, Qualified Matching Contribution account, and/or Elective Deferral account.

7.14 Distributions to Minors and Individuals who are Legally Incompetent Benefits payable to either a minor or an individual who has been declared legally incompetent shall be paid, at the direction of the Plan Administrator, to the parent, guardian or other individual including the conservator appointed under applicable state law for the benefit of said minor or incompetent.

7.15 Unclaimed Benefits

۶.

- (a) The Plan Administrator shall notify Participants or beneficiaries by certified or registered mail to his or her last known address of record with the Employer when their benefits become distributable as provided at paragraph 6.11 hereof. If a Participant does not respond to the notice within 90 days of the date of the notice, the Plan Administrator may take reasonable steps to locate the Participant including, but not limited to, requesting assistance from the Employer, Employees, Social Security Administration and/or the Internal Revenue Service.
- (b) If the Participant cannot be located after a period of twelve months, or such other period determined in a uniform and nondiscriminatory manner by the Plan Administrator, the Plan Administrator shall treat the benefit as a forfeiture pursuant to paragraph 9.7. The forfeitures provisions of this paragraph 7.15(b) apply only to the Participant's or beneficiary's account balance which is less than \$5,000. If the Employer does not make a contribution for the Plan Year during which the forfeiture takes place, such amount shall first be applied to pay Plan expenses and, if there are no such expenses, shall then be allocated to eligible Participant accounts as if the amount were the Employer's contribution for such Plan Year.
- (c) If a Participant or Designated Beneficiary later makes a claim for such benefit, the Plan Administrator shall validate such claim and provide the Participant or Designated Beneficiary with all notices and other information necessary for the Participant or Designated Beneficiary to perfect the claim. If the claim for benefits is validated by the Plan Administrator, the Participant's account balance shall be restored to the benefit amount treated as a forfeiture. Such benefit shall not be adjusted for investment earnings or losses during the period beginning on the date of forfeiture and ending on the date of restoration. The funds necessary to restore the Participant's account will first be taken from amounts eligible for reallocation or other disposition as forfeitures with respect to the Plan Year. If such funds do not exist or if such funds are insufficient, the Employer will make a contribution prior to the date on which the benefit is payable to restore such Participant's account. Such benefit shall be paid or commence to be paid in the same manner as if the benefit was eligible for distribution on the date the claim for benefit is validated.
- (d) The Plan Administrator shall follow the same procedure in locating and subsequently treating as a forfeiture the benefit of a Participant whose benefit has been properly paid under Plan terms but where the Participant or Designated Beneficiary has not negotiated the benefit check(s).
- (e) Alternatively, if the Participant's account balance is less than \$5,000, the Plan Administrator may remit the entire amount to the Internal Revenue Service as federal tax withholding.
- (f) If the Participant's account balance is over \$5,000, the Plan Administrator after using all reasonable measures to locate the Participant or Designated Beneficiary, consistent with the procedures specified above, all applicable laws, regulations and other pronouncements under ERISA, may use any reasonable procedure to dispose of distributable Plan assets including but not limited to any of the following: (i) establishing an IRA in the name of the Participant or Designated Beneficiary with any institution, (ii) purchasing an annuity contract in the name of the Participant or Designated Beneficiary with the assets attributable to them in the Trust, or (iii) establishing a bank account for and in the name of the Participant or Designated Beneficiary.

ARTICLE VIII

JOINT AND SURVIVOR ANNUITY REQUIREMENTS

8.1 Applicability of Provisions The provisions of this Article shall apply to any Participant who is credited with at least one Hour of Service with the Employer on or after August 23, 1984 and such other Participants as provided in paragraph 8.8.

8.2 Payment of Qualified Joint and Survivor Annuity Unless an optional form of benefit is selected pursuant to a Qualified Election within the 90-day period ending on the Annuity Starting Date, a Participant's Vested Account Balance will be paid in the form of a Qualified Joint and Survivor Annuity. For this purpose, a Qualified Joint and Survivor Annuity with respect to an unmarried Participant's Vested Account Balance will be paid in the form of a life annuity. The Participant may elect to have such annuity distributed upon attainment of the Early Retirement Age under the Plan.

8.3 Payment of Qualified Pre-Retirement Survivor Annuity Unless an optional form of benefit has been selected within the Election Period pursuant to a Qualified Election, if a Participant dies before benefits have commenced then the Participant's Vested Account Balance shall be paid in the form of a life annuity for the life of the surviving Spouse. The surviving Spouse may elect to have such annuity distributed within a reasonable period after the Participant's death. If no election has been made within the Election Period prior to the Participant's death, the surviving Spouse shall have the right to select an optional form of benefit after the Participant's death. Such election will only be permitted if the surviving Spouse is provided with a notice similar to that required under paragraph 8.5 except that the notice will be modified to explain a life annuity rather than a Qualified Joint and Survivor Annuity.

A Participant who does not meet the age 35 requirement set forth in the Election Period as of the end of any current Plan Year may make a special qualified election to waive the Qualified Pre-Retirement Survivor Annuity for the period beginning on the date of such election and ending on the first day of the Plan Year in which the Participant will attain age 35. Such election shall not be valid unless the Participant receives a written explanation of the Qualified Pre-Retirement Survivor Annuity in such terms as are comparable to the explanation required under paragraph 8.5. Qualified Pre-Retirement Survivor Annuity coverage will be automatically reinstated as of the first day of the Plan Year in which the Participant attains age 35. Any new waiver on or after such date shall be subject to the full requirements of this Article.

8.4 *Qualified Election* A Qualified Election is an election to either waive a Qualified Joint and Survivor Annuity or a Qualified Pre-Retirement Survivor Annuity. Any such election shall not be effective unless:

- (a) the Participant's Spouse consents in writing to the election,
- (b) the election designates a specific beneficiary, including any class of beneficiaries or any contingent beneficiaries, which may not be changed without spousal consent or the Spouse expressly permits designations by the Participant without any further spousal consent,
- (c) the Spouse's consent acknowledges the effect of the election, and
- (d) the Spouse's consent is witnessed by a Plan representative or notary public.

A Participant's waiver of the Qualified Joint and Survivor Annuity shall not be effective unless the election designates a form of benefit payment which may not be changed without spousal consent or the Spouse expressly permits designations by the Participant without any further spousal consent. If it is established to the satisfaction of the Plan Administrator that the Participant is unmarried or that the Spouse cannot be located, a waiver will be deemed a Qualified Election. Any consent by a Spouse obtained under this provision (or establishment that the consent of a Spouse cannot be obtained) shall be effective only with respect to such Spouse. A consent that permits designations by the Participant without any requirement of further consent by such Spouse must acknowledge that the Spouse has the right to limit con sent to a specific beneficiary, and a specific form of benefit where applicable.

and that the Spouse voluntarily elects to relinquish either or both of such rights. A revocation of a prior waiver may be made by a Participant without the consent of the Spouse at any time before the commencement of benefits. The number of revocations shall not be limited. No consent obtained under this provision shall be valid unless the Participant has received notice as provided in paragraphs 8.5 and 8.6 below.

8.5 Notice Requirements for Qualified Joint and Survivor Annuity In the case of a Qualified Joint and Survivor Annuity, the Plan Administrator shall, no less than 30 days, and no more than 90 days prior to the Annuity Starting date, provide each Participant a written explanation of:

- (a) the terms and conditions of a Qualified Joint and Survivor Annuity,
- (b) the Participant's right to make and the effect of an election to waive the Qualified Joint and Survivor Annuity form of benefit,
- (c) the rights of a Participant's Spouse, and

ş

(d) the right to make, and the effect of, a revocation of a previous election to waive the Qualified Joint and Survivor Annuity.

The Participant and the Participant's Spouse may consent to waiving the minimum 30 day notice period described above and may receive notice no less than 7 days prior to the Annuity Starting Date, provided that:

- (e) the Plan Administrator clearly informs the Participant and the Participant's Spouse that they have a right to a period of at least 30 days after receiving the notice to consider the decision of whether or not to elect a distribution (and, if applicable a particular distribution option), and
- (f) the Participant and the Participant's Spouse, after receiving the notice, affirmatively elect a distribution.

8.6 Notice Requirements for Qualified Pre-Retirement Survivor Annuity In the case of a Qualified Pre-Retirement Survivor Annuity as described in paragraph 8.3, the Plan Administrator shall provide each Participant within the applicable period for such Participant a written explanation of the Qualified Pre-Retirement Survivor Annuity in such terms and in such manner as would be comparable to the explanation provided for meeting the requirements of paragraph 8.5 applicable to a Qualified Joint and Survivor Annuity. The applicable period for a Participant is whichever of the following periods ends at the latest date:

- (a) the period beginning with the first day of the Plan Year in which the Participant attains age 32 and ending with the close of the Plan Year preceding the Plan Year in which the Participant attains age 35,
- (b) *a* reasonable period ending after the individual becomes a Participant,
- (c) a reasonable period ending after this Article first applies to the Participant. Notwithstanding the foregoing, notice must be provided within a reasonable period ending after separation from Service in the case of a Participant who separates from Service before attaining age 35.

For purposes of applying the preceding paragraph, a reasonable period ending after the events described in (b) and (c) is the end of the two-year period beginning one year prior to the date the applicable event occurs, and ending one year after that date. In the case of a Participant who separates from Service before the Plan Year in which age 35 is attained, notice shall be provided within the two-year period beginning one year prior to separation and ending one year after separation. If such a Participant subsequently returns to employment with the Employer, the applicable period for such Participant shall be redetermined.

8.7 Automatic Joint and Survivor Annuity and Early Survivor Annuity Any Participant who has elected pursuant to paragraph 8.8(b) and any Participant who does not elect under paragraph 8.8(a) or who meets the requirements of paragraph 8.8(a), except that such Participant does not have at least 10 years of vesting Service

when he or she separates from Service, shall have his or her benefits distributed in accordance with all of the following requirements if benefits would have been payable in the form of a life annuity.

- (a) Automatic Joint and Survivor Annuity If benefits in the form of a life annuity become payable to a married Participant who:
 - (1) begins to receive payments under the Plan on or after Normal Retirement Age, or
 - (2) dies on or after Normal Retirement Age while still working for the Employer, or
 - (3) begins to receive payments on or after the Qualified Early Retirement Age, or
 - (4) separates from Service on or after attaining Normal Retirement Age (or the Qualified Early Retirement Age) and after satisfying the eligibility requirements for the payment of benefits under the Plan and thereafter dies before beginning to receive such benefits.

Such benefits will be received under this Plan in the form of a Qualified Joint and Survivor Annuity, unless the Participant has elected otherwise during the Election Period. The Election Period must begin at least 6 months before the Participant attains Qualified Early Retirement Age and end not more than 90 days before the commencement of benefits. Any election will be in writing and may be changed by the Participant at any time.

- (b) Election of Early Survivor Annuity A Participant who is employed after attaining the Qualified Early Retirement Age will be given the opportunity to elect, during the Election Period, to have a survivor annuity payable on death. If the Participant elects the survivor annuity, payments under such annuity must not be less than the payments which would have been made to the Spouse under the Qualified Joint and Survivor Annuity if the Participant had retired on the day before his or her death. Any election under this provision will be in writing and may be changed by the Participant at any time. The Election Period begins on the later of:
 - (1) the 90th day before the Participant attains the Qualified Early Retirement Age, or
 - (2) the date on which participation begins, and ends on the date the Participant terminates employment.

8.8 Annuity Contracts Any annuity contract distributed under this Plan must be nontransferable. The terms of any annuity contract purchased and distributed by the Plan to a Participant or Spouse shall comply with the requirements of this Plan.

ARTICLE IX VESTING

9.1 *Employee Contributions* A Participant shall always have a 100% vested and nonforfeitable interest in his or her Elective Deferrals, Qualified Non-Elective Contributions, Qualified Matching Contributions, Rollover Contributions, and Transfer Contributions plus the earnings thereon. No forfeiture of Employer related contributions (including any minimum contributions made under paragraph 15.2) will occur solely as a result of an Employee's withdrawal of any Employee contributions.

9.2 *Employer Contributions* A Participant shall acquire a vested and nonforfeitable interest in his or her account attributable to Employer contributions in accordance with the table selected in the Adoption Agreement, provided that if a Participant is not already fully vested, he or she shall become so upon attaining Normal Retirement Age, Early Retirement Age, on death prior to normal retirement (provided the Participant has not terminated employment prior to death), on retirement due to Disability, or on termination of the Plan.

9.3 Computation Period The computation period for purposes of determining Years of Service and Breaks in Service for purposes of computing a Participant's nonforfeitable right to his or her account balance derived from Employer contributions shall be determined by the Employer in the Adoption Agreement. In the event a former Participant with no vested interest in his or her Employer contribution account requalifies for participation in the Plan after incurring a Break in Service, such Participant shall be credited for vesting with all pre-break and postbreak Service.

9.4 Re-qualification prior to Five Consecutive One-Year Breaks in Service Subject to Article VI, the account balance of a Participant who is re-employed prior to incurring five consecutive one-year Breaks in Service shall consist of any undistributed amount in his or her account as of the date of re-employment plus any future contributions added to such account plus the investment earnings on the account. The Vested Account Balance of such Participant shall be determined by multiplying the Participant's account balance (adjusted to include any distribution or redeposit made under paragraph 6.3) by such Participant's vested percentage. All Service of the Participant, both prior to and following the break, shall be counted when computing the Participant's vested percentage.

9.5 Re-qualification after Five Consecutive One-Year Breaks in Service Subject to Article VI, if a Participant was not fully vested prior to termination of employment and is re-employed after incurring five consecutive oneyear Breaks in Service, a new account shall be established for such Participant to separate his or her deferred vested and nonforfeitable account, if any, from the account to which new allocations will be made. The Participant's deferred account to the extent remaining shall be fully vested and shall continue to share in earnings and losses of the Trust. When computing the Participant's vested portion of the new account, all pre-break and post-break Service shall be counted. However, notwithstanding this provision, no such former Participant who has had five consecutive one-year Breaks in Service shall acquire a larger vested and nonforfeitable interest in his or her prior account balance as a result of requalification hereunder.

9.6 Calculating Vested Interest A Participant's vested and nonforfeitable interest shall be calculated by multiplying the fair market value of his or her account attributable to Employer contributions on the Valuation Date concurrent with or preceding distribution by the decimal equivalent of the vested percentage as of his or her termination date. The amount attributable to Employer contributions for purposes of the calculation includes amounts previously paid out pursuant to paragraph 6.3 and not repaid. The Participant's vested and nonforfeitable interest, once calculated above, shall be reduced to reflect those amounts previously paid out to the Participant and not repaid by the Participant. The Participant's vested and nonforfeitable interest so determined shall continue to share in the investment earnings and any increase or decrease in the fair market value of the Trust up to the Valuation Date preceding or coinciding with payment.

9.7 Forfeitures Any balance in the account of a Participant who has separated from Service to which he or she is not entitled under the foregoing provisions, shall be forfeited and applied as provided in the Adoption Agreement. The reallocation or other disposition of a nonvested benefit may only occur if the Participant has received payment of his or her entire vested benefit from the Plan or if the Participant has incurred five consecutive one-year Breaks in

Plan Document

Service. While awaiting reallocation or other disposition, the Plan Administrator or his designate, if applicable, shall have the right to leave the nonvested benefit in the Participant's account or may transfer the nonvested benefit to a forfeiture holding account. Amounts held in a forfeiture suspense account may share in any increase or decrease in fair market value of the assets of the Trust in accordance with Article V of the Plan. Such determination shall be made by the Plan Administrator or his designate, if applicable. For purposes of this paragraph, if the value of a Participant's Vested Account Balance is zero, the Participant shall be deemed to have received a distribution of such Vested Account Balance. A Highly Compensated Employee's Matching Contributions may be forfeited, even if vested, if the contributions to which they relate are Excess Deferrals, Excess Contributions or Excess Aggregate Contributions. Benefits with respect to Participants who cannot be located as provided at paragraph 7.15 hereof will be treated in the same manner as a forfeiture.

9.8 Amendment of Vesting Schedule No amendment to the Plan shall have the effect of decreasing a Participant's vested interest determined without regard to such amendment as of the later of the date such amendment is adopted or the date it becomes effective. Further, if the vesting schedule of the Plan is amended, or the Plan is amended in any way that directly or indirectly affects the computation of any Participant's nonforfeitable percentage or if the Plan is deemed amended by an automatic change to or from a Top-Heavy vesting schedule, each Participant with at least three Years of Service with the Employer may elect, during the election period defined herein, to have his or her nonforfeitable percentage computed under the Plan without regard to such amendment. For Participants who do not have at least one Hour of Service in any Plan Year beginning after 1988, the preceding sentence shall be applied by substituting "five Years of Service" for "three Years of Service" where such language appears. The period during which the election may be made shall commence with the date the amendment is adopted and shall end on the later of:

- (a) 60 days after the amendment is adopted,
- (b) 60 days after the amendment becomes effective, or
- (c) 60 days after the Participant is issued written notice of the amendment by the Employer or the Trustee.

If the Trustee notifies the Participants involved, the Plan may be charged for the costs thereof.

No amendment to the Plan shall be effective to the extent that it has the effect of decreasing a Participant's accrued benefit. Notwithstanding the preceding sentence, a Participant's account balance may be reduced to the extent permitted under Code Section 412(c)(8) relating to financial hardships. For purposes of this paragraph, a Plan amendment which has the effect of decreasing a Participant's account balance or eliminating an optional form of benefit, with respect to benefits attributable to Service before the amendment, shall be treated as reducing an accrued benefit.

9.9 Service with Controlled Groups All Years of Service with other members of a controlled group of corporations [as defined in Code Section 414(b)], trades or businesses under common control [as defined in Code Section 414(c)], or members of an affiliated service group [as defined in Code Section 414(m)] shall be considered for purposes of determining a Participant's nonforfeitable percentage.

9.10 Compliance with Uniformed Service Employment and Reemployment Rights Act of 1994 Notwithstanding any provision of this Plan to the contrary, Years of vesting Service will be credited to Participants with respect to periods of qualified military service as provided in Code Section 414(u).

ARTICLE X

LIMITATIONS ON ALLOCATIONS

10.1 Participation in this Plan Only If the Participant does not participate in and has never participated in another qualified plan, a Welfare Benefit Fund, individual medical account as defined in Code Section 415(1)(2), or a Simplified Employee Pension Plan maintained by the adopting Employer, which provides an Annual Addition, the amount of Annual Additions which may be credited to the Participant's account for any Limitation Year will not exceed the lesser of the Maximum Permissible Amount or any other limitation contained in this Plan. If the Employer contribution that would otherwise be contributed or allocated to the Participant's account would cause the Annual Additions for the Limitation Year to exceed the Maximum Permissible Amount, the amount contributed or allocated will be reduced so that the Annual Additions for the Limitation Year will equal the Maximum Permissible Amount. Prior to determining the Participant's actual Compensation for the Limitation Year, the Employer may determine the Maximum Permissible Amount for a Participant on the basis of a reasonable estimate of the Participant's Compensation for the Limitation Year, uniformly determined for all Participants similarly situated. As soon as is administratively feasible after the end of the Limitation Year, the Maximum Permissible Amount for the Limitation Year will be determined on the basis of the Participant's actual Compensation for the Limitation Year.

10.2 Disposition of Excess Annual Additions If there is an Excess Amount due to an error in estimating a Participant's Compensation for a Limitation Year under paragraph 10.1 or as a result of the allocation of forfeitures there is an Excess Amount, the excess will be distributed to the affected Participant in the order which follows:

- (a) Return Voluntary Contributions plus the investment earnings thereon.
- (b) Return Required Contributions plus the investment earnings thereon.
- (c) Return Elective Deferrals plus the investment earnings thereon to non-Highly Compensated Employees to the extent that such return will not cause a failure of either the ADP or ACP Test.
- (d) Return unmatched Elective Deferrals plus the investment earnings thereon to Highly Compensated Employees.

If an Excess Amount remains after completing the above steps, the Plan Administrator will, as elected in the Adoption Agreement, either forfeit the remaining excess in accordance with the "spillover method" or hold the excess in a "suspense account" and allocate the excess to the Participant in the Limitation Year following the Limitation Year in which the excess occurred. If a Participant has terminated employment prior to the end of the current Limitation Year, any Excess Amount will be forfeited and applied to reduce future Employer contributions for all remaining Participants in the next Limitation Year, and each succeeding Limitation Year pursuant to this paragraph, it will not participate in the allocation of investment gains and losses. If a suspense account is in existence at any time during a Limitation to Participant accounts before any Employer contributions or any Employee contributions may be made to the Plan for that Limitation Year.

10.3 Participation in Multiple Defined Contribution Plans The Annual Additions which may be credited to a Participant's account under this Plan for any Limitation Year will not exceed the Maximum Permissible Amount. With respect to this Plan, the Maximum Permissible Amount is reduced by the Annual Additions credited to a Participant's account under the other qualified Master or Prototype Defined Contribution Plans, Welfare Benefit Funds, individual medical accounts, as defined in Code Section 415(1)(2), and Simplified Employee Pension Plans maintained by the Employer, which provide an Annual Addition for the same Limitation Year. If the Annual Additions with respect to the Participant under other Defined Contribution Plans, Welfare Benefit Funds, individual medical accounts and Simplified Employee Pension Plans maintained by the Employer are less than the Maximum Permissible Amount and the Employer contribution that would otherwise be contributed or allocated to the Participant's account under this Plan would cause the Annual Additions for the Limitation Year to exceed this limitation, the amount contributed or allocated under this Plan will be reduced so that the Annual Additions under all such plans and funds for the Limitation Year will equal the Maximum Permissible Amount. If the Annual Additions with respect to the Participant under such other Defined Contribution Plans and Welfare Benefit Funds in the aggregate are equal to or greater than the Maximum Permissible Amount, no amount will be contributed or allocated to the Participant's account under this Plan for the Limitation Year. Prior to determining the Participant's actual Compensation for the Limitation Year, the Employer may determine the Maximum Permissible Amount for a Participant in the manner described in paragraph 10.1. As soon as administratively feasible after the end of the Limitation Year, the Maximum Permissible Amount for the Limitation Year, the Compensation for the Limitation Year. If the Participant is covered under another qualified Defined Contribution Plan maintained by the Employer which is not a Master or Prototype Plan, Annual Additions which may be credited to the Participant's account under this Plan for any Limitation Year will be limited in accordance with this paragraph as though the other plan were a Master or Prototype Plan unless the Employer specifies other limitations in the Adoption Agreement.

10.4 Disposition of Excess Annual Additions under Two Plans If a Participant's Annual Additions under this Plan and such other plans would result in an Excess Amount for a Limitation Year due to an error in estimating a Participant's Compensation for a Limitation Year under paragraph 10.3 or as a result of forfeitures, a Participant's Annual Additions under this Plan and such other plans would result in an Excess Amount for a Limitation Year, the Excess Amount will be deemed to consist of the Annual Additions last allocated except that Annual Additions attributable to a Simplified Employee Pension Plan will be deemed to have been allocated first and then Annual Additions to a Welfare Benefit Fund or individual medical account as defined in Code Section 415(1)(2) will be deemed to have been allocated next regardless of the actual allocation date. If an Excess Amount was allocated to a Participant on an Allocation Date of this Plan which coincides with an allocation date of another plan, the Excess Amount attributed to this Plan will be the product of:

- (a) the total Excess Amount allocated as of such date, times
- (b) the ratio of:
 - (1) the Annual Additions allocated to the Participant for the Limitation Year as of such date under this Plan, to
 - (2) the total Annual Additions allocated to the Participant for the Limitation Year as of such date under this and all the other qualified Master or Prototype Defined Contribution Plans.

Any Excess Amount attributed to this Plan will be disposed of in the manner described in paragraph 10.2.

10.5 Limitations on Benefits For any Limitation Year beginning before January 1, 2000, the sum of a Participant's Defined Benefit Plan Fraction and Defined Contribution Plan Fraction will not exceed 1.0. For any Limitation Year beginning before January 1, 2000 during which the Plan is Top-Heavy, the Defined Benefit and Defined Contribution Plan Fractions shall be calculated in accordance with Code Section 416(h). The Annual Additions which may be credited to the Participant's account under this Plan for any Limitation Year will be limited in accordance with the provisions attached to the Adoption Agreement.

10.6 Participation in this Plan and a Defined Benefit Plan If the Employer maintains, or at any time maintained, a qualified Defined Benefit Plan (other than Paired Plan #02001, #02002, #02003, #02004) covering any Participant in this Plan, the sum of the Participant's Defined Benefit Plan Fraction and Defined Contribution Plan Fraction will not exceed 1.0 in any Limitation Year. For any Plan Year during which the Plan is Top-Heavy, the Defined Benefit and Defined Contribution Plan Fractions shall be calculated in accordance with Code Section 416(h). The Annual Additions which may be credited to the Participant's account under this Plan for any Limitation Year will be limited in accordance with the Adoption Agreement.

ARTICLE XI

ANTIDISCRIMINATION TESTING

11.1 General Testing Requirements With respect to each Plan Year, an Employer's Plan which offers a Code Section 401(k) cash or deferred arrangement and any contributions made thereunder must satisfy the Average Deferral Percentage Test (ADP Test) and, if applicable, the Average Contribution Percentage Test (ACP Test). Under each of these tests, the Average Deferral Percentage and Average Contribution Percentage for Highly Compensated Employees may not exceed the Average Deferral Percentage and Average Contribution Percentage for non-Highly Compensated Employees by more than the amount permitted by application of the basic limit or the alternative limit. These limits are described at paragraph 11.2 hereof. If the Average Deferral Percentage or Average Contribution Percentage for Highly Compensated Employees must be reduced to the maximum permitted under the most liberal limit. The reduction in the average is determined in accordance with paragraph 11.3 hereof.

The Employer may elect to make an additional Qualified Non-Elective Contribution (QNEC) and/or a Qualified Matching Contribution (QMAC) for non-Highly Compensated Employees to increase their Average Deferral Percentage and/or Average Contribution Percentage to the point where the Plan satisfies the ADP and/or the ACP Test. These qualified contributions are described at paragraph 11.4 hereof.

If the Plan can only satisfy the ADP Test and the ACP Test by application of the alternative limit, the Plan must apply the aggregate limit as described at paragraph 11.2 hereof. If the Plan fails to satisfy the aggregate limit, the Employer must either make correcting distributions to affected Highly Compensated Employees or make QNEC and/or QMAC contributions for non-Highly Compensated Employees. The Employer may instead elect to make an additional Qualified Non-Elective contribution (QNEC) and/or Qualified Matching Contribution (QMAC) to increase the Average Deferral Percentage and/or Average Contribution Percentage of non-Highly Compensated Employees to the point where the Plan satisfies the ADP and/or ACP Test. The Employer may allocate correcting distributions and/or qualified contributions between amounts included in the ADP Test or the ACP Test.

11.2 Testing Limitations

...

- (a) **Prior Year Testing** The Actual Deferral Percentage (hereinafter "ADP") for a Plan Year for Participants who are Highly Compensated Employees for each Plan Year and the prior year's ADP for Participants who were non-Highly Compensated Employees for the prior Plan Year must satisfy one of the following tests:
 - (1) The ADP for a Plan Year for Participants who are Highly Compensated Employees for the Plan Year shall not exceed the prior year's ADP for Participants who were non-Highly Compensated Employees for the prior Plan Year multiplied by 1.25: or
 - (2) The ADP for a Plan Year for Participants who are Highly Compensated Employees for the Plan Year shall not exceed the prior year's ADP for Participants who were non-Highly Compensated Employees for the prior Plan Year multiplied by 2.0, provided that the ADP for Participants who are Highly Compensated Employees does not exceed the ADP for Participants who were non-Highly Compensated Employees in the prior Plan Year by more than 2 percentage points.

For the first Plan Year in a Plan which permits a Participant to make Elective Deferrals and is not a successor Plan, for purposes of the foregoing tests, the prior year's non-Highly Compensated Employees' ADP shall be 3% or the actual ADP if greater, unless the Employer has elected in the Adoption Agreement to use the Plan Year's ADP for these Participants. (b) Current Year Testing - If elected by the Employer in the Adoption Agreement, the ADP tests in 1 and 2, above, will be applied by comparing the current Plan Year's ADP for Participants who are Highly Compensated Employees with the current Plan Year's ADP for Participants who are non-Highly Compensated Employees. Once made, this election can only be undone if the Plan meets the requirements for changing to prior year testing set forth in Notice 98-1 (or superseding guidance).

11.3 Calculation and Distribution of Excess Contributions and Excess Aggregate Contributions

- (a) Reducing the Average for Highly Compensated Employees The Average Deferral Percentage and/or Average Contribution Percentage for Highly Compensated Employees is reduced to the maximum allowed by the applicable limit at paragraph 11.2. The average is reduced on a step-bystep leveling basis beginning by reducing the Actual Deferral Percentage or the Actual Contribution Percentage for the Highly Compensated Employee with the highest percentage until the average is reduced to the maximum allowed or until the Actual Deferral Percentage or Actual Contribution Percentage for such Highly Compensated Employee is lowered to that of the Highly Compensated Employee with the next highest percentage. This process continues until the Average Deferral Percentage and/or the Average Contribution Percentage is lowered to the maximum allowed for the Plan Year. The excess dollar amount attributable to each affected Highly Compensated Employee is then totaled for purposes of correcting distributions determined at paragraph (b) below.
- Correcting Distributions to Highly Compensated Employees The total amount to be distributed (b) as determined under paragraph (a) is allocated to Highly Compensated Employees on the basis of the dollar amount included for such Employee in the numerator of the Actual Deferral Percentage or the Actual Contribution Percentage, as applicable. The distribution for each affected Highly Compensated Employee is determined on a leveling basis similar to that described at paragraph (a) except that the process is based on dollars rather than percentages. Excess Contributions are allocated to the Highly Compensated Employees with the largest amount of Employer contributions taken into account in calculating the Average Deferral Percentage or Average Contribution Percentage test for the year in which the excess arose, beginning with the Highly Compensated Employee with the largest amount of such Employer contributions and continuing in descending order until all the Excess Contributions have been allocated. For purposes of the preceding sentence, the "largest amount" is determined after distribution of any Excess Contribution. After correcting distributions are allocated, it is not necessary to recompute the Highly Compensated Employee averages to determine if they satisfy the ADP Test and/or the ACP Test. Distributions of Excess Contributions and Excess Aggregate Contributions are to be made in accordance with paragraphs 7.12 and 7.13 hereof.

11.4 Qualified Non-Elective and/or Matching Contributions If provided in the Adoption Agreement, the Employer may make a QNEC or QMAC for non-Highly Compensated Employees to increase the Average Deferral Percentage and/or Average Contribution Percentage to the point where the Plan passes the ADP Test and/or the ACP Test. The following rules apply with respect to such contributions:

- (a) A QNEC or QMAC used in the ADP Test may not also be included in the ACP Test.
- (b) If the Employer elects to test on the basis of current year data, QNECs and/or QMACs must be made and credited to Participant accounts not later than the last day of the 12-consecutive month period following the end of the Plan Year being tested.
- (c) If the Employer elects to test on the basis of prior year data for non-Highly Compensated Employees, QNECs and/or QMACs for such Employees must be contributed not later than the last day of the Plan Year being tested.

11.5 Special Rules Relating to Application of the ADP Test A Participant is a Highly Compensated Employee for a particular Plan Year if he or she meets the definition of a Highly Compensated Employee in effect for that Plan Year. Similarly, a Participant is a non-Highly Compensated Employee for a particular Plan Year if he or she does not meet the definition of a Highly Compensated Employee in effect for that Plan Year.

- (a) The Actual Deferral Percentage for any Participant who is a Highly Compensated Employee for the Plan Year and who is eligible to have Elective Deferrals (and Qualified Non-Elective Contributions or Qualified Matching Contributions, or both, if treated as Elective Deferrals for purposes of the ADP test) allocated to his or her accounts under two or more arrangements described in Code Section 401(k), that are maintained by the Employer, shall be determined as if such Elective Deferrals (and, if applicable, such Qualified Non-Elective Contributions or Qualified Matching Contributions, or both) were made under a single arrangement. If a Highly Compensated Employee participates in two or more cash or deferred arrangements that have different Plan Years, all cash or deferred arrangements ending with or within the same calendar year shall be treated as a single arrangement. Notwithstanding the foregoing, certain plans shall be treated as separate if mandatorily disaggregated under regulations under Code Section 401(k).
- (b) In the event that this Plan satisfies the requirements of Code Sections 401(k), 401(a)(4), or 410(b) only if aggregated with one or more other plans, or if one or more other plans satisfy the requirements of such Code Sections only if aggregated with this Plan, then this section shall be applied by determining the Actual Deferral Percentage of Employees as if all such plans were a single plan. Any adjustments to the non-Highly Compensated Employee ADP for the prior year will be made in accordance with Notice 98-1 and any superseding guidance, unless the Employer has elected in the Adoption Agreement to use the current year testing method. Plans may be aggregated in order to satisfy Code Section 401(k) only if they have the same Plan Year and use the same ADP testing method.
- (c) The Employer shall maintain records sufficient to demonstrate satisfaction of the ADP test and the amount of Qualified Non-Elective Contributions or Qualified Matching Contributions, or both, used in such test.
- (d) The determination and treatment of the Actual Deferral Percentage amounts of any Participant shall satisfy such other requirements as may be prescribed by the Secretary of the Treasury.
- (e) Solely for the first Plan Year to which paragraph 11.1 applies, the Employer can elect in the Adoption Agreement to determine the ADP of the non-Highly Compensated Employees to be the actual ADP determined for the first Plan Year or 3%.
- (f) For purposes of determining the ADP test, Elective Deferrals, Qualified Non-Elective Contributions and Qualified Matching Contributions must be made before the end of the 12month period immediately following the Plan Year to which the contributions relate.

11.6 **Recharacterization** If the Employer allows for Voluntary Contributions in the Adoption Agreement, a Participant may treat his or her Excess Contributions as an amount distributed to the Participant and then contributed by the Participant to the Plan. Recharacterized amounts will remain nonforfeitable and subject to the same distribution requirements as Elective Deferrals. Amounts may not be recharacterized by a Highly Compensated Employee to the extent that such amount in combination with other Employee contributions made by that Employee would exceed any stated limit under the Plan on Voluntary Contributions. Recharacterization must occur no later than two and one-half months after the last day of the Plan Year in which such Excess Contributions arose and is deemed to occur no earlier than the date the last Highly Compensated Employee is informed in writing of the amount recharacterized and the consequences thereof. Recharacterized amounts will be taxable to the Participant for the Participant's tax year in which the Participant would have received them in cash. 11.7 Special Rules Relating to Application of the ACP Test Employee contributions and Matching Contributions must meet the nondiscrimination requirements of Code Section 401(a)(4) and the Average Contribution Percentage (hereinafter ACP) test of Code Section 401(m). If Employee contributions (including any Elective Deferrals recharacterized as Voluntary Contributions) or Matching Contributions are made in connection with a cash or deferred arrangement, the ACP is in addition to the ADP under Code Section 401(k). Qualified Matching Contributions and Qualified Non-Elective Contributions used to satisfy the ADP test may not be used to satisfy the ACP test.

- (a) **Prior Year Testing** The ACP for a Plan Year for Participants who are Highly Compensated Employees for each Plan Year and the prior year's ACP for Participants who were non-Highly Compensated Employees for the prior Plan Year must satisfy one of the following tests
 - (1) The ACP for a Plan Year for Participants who are Highly Compensated Employees for the Plan Year shall not exceed the prior year's ACP for Participant's who were non-Highly Compensated Employees for the prior Plan Year multiplied by 1.25; or
 - (2) The ACP for a Plan Year for Participants who are Highly Compensated Employees for the Plan Year shall not exceed the prior year's ACP for Participants who were non-Highly Compensated Employees for the prior Plan Year multiplied by 2, provided that the ACP for Participants who are Highly Compensated Employees does not exceed the ACP for Participants who were non-Highly Compensated Employees in the prior Plan Year by more than 2 percentage points.

For the first Plan Year, this Plan permits any Participant to make Employee contributions, provides for Matching Contributions or both, and this is not a successor Plan, for purposes of the foregoing tests, the prior year's non-Highly Compensated Employee's ACP shall be 3% unless the Employer has elected in the Adoption Agreement to use the Plan Year's ACP for these Participants.

- (b) Current Year Testing If elected by the Employer in the Adoption Agreement, the ACP tests in 1 and 2, above, will be applied by comparing the current Plan Year's ACP for Participants who are Highly Compensated Employees for each Plan Year with the current Plan Year's ACP for Participants who are non-Highly Compensated Employees. Once made, this election can only be undone if the Plan meets the requirements for changing to prior year testing set forth in Notice 98-1 (or superseding guidance).
- (c) A Participant is a Highly Compensated Employee for a particular Plan Year if he or she meets the definition of a Highly Compensated Employee in effect for that Plan Year. Similarly, a Participant is a non-Highly Compensated Employee for a particular Plan Year if he or she does not meet the definition of Highly Compensated Employee in effect for that Plan Year.
- (d) If one or more Highly Compensated Employees participate in both a cash or deferred arrangement and a plan subject to the ACP test maintained by the Employer and the sum of the ADP and ACP of those Highly Compensated Employees subject to either or both tests exceeds the Aggregate Limit, then the ADP or ACP of those Highly Compensated Employees who also participate in a cash or deferred arrangement will be reduced in accordance with paragraph 11.3 so that the limit is not exceeded. The amount by which each Highly Compensated Employee's Contribution Percentage Amounts is reduced shall be treated as an Excess Aggregate Contribution. The ADP and ACP of the Highly Compensated Employees are determined after any corrections required to meet the ADP and ACP tests and are deemed to be the maximum permitted under such tests for the Plan Year. Multiple use does not occur if either the ADP and ACP of the Highly Compensated Employee does not exceed 1.25 multiplied by the ADP and ACP of the non-Highly Compensated Employees.

Plan Document

- (e) For purposes of this Article, the Contribution Percentage for any Participant who is a Highly Compensated Employee and who is eligible to have Contribution Percentage Amounts allocated to his or her account under two or more plans described in Code Section 401(a) or arrangements described in Code Section 401(k) that are maintained by the Employer, shall be determined as if the total of such Contribution Percentage Amounts was made under each plan. If a Highly Compensated Employee participates in two or more cash or deferred arrangements that have different Plan Years, all cash or deferred arrangements ending with or within the same calendar year shall be treated as a single arrangement. Notwithstanding the foregoing, certain plans shall be treated as separate if mandatory disaggregation under the Regulations issued under Code Section 401(k) apply.
- (f) In the event that this Plan satisfies the requirements of Code Sections 401(a)(4), 401(m), or 410(b) only if aggregated with one or more other plans, or if one or more other plans satisfy the requirements of such Code Sections only if aggregated with this Plan, then this section shall be applied by determining the Contribution Percentage of Employees as if all such plans were a single plan. Any adjustments to the non-Highly Compensated Employee ACP for the prior year will be made in accordance with Notice 98-1 and any superseding guidance, unless the Employer has elected in the Adoption Agreement to use the current year testing method. Plans may be aggregated in order to satisfy Code Section 401(m) only if the aggregated plans have the same Plan Year and use the same ACP testing method.
 - (g) For purposes of determining the Contribution Percentage test, Employee contributions are considered to have been made in the Plan Year in which contributed to the Plan. Matching Contributions and Qualified Non-Elective Contributions will be considered made for a Plan Year if made no later than the end of the twelve-month period beginning on the day after the close of the Plan Year.
 - (h) The Employer shall maintain records sufficient to demonstrate satisfaction of the ACP test and the amount of Qualified Non-Elective Contributions or Qualified Matching Contributions, or both, used in such test.
 - (i) The determination and treatment of the Contribution Percentage of any Participant shall satisfy such other requirements as may be prescribed by the Secretary of the Treasury.
 - (j) Solely for the first Plan Year to which paragraph 11.1 applies, the Employer can elect in the Adoption Agreement to determine the ACP of the non-Highly Compensated Employees to be the actual ACP determined for the first Plan Year or 3%.

52

ARTICLE XII

ADMINISTRATION

12.1 Plan Administrator The Employer shall be the named fiduciary and serve as Plan Administrator unless an individual or other entity (excluding the Trustee) is named to serve in such capacity. The Plan Administrator may appoint or allocate the duties of the Plan Administrator among several individuals or entities. The Plan Administrator's duties shall include:

- (a) appointing the Plan's attorney, accountant, Recordkeeper, actuary, Trustee, Custodian, investment manager, or any other party needed to administer the Plan,
- (b) directing the appropriate party with respect to payments from the Trust,
- (c) communicating with Employees regarding their participation and benefits under the Plan, including the administration of all claims procedures,
- (d) maintaining all necessary records for the administration of the Plan and filing any returns and reports with the Internal Revenue Service, Department of Labor, or any other governmental agency,
- (e) reviewing and approving any financial reports, investment reviews, or other reports prepared by any party appointed by the Employer under paragraph (a),
- (f) establishing a funding policy and investment objectives consistent with the purposes of the Plan and the Employee Retirement Income Security Act of 1974,
- (g) construing and resolving any question of Plan interpretation and questions of fact. The Plan Administrator's interpretation of Plan provisions and resolution of questions of facts including eligibility and benefits under the Plan, is final, and unless it can be shown to be arbitrary and capricious will not be subject to "de novo" review;
- (h) monitoring the activities of the Trustee and the performance of and making changes when necessary to the portfolio of the Plan,
- (i) obtaining a legal determination of the qualified status of all domestic relations orders and complying with the requirements of the law with regard thereto, and
- (j) ensuring that any and all loans made by the Plan are in compliance with the requirements of the Internal Revenue Code and the regulations issued thereunder, and the regulations issued by the Department of Labor.

12.2 **Trustee** Unless the Employer retains investment management responsibility or delegates investment management responsibility to an investment manager or to Plan Participants with respect to all or a portion of the Plan's assets, the Trustee shall be responsible for the investment management of the Trust.

(a) As investment manager, the Trustee shall invest the Trust in accordance with the Plan's investment policy statement and the investment alternatives permitted at paragraph 14.3 hereof. The Trustee shall also perform custodial functions described at paragraph 12.3 hereof for the Trust with respect to Plan assets over which the Trustee has investment management responsibility. The Trustee may also perform custodial functions for the Trust with respect to Plan assets the Trustee does not manage, to the extent agreed to between the Trustee and the Employer, if the Trustee is appointed Custodian for some or all of such assets in accordance with the terms of the Plan. The Trustee may execute any additional documents as required which shall be treated as an addendum to this Basic Plan Document #01. No such agreement may conflict with any provision of this

Basic Plan Document #01. In addition, any provision of such an agreement which would jeopardize the tax-qualified status of the Plan and Trust shall be null and void. The Trustee's administrative duties shall be limited to those agreed to between the parties. The Employer or his designate shall be responsible for other administrative duties required under the Plan or by applicable law.

Within 90 days following the close of each Plan Year, or at such other times as may be agreed to (b) between the Employer and the Trustee, and within 90 days following its removal or resignation, the Trustee shall file with the Employer a report of that part of the Trust under the investment management of the Trustee during such year or from the end of the preceding Plan Year to the date of removal or resignation. Such report shall include a statement of receipts and disbursements, the net income, or loss of the Trust, the gains or losses realized by the Trust upon sale or other disposition of the assets, the increase, or decrease, in the value of the Trust, all payments and distributions made from the Trust since the date of its last report and shall contain a schedule of assets listing the fair market value of investments held in the Trust as of the end of the Plan Year or the date of removal or resignation, as applicable. The fair market value of investments for which there is a ready market shall be determined using the most recent price quoted on a national or other recognized securities exchange or over- the-counter market. The fair market value of illiquid investments shall be obtained by a valuation performed by an independent appraiser appointed by the Trustee for this purpose. His determination shall be final. The Employer shall review the Trustee's report and notify the Trustee in the event of its disapproval of the report within 90 days, providing the Trustee with a written description of the items in question. The Trustee shall have 60 days to provide the Employer with a written explanation of the items in question. If the Employer again disapproves, the Trustee shall have the right to file its report in a court of competent jurisdiction for audit and adjudication. In the event the Employer fails to file a written objection to the Trustee's report within the 90 day period following receipt of the report, the Employer shall be deemed to have approved the report. In such case, the Trustee shall be released and discharged with respect to all matters contained in the report.

12.3 Administrative Fees and Expenses All reasonable fees, charges and expenses incurred by the Trustee in connection with the administration of the Trust and all reasonable fees, charges and expenses incurred by the Plan Administrator in connection with the administration of the Plan (including such reasonable compensation to the Trustee and the Plan Administrator as may be agreed upon from time to time between the Employer, the Trustee or the Employer and Plan Administrator) and fees for legal services rendered to the Trustee or Plan Administrator shall be paid from the Plan unless:

- (a) The payment of such expense would constitute a "prohibited transaction" within the meaning of ERISA Section 406 or Code Section 4975.
- (b) The Employer actually pays such expenses directly. Any and all reasonable additional administrative expenses incurred to effect investment directives made by the Participants and by each Designated Beneficiary under this Plan shall be paid by the Trust and as determined by the Employer shall either be charged (in accordance with such reasonable nondiscriminatory rules as the Employer deems appropriate under the circumstances) to the account of the individual issuing such directive, or treated as a general expense of the Trust. Notwithstanding the foregoing, nothing in this section shall prevent the Employer from paying such administrative expenses directly.
- (c) All transaction related expenses incurred to effect a specific investment for a Participant directed account (such as brokerage commissions and other transaction related expenses), shall, as determined by the Employer, either be paid from or otherwise be charged directly to the account of the Participant providing such direction or treated as a general expense of the Trust.
- (d) If there are insufficient liquid assets of the Trust to cover the fees of the Trustee or the Employer then assets of the Trust shall be liquidated to the extent necessary to cover fees.

- (e) Notwithstanding the foregoing, no compensation other than reimbursement for expenses incurred shall be paid to a Plan Administrator who is the Employer or Employee of the Employer.
- (f) In the event any part of the Plan becomes subject to tax, all taxes incurred will be paid from the Plan at the direction of the Plan Administrator.

12.4 Limitation on Liability and Indemnification

- (a) The Trustee shall have the authority and discretion to manage and govern the Plan to the extent provided in this instrument, but does not guarantee the Plan in any manner against investment loss or depreciation in asset value, or guarantee the adequacy of the Trust to meet and discharge all or any liabilities of the Plan.
- (b) The Trustee shall not be liable for the making, retention or sale of any investment or reinvestment made by it, as herein provided, or for any loss to, or diminution of the Trust, or for any other loss or damage which may result from the discharge of its duties hereunder except to the extent it is judicially determined such loss or damage is attributable to the Trustee/Custodian's breach of its duties hereunder or under ERISA.
- (c) An institution acting as a nondiscretionary Trustee shall have no discretion or investment management responsibility and shall only be responsible to perform the functions described at paragraph 12.3 hereof. The Custodian or Trustee (whether nondiscretionary or discretionary) has no responsibility with respect to Plan investments and does not guarantee the adequacy of the Trust to meet and discharge any or all liabilities associated with the Plan.
- (d) The Employer warrants that all directions issued to the Trustee by it or the Plan Administrator will be in accordance with the terms of the Plan and not contrary to the provisions of the Employee Retirement Income Security Act of 1974, as amended, and the regulations issued thereunder.
- (e) The Trustee shall not be answerable for any action taken pursuant to any direction, consent, certificate, or other paper or document in the belief that the same is genuine. All directions by the Employer, Participant, the Plan Administrator or an investment manager shall be in writing, shall be given orally and promptly confirmed in writing in accordance with reasonable procedures developed with the Trustee or shall be made pursuant to pre-approved communication procedures to which the Employer, Participant or Plan Administrator or an investment manager has consented to in writing. The Employer shall deliver to the Trustee written notification identifying the individual or individuals authorized to act for the Plan and shall deliver specimens of their signatures to the Trustee.
- (f) The duties and obligations of the Trustee shall be limited to those expressly imposed by this instrument or subsequently agreed upon by the parties in writing. Responsibility for administrative duties required under the Plan or applicable law not expressly imposed upon or agreed to by the Trustee shall rest solely with the Employer.
- (g) The Trustee shall be indemnified and saved harmless by the Employer from and against any and all liabilities to which the Trustee may be subjected, including all expenses reasonably incurred in its defense, except any liability judicially determined to result directly from the Trustee's or Custodian's negligence for any action or failure to act resulting from compliance with the instructions of the Employer, Plan Participants, investment manager, the Employees or agents of the Employer, the Plan Administrator, or any other fiduciary to the Plan, and for any liability arising from the actions or non-actions of any predecessor Trustee or fiduciary under the Plan.
- (h) The Trustee shall not be responsible in any way for the application of any payments it is directed to make or for the adequacy of the Trust to meet and discharge any and all liabilities under the Plan.

(i) The Employer shall indemnify the Trustee/ against, and agrees to hold the Trustee harmless from, all liabilities and claims including attorney's fees and expenses in defending against such liabilities and claims against the Trustee, arising from the Trustee's performance of its duties under this Article unless such liability or claim results from the gross negligence, willful misconduct, or failure to satisfy the standard of conduct found paragraph 14.1 herein of the Trustee, or where the Trustee is found to have breached its duties under the Article, or Part 4 of Title I of ERISA by a final judgment of a court of competent jurisdiction. The Employer also shall indemnify the Trustee including attorney's fees and other expenses in defending against such liabilities or claims, arising from any actions or breach or responsibility by any party other than the Trustee, including without limitation by specification any acts of a prior Trustee or of another Trustee appointed by the Employer.

Anything in the prior paragraph to the contrary notwithstanding, the Employer expressly agrees to indemnify the Trustee against any liability or claim (including attorney's fees and expenses in defending against such liabilities or claims) arising as a result of any act taken or failure to act, in accordance with the directions from the Employer, Plan Administrator, investment manager, Participant, or by a designee specified by the Employer, directly or transmitted by a designated service provider to the Plan and without limitation by specification.

- (j) The Trustee will take all reasonable steps to assure the security of any data received from the Employer in connection with services provided to the Plan. The Employer will be responsible for retaining duplicate copies of any such data or materials it forwards to the Trustee/Custodian and for taking all other reasonable and necessary precautions in event such data or materials are lost or destroyed, regardless of cause, or in the event reprocessing is needed for any reason. The Trustee will maintain records in connection with the performance of services hereunder for the applicable period as required by law, or if no period is required, for such period as is reasonable under the law.
- (k) The Trustee shall not be responsible for damages, defaults or delays in performance caused by acts of God, governmental authority, fire, telephone system outages and similar occurrences, as long as the Trustee is diligently attempting to correct the cause of the default or delay.
- (1) No waiver of any breach of this Article shall constitute a waiver of any other breach, whether of the same or any other covenant, term or condition. The subsequent performance of any of the terms, covenants and conditions of this Agreement shall not constitute a waiver of any preceding breach, nor shall any delay or omission of any party's exercise of any rights arising from any default effect or impair the party's rights as to the same or future default.
- (m) The Trustee shall not be responsible in any way for any actions taken, or failure to act by a prior Trustee under a prior document. The Employer shall indemnify and hold harmless the Trustee for such prior Trustee acts or inaction for any periods applicable, including periods for which the Trustee must restate the Plan retroactively to comply with any tax law or regulations thereunder.
- (n) A fiduciary with respect to the Plan shall not be liable for a breach of fiduciary responsibility of another fiduciary with respect to the Plan except to the extent that:
 - (1) it participates knowingly in, or knowingly undertakes to conceal, an act or omission of such other fiduciary, knowing such act or omission is a breach;
 - (2) by its failure to comply with Section 404(a)(1) of the Employee Retirement Income Security Act of 1974 in the administration of its specific responsibilities which give rise to its status as a fiduciary, it has enabled such other fiduciary to commit a breach; or
 - (3) it has knowledge of a breach by such other fiduciary, unless it makes reasonable efforts under the circumstances to remedy the breach.
(o) If the assets of the Plan are held by two or more Trustees, each Trustee will use reasonable care to prevent a co-Trustee from committing a breach of duty under the Employee Retirement Income Security Act of 1974, as amended, and they shall jointly manage and control the assets of the Plan; provided, however, that such co-Trustee shall be authorized to allocate specific responsibilities, obligations or duties among the co-Trustees pursuant to a written agreement. If co-Trustees do enter into such an agreement, then a Trustee to whom certain responsibilities, obligations or duties have not been allocated shall not be liable either individually or as Trustee for any loss resulting to the Plan arising from the acts or omissions on the part of another Trustee to which such responsibilities, obligations or duties have been allocated.

12.5 **Persons Serving as Plan Administrator** If the Employer is no longer in existence, and the Plan does not specify the person to take an action or otherwise serve in the place of the Employer, in connection with the operation of the Plan, the Plan Administrator shall so act or serve, but if there is no person serving as Plan Administrator, the Employer shall then be the Plan Administrator or designate a successor Plan Administrator. If the Employer has died if a sole proprietorship, dissolved if a corporation, or terminated if a partnership, a successor shall be designated in writing by a majority of the Participants for whom accounts not yet fully distributed are then in the Plan. A majority of the legally competent Designated Beneficiaries of a deceased Participant then entitled to receive benefits may exercise the deceased Participant's rights to participate in that designation and shall be considered for that purpose to be one Participant, in the Participant's place.

ARTICLE XIII

TRUST FUND

13.1 The Trust The Trust shall consist of all money and property received under Article III and Article IV of the Plan and Trust document increased by any income on or increment in such assets and decreased by any investment loss, expense, benefit payment, withdrawal or other distribution. The Trustee shall hold the Trust, without distinction between principal and income. The Trust shall be administered in accordance with the provisions of this document.

13.2 Control of Plan Assets The assets of the Trust or evidence of ownership shall be held by the Trustee and/or the Custodian under the terms of the Plan and Trust/Custodial Account. If the assets represent amounts transferred from another Trustee under a former plan, the Trustee named hereunder shall not be responsible for any actions of the prior fiduciary including the propriety of any investment decision made by the prior Trustee under any prior plan. Such review shall be the responsibility of the Employer.

13.3 *Exclusive Benefit Rules* No part of the Trust shall be used for, or diverted to, purposes other than for the exclusive benefit of Participants, former Participants with a vested interest, and the beneficiary or beneficiaries of deceased Participants having a vested interest in the Plan at death.

13.4 Assignment and Alienation of Benefits Except as provided in paragraphs 13.5 or 14.5, no right or claim to, or interest in, any part of the Plan, or any payment from the Plan, shall be assignable, transferable, or subject to sale, mortgage, pledge, hypothecation, commutation, anticipation, garnishment, attachment, execution, or levy of any kind. The Trustee shall not recognize any attempt to assign, transfer, sell, mortgage, pledge, hypothecate, commute, or anticipate the same, except to the extent required by law. The preceding sentences shall also apply to the creation, assignment, or recognition of a right to any benefit payable with respect to a Participant pursuant to a domestic relations order, unless such order is determined to be a Qualified Domestic Relations Order, as defined in Code Section 414(p), or any domestic relations order entered before January 1, 1985 which the Plan's attorney and Plan Administrator deem to be qualified.

13.5 Determination of Qualified Domestic Relations Order (QDRO) A domestic relations order shall specifically state all of the following in order to be deemed a Qualified Domestic Relations Order ("QDRO"):

- (a) The name, social security number and last known mailing address (if any) of the Participant and of each alternate payee covered by the QDRO. However, if the QDRO does not specify the current mailing address of the alternate payee, but the Plan Administrator has independent knowledge of that address, the QDRO will still be valid.
- (b) The dollar amount or percentage of the Participant's benefit to be paid by the Plan to each alternate payee, or the manner in which the amount or percentage will be determined.
- (c) The number of payments or period for which the order applies.
- (d) The specific Plan (by name) to which the domestic relations order applies.

The domestic relations order shall not be deemed a QDRO if it requires the Plan to provide:

- (e) any type or form of benefit, or any option not already provided for in the Plan;
- (f) increased benefits, or benefits in excess of the Participant's vested rights;
- (g) payment of a benefit earlier than allowed by the Plan's earliest retirement provisions or, in the case of a profit-sharing plan, prior to the first date on which an in-service withdrawal is allowed; or

(h) payment of benefits to an alternate payee which are required to be paid to another alternate payee under another ODRO.

Promptly, upon receipt of a domestic relations order ("Order") which may or may not be "qualified", the Plan Administrator shall notify the Participant and any alternate payee(s) named in the Order of such receipt, and include a copy of this paragraph. The Plan Administrator shall establish written procedures to establish the qualified status of a domestic relations order, which may include forwarding the Order to the Plan's legal counsel for an opinion as to whether or not the Order is in fact "qualified" as defined in Code Section 414(p). Within a reasonable time after receipt of the Order, not to exceed 60 days, the Plan's legal counsel shall make a determination as to its "qualified" status and the Participant and any alternate payee(s) shall be promptly notified in writing of the determination.

If the "qualified" status of the Order is in question, there will be a delay in any payout to any payee including the Participant, until the status is resolved. In such event, the Plan Administrator shall segregate the amount that would have been payable to the alternate payee(s) if the Order had been deemed a QDRO. If the Order is not qualified, or the status is not resolved (for example, it has been sent back to the Court for clarification or modification) within 18 months beginning with the date the first payment would have to be made under the Order, the Plan Administrator shall pay the segregated amounts plus interest to the person(s) who would have been entitled to the benefits had there been no Order. If a determination as to the qualified status of the Order is made after the 18-month period described above, then the Order shall only be applied on a prospective basis. If the Order is determination. Once an Order is deemed a QDRO, the Plan Administrator shall pay to the alternate payee(s) all the amounts due under the QDRO, including segregated amounts plus earnings, if any, which may have accrued during a dispute as to the Order's qualification.

Unless specified otherwise in the Adoption Agreement, the QDRO retirement age with regard to the Participant against whom the order is entered shall be the date the order is determined to be qualified. This will only allow payouts to alternate payee(s) and not the Participant.

59

ARTICLE XIV

INVESTMENTS

14.1 *Fiduciary Standards* The Trustee shall invest and reinvest principal and income of the Trust in accordance with the funding policy and investment objectives established by the Employer, provided that:

- (a) such investments are prudent under the Employee Retirement Income Security Act of 1974, as amended, and the regulations thereunder,
- (b) such investments are sufficiently diversified to minimize the risk of large losses,
- (c) such investments are made in accordance with the provisions of this Plan and Trust/Custodial Account document, and
- (d) such investments are made with the care, skill, prudence and diligence under the circumstances then prevailing that a prudent man, acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character with like aims.

14.2 **Funding Arrangement** The Employer shall, in the Adoption Agreement, appoint a Trustee. Such appointment shall be made in the Adoption Agreement. The Employer may execute a separate trust outlining the Trustee's duties and responsibilities which shall be incorporated and made part of this Plan and Trust Account document. Such ancillary agreement may not conflict with any provision of this Trust Account document. Any provision which would jeopardize the tax-qualified status of this Plan and Trust shall be null and void.. The Trustee shall invest the Trust in any of the investment alternatives under paragraph 14.3.

14.3 Investment Alternatives of the Trustee The Trustee shall implement an investment program based on the Employer's investment policy statement, investment objectives and the Employee Retirement Income Security Act of 1974, as amended. In addition to powers given by law, the Trustee may:

- (a) invest the Trust in any form of property, including common and preferred stocks, exchange-traded covered put and call options, bonds, money market instruments, mutual funds (including funds for which the Sponsor, Trustee or its affiliates receive Compensation for providing investment advisory, custody, transfer agency or other services), savings accounts, certificates of deposit, securities issued by the U.S. government or by governmental agencies, insurance policies and contracts, or in any other property, real or personal, having a ready market, including securities issued by the Trustee and/or affiliates of the Trustee as permitted by law. The Trustee may invest in time deposits (including, if applicable, its own or those of affiliates) which bear a reasonable interest rate. No portion of any Qualified Voluntary Contribution, or the earnings thereon, may be invested in life insurance contracts or, as with any Participant-directed investment, in tangible personal property characterized by the IRS as a collectible;
- (b) invest any assets of the Trust in a group or collective trust established to permit the pooling of funds of separate pension and profit-sharing trusts, provided the Internal Revenue Service has ruled such group or collective trust to be qualified under Code Section 401(a) and exempt under Code Section 501(a) (or the applicable corresponding provision of any other Revenue Act) or to any other common, collective, or commingled trust fund which has been or may hereafter be established and maintained by the Trustee, affiliate(s) of the Trustee, the Custodian or investment manager. Such commingling of assets of the Trust with assets of other qualified trusts is specifically authorized, and to the extent of the investment of the Trust in such a group or collective trust, the terms of the instrument establishing the group or collective trust shall be a part hereof as though set forth herein. The Employer expressly understands and agrees that any such collective fund may provide for the lending of its securities by the collective fund trustee and that such collective fund's trustee will receive compensation from such collective fund for the lending

of securities that is separate from any compensation of the Trustee hereunder, or any compensation of the collective fund trustee for the management of such collective fund;

- (c) invest up to 100% of the Trust in the common stock, debt obligations, or any other security issued by the Employer or by an affiliate of the Employer within the limitations provided under Sections 406, 407, and 408 of the Employee Retirement Income Security Act of 1974, as amended, and further provided that such investment does not constitute a prohibited transaction under Code Section 4975. Any such investment in Employer securities shall only be made upon written direction of the Employer who shall be solely responsible for the propriety of such investment. Additional directives regarding the purchase, sale, retention or valuing of such securities may be addressed in an investment management or trust agreement which may be an attachment to this document. If there are any conflicts between this document and the above referenced agreements, this document shall govern;
- (d) hold cash uninvested and deposit same with any banking or savings institution, including its own banking department or banking department of an affiliate;
- (e) join in or oppose the reorganization, recapitalization, consolidation, sale or merger of corporations or properties, including those in which it or its affiliates are interested as Trustee, upon such terms as it deems advisable;
- (f) hold investments in nominee or bearer form;
- (g) exercise all ownership rights including the voting of proxies and the exercise of tender offers but only with respect to assets over which the Trustee has investment management responsibility and specifically excluding any securities described in 14.3(c); and
- (h) to apply for and procure from an insurance company as an investment of the Trust such annuity, or other contracts on the life of any Participant as the Plan Administrator shall deem proper; to exercise, at any time or from time to time, whatever rights and privileges may be granted under such annuity, or other contracts; to collect, receive, and settle for the proceeds of all such annuity, or other contracts as and when entitled to do so under the provisions thereof.
- (i) With respect to the voting of, or in the event of any tender or other offer with respect to shares of Employer securities held in the Trust, the Trustee/Custodian shall follow the direction of each Participant with respect to the shares allocated to the account of each Participant. With respect to shares which are not allocated or for which no Participant direction has been received ("nondirected shares"):
 - (1) As to the voting of "non-directed shares" on an item determined by the Trustee/Custodian to be an extraordinary item, or with respect to a tender offer, the "non-directed shares" shall be voted or tendered as determined by an investment manager, or independent fiduciary selected for this purpose by the Employer or if such selection is not timely made, by the Trustee.
 - (2) With respect to the voting of "non-directed shares" on any item not an extraordinary item, the Trustee/Custodian shall vote these shares in the same proportion as shares for which direction was received.
 - (3) The Employer shall be responsible for preparing and distributing all required prospectuses for Employer securities and making such materials available to Plan Participants.

14.4 Participant Loans If permitted by the Employer in the Adoption Agreement, a Plan Participant and Beneficiaries may make application to the Employer requesting a loan from the Trust. The Employer shall have the

sole right to approve or deny a Participant's application provided that loans shall be made available to all Participants on a reasonably equivalent basis. Loans shall not be made available to Highly Compensated Employees in an amount greater than the amount made available to other Employees. Any loan granted under the Plan shall be made in accordance with the terms of a loan policy adopted by the Employer and shall be subject to the following rules to the extent such rules are not inconsistent with such loan policy:

- (a) No loan, when aggregated with any outstanding Participant loan(s), shall exceed the lesser of (i) \$50,000 reduced by the excess, if any, of the highest outstanding balance of all loans on any day during the one year period ending on the day before the loan is made, over the outstanding balance of loans from the Plan on the date the loan is made or (ii) one-half of the fair market value of a Participant's Vested Account Balance consisting of Employer contributions, Voluntary Contributions, and Rollover Contributions. If the Participant's Vested Account Balance is \$20,000 or less, the maximum loan shall not exceed the lesser of \$10,000 or 100% of the Participant's Vested Account Balance. For the purpose of the above limitation, all loans from all plans of the Employer and other members of a group of employers described in Code Sections 414(b), 414(c), and 414(m) are aggregated. An assignment or pledge of any portion of the Participant's interest in the Plan and a loan, pledge, or assignment with respect to any insurance contract purchased under the Plan, will be treated as a loan under this paragraph.
- (b) All applications must be in accordance with procedures adopted by the Employer.
- (c) Any loan shall bear interest at a rate reasonable at the time of application, considering the purpose of the loan and the rate being charged by representative commercial banks in the local area for a similar loan unless the Employer sets forth a different method for determining loan interest rates in its loan procedures. The loan agreement shall also provide that the payment of principal and interest be amortized in level payments not less frequently than quarterly.
- (d) The term of such loan shall not exceed five years except in the case of a loan for the purpose of acquiring any house, apartment, condominium, or mobile home which is used or is to be used within a reasonable time as the principal residence of the Participant. The term of such loan shall be determined by the Employer considering the maturity dates quoted by representative commercial banks in the local area for a similar loan.
- (e) The principal and interest paid by a Participant on his or her loan shall be credited to the Trust in the same manner as for any other Plan investment. If elected in the Adoption Agreement, loans may be treated as segregated investments of the individual Participants. This provision is not available if its election will result in discrimination in operation of the Plan.
- (f) If a Participant's loan request is approved by the Employer, it shall be evidenced by a note, loan agreement, and assignment of 50% of his or her interest in the Trust as collateral for the loan. The Participant, except in the case of a profit-sharing plan satisfying the requirements of paragraph 8.7 must obtain the consent of his or her Spouse, if any, within the 90-day period before the time his or her account balance is used as security for the loan. A new consent is required if the account balance is used for any renegotiation, extension, renewal or other revision of the loan, including an increase in the loan amount. The consent must be written, must acknowledge the effect of the loan, and must be witnessed by a Plan representative or notary public. Such consent shall subsequently be binding with respect to the consenting Spouse or any subsequent Spouse.
- (g) If a valid Spousal consent has been obtained, then, notwithstanding any other provision of this Plan, the portion of the Participant's Vested Account Balance used as a security interest held by the Plan by reason of a loan outstanding to the Participant shall be taken into account for purposes of determining the amount of the account balance payable at the time of death or distribution, but only if the reduction is used as repayment of the loan. If less than 100% of the Participant's Vested Account Balance (determined without regard to the preceding sentence) is payable to the surviving Spouse, then the account balance shall be adjusted by first reducing the Vested Account

••

Balance by the amount of the security used as repayment of the loan, and then determining the benefit payable to the surviving Spouse.

- (h) The Plan Administrator may require additional collateral in order to adequately secure the loan.
- (i) Any loan made hereunder shall be subject to the provisions of a loan agreement, promissory note, security agreement, payroll withholding authorization and, if applicable, financial disclosure. Such documentation may contain additional loan terms and conditions not specifically itemized in this section provided that such terms and conditions do not conflict with this section. Such additional terms and conditions may include, but are not limited to, procedures regarding default, a grace period for missed payments and acceleration of a loan's maturity date on specific events such as termination of employment.
- (j) No loans will be made to Owner-Employees or Shareholder Employees, unless the Employer obtains a prohibited transaction exemption from the Department of Labor.
- (k) Liquidation of a Participant's assets for the purpose of the loan will be allocated on a pro-rata basis across all the investment alternatives in a Participant's account.
- (1) If the Plan permits loans to Participants, the Trustee may appoint the Employer as its agent, and if the Employer accepts such appointment, agrees to hold all notes and other evidence of any loans to Participants. The Employer shall hold such notes and evidence under such conditions of safekeeping as is prudent as required by ERISA. The Trustee may account for all loans in the aggregate so that all Participant loans will be shown collectively as a single asset of the Plan.
- (m) Unless elected otherwise in the Adoption Agreement, loan payments will be suspended under this Plan as permitted under Code Section 414(u).

14.5 Insurance Policies If agreed upon by the Trustee and permitted by the Employer in the Adoption Agreement, Employees may elect the purchase of life insurance policies under the Plan. If elected, the maximum annual premium for a whole life policy shall be less than 50% of the aggregate Employer contributions allocated to the account of a Participant. For profit-sharing plans, the 50% test need only be applied against Employer contributions allocated in the last two years. Whole life policies are policies with both fixed death benefits and fixed premiums. The maximum annual premium for term contracts or universal life policies and all other policies which are not whole life shall not exceed 25% of aggregate Employer contributions allocated to the profit-sharing account of a Participant in the last two years. The 25% test need only be applied against Employer contributions allocated in the last two years. The 25% test need only be applied against Employer contributions allocated in the last two years. The 25% test need only be applied against Employer contributions allocated in the last two years. The 25% test need only be applied against Employer contributions allocated in the last two years. The 25% test need only be applied against Employer contributions allocated in the last two years. The maximum annual premiums for a Participant with both a whole life and a term contract or universal life policies shall be limited to one-half of the whole life premium plus the term premium, but shall not exceed 25% of the aggregate Employer contributions allocated to the account of a Participant, subject to the two year rule for profit-sharing plans. Any policies purchased under this Plan shall be held subject to the following rules:

- (a) The Trustee shall be applicant and owner and named beneficiary of any policies issued hereunder.
- (b) All policies or contracts purchased hereunder, shall be endorsed as nontransferable, and must provide that proceeds will be payable to the Trustee; however, the Trustee shall be required to pay all proceeds of the contracts to the Participant's Designated Beneficiary in accordance with the distribution provisions of this Plan. Under no circumstances shall the Trust retain any part of the proceeds.
- (c) The Employer/Plan Administrator shall select the insurance company and the policy. The Employer/Plan Administrator will direct the Trustee as to the purchase of the insurance contract. Such direction shall include but not be limited to the term, price and the insurance company from which the policy should be purchased.

. -

- (d) The Trustee must be the named beneficiary on any policy to receive the proceeds of the contract after settlement is received by the Trustee. Such designation shall remain in force, until revoked by the Participant, by filing a new beneficiary form with the Trustee. A Participant's Spouse will be the Designated Beneficiary of the proceeds in all circumstances unless a Qualified Election has been made in accordance with paragraph 8.4. The beneficiary of a deceased Participant shall receive, in addition to the proceeds of the Participant's policy or policies, the amount credited to such Participant's investment account.
- (e) A Participant who is uninsurable or insurable at substandard rates, may elect to receive a reduced amount of insurance, if available, or may waive the purchase of any insurance.
- (f) All dividends or other returns received on any policy purchased shall be applied to reduce the next premium due on such policy, or if no further premium is due, such amount shall be credited to the Trust as part of the account of the Participant for whom the policy is held.
- (g) If Employer contributions are inadequate to pay all premiums on all insurance policies, the Trustee may, at the option of the Employer, utilize other amounts remaining in each Participant's account to pay the premiums on his or her respective policy or policies, allow the policies to lapse, reduce the policies to a level at which they may be maintained, or borrow against the policies on a prorated basis, provided that the borrowing does not discriminate in favor of the policies on the lives of Highly Compensated Employees.
- (h) On retirement or termination of employment of a Participant, termination of the Plan, or the contract would but for the sale, be surrendered by the Plan, the Employer shall direct the Trustee to surrender the Participant's policy and credit the proceeds to his or her account for distribution under the terms of the Plan. However, before so doing, the Trustee shall first offer to transfer ownership of the policy to the Participant. Prior to such transfer, the Participant may elect to make payment to the Trust of the cash value of the policy. Such payment shall be credited to the Participant's account for distribution under the terms of the Plan. All distributions resulting from the application of this paragraph shall be subject to the Joint and Survivor Annuity Rules of Article VIII, if applicable.
- (i) The Employer shall be solely responsible to ensure the insurance provisions are administered properly and that if there is any conflict between the provisions of this Plan and any insurance contracts issued, the terms of this Plan will control.
- (j) Notwithstanding the above, in profit-sharing plans, the limitations imposed herein with respect to the purchase of life insurance shall not apply to any Participant who has participated in this Plan for five (5) or more years or to the portion of a Participant's Vested Account Balance, that would be eligible for withdrawal under paragraph 6.8 whether or not in-service withdrawals are actually allowed under this Plan, that has accumulated for at least two (2) Plan Years. In addition, under such Plans, a Participant may, subject to the limitations set forth in this subsection, elect to have keyman life insurance purchased on the life of any Participant who is considered essential to the success of the Employer's business. In such case, the proceeds of such a life insurance contract in excess of such contract's cash value as of the date of death of such insured shall be paid to the beneficiaries named with respect to such contract. Death benefits, including those in the previous sentence, payable from a life insurance contract shall be paid in accordance with paragraph 8.7, if this Plan meets the safe-harbor provisions in that paragraph, or in accordance with paragraph 8.2 or 8.3, whichever may be applicable. The cash value of the contract shall be added to the Participant's Vested Account Balance.

14.6 Allocation of Investment Responsibility

(a) In the Adoption Agreement, the Employer shall elect to direct Plan investments or to allocate investment management authority to the Trustee, an independent investment manager under ERISA, or to Participants. The Employer is the named fiduciary for investment purposes if the Employer directs investments pursuant to this paragraph. If an investment manager is named, such entity or individual must be registered as an investment manager under the Investment Advisors Act of 1940 or under applicable state law, be a bank as defined in said Act or an insurance company qualified under the laws of more than one state to perform investment management services. An investment manager shall acknowledge in writing its appointment and fiduciary status hereunder and shall agree to comply with all applicable provisions of this Plan and Trust/Custodial Account. A copy of the investment management agreement (and any modifications or termination thereof) must be given to the Trustee or Custodian. Written notice of each appointment of an investment manager shall be given to the Trustee/Custodian in advance of the effective date of the appointment. Such notice shall specify what portion of the Trust will be subject to the investment manager's discretion.

- (b) The Employer may also enter into any other ancillary agreement with the Sponsor, Trustee, Custodian or investment manager as may be necessary to clarify each of those party's duties and responsibilities. Such agreement shall pertain to Plan assets held by the separate Trustee and shall be incorporated and made part of this Plan and Trust/Custodial Account document. In such event, the Trustee and the ancillary Trustee shall have no liability with respect to assets held by the other as permitted by Section 405(b)(1)(B) of ERISA. Such ancillary agreement may not conflict with any investment or other provision contained herein. Any language which is in conflict with the provisions of this Trust/Custodial Account or which would jeopardize the tax qualified status of this Plan and Trust/Custodial Account shall be null and void. The Employer may allocate investment management responsibility to more than one party. In such event, each of the investment managers shall only be responsible for that part of the Trust Fund which is under their investment management.
- (c) The Employer shall advise the Trustee in writing regarding the retention of investment powers. The appointment of an investment manager will be given in accordance with paragraph 12.5(e) herein, or the delegation of investment powers to the Trustee. The Employer is a named fiduciary for investment management purposes if the Employer directs investments pursuant to this paragraph. Any investment directive under this Plan shall be made in writing or such other form as agreed to by the Employer, Trustee and investment manager. In the absence of such directive, cash shall be automatically invested in such investment or investments as the Employer or other fiduciary shall select from the investments the Trustee/Custodian makes available for that purpose unless and until the person or persons responsible for giving directions directs otherwise. Such automatic investment shall be made at regular intervals and pursuant to procedures provided by the Trustee/Custodian (which procedures may without limitation, provide for more frequent intervals only if uninvested balances exceed a stated amount). Absent a contrary direction in accordance with the preceding provisions of this paragraph 14.7(c), such instructions regarding the delegation of investment responsibility shall remain in force until revoked or amended in writing. The Trustee shall not be responsible for the propriety of any directed investment made and shall not be required to consult with or advise the Employer regarding the investment quality of any directed investment held hereunder. If the Employer fails to designate an investment manager, the Trustee shall have full investment management authority. If the Employer does not issue investment directions with regard to specific assets held in the Trust, the Trustee shall have authority to invest those assets in the Trust in its sole discretion. While the Employer may direct the Trustee with respect to Plan investments, the Employer may not:
 - (1) borrow from the Plan or pledge any of the assets of the Plan as security for a loan,
 - (2) buy property or assets from or sell property or assets to the Plan,
 - (3) charge any fee for services rendered to the Plan, or
 - (4) receive any services from the Plan on a preferential basis.

Prohibited Transactions The Trustee, Employer, investment manager or Participant shall not knowingly 14.7 enter into any transaction, engage in any activity, or direct the purchase or acquisition of any investment with respect to the Plan which would constitute a prohibited transaction under ERISA or the Internal Revenue Code. The Trustee or Custodian shall not receive investment advisory or other fees from a registered investment company (a mutual fund) which duplicates other investment management fees charged by the Trustee for investment management services. Such fees may be received by the Trustee where there is no duplication of fees and the Trustee provides complete disclosure to the Employer as an independent fiduciary under the Plan. The Trustee or Custodian shall be permitted to receive fees from a regulated investment company (a mutual fund) if the Trustee or Custodian has made a good faith determination that the receipt of such fees is not a prohibited transaction pursuant to any guidance issued by the Department of Labor from time to time. A nondiscretionary Trustee is permitted to accept fees from a registered investment company for services provided in connection with the administration (other than fees received as investment advisor) or distribution of mutual fund shares. The receipt of such fees is subject to full disclosure to the Employer as an independent fiduciary. A nondiscretionary Trustee is one who has no investment management responsibility or discretion over any part of the Trust. The Custodian may accept such fees, as it never has discretion or investment management responsibility over any part of the Trust.

14.8 Participant Investment Direction If elected by the Employer in the Adoption Agreement, Participants shall be given the option to direct the investment of such part of their account balances as specified in the Adoption Agreement. The investment vehicles available from time to time shall be selected by the Employer or the Employer's designated fiduciary, independent of the Trustee, who shall be responsible for reviewing the performance of such investments. The following administrative procedures shall apply to the administration of specific investment vehicles selected by the Employer:

- (a) The program shall be administered by the Plan Administrator or other party appointed by the Employer.
- (b) At the time an Employee becomes eligible for the Plan, he or she shall complete an investment designation form stating the percentage of his or her contributions to be invested in the available alternatives.
- (c) A Participant may change his or her election with respect to future contributions by notifying the Employer, Trustee or other service provider, as they shall mutually agree, in accordance with the procedures established by the Plan Administrator.
- (d) A Participant may elect to transfer all or part of his or her balance from one investment alternative to another by notifying the Employer in accordance with the procedures established by the Plan Administrator.
- (e) The Employer or other service provider to the Plan, when transmitting contributions shall be responsible for providing the Trustee with investment directives setting forth the necessary information so that the Trustee can purchase the proper amount of each investment alternative available under the Plan.
- (f) Except as otherwise provided in the Plan, neither the Trustee, nor the Employer, nor any fiduciary of the Plan shall be liable to the Participant or any of his or her beneficiaries for any loss resulting from action taken at the direction of the Participant. All fiduciaries of the Plan shall be relieved of their fiduciary liability with respect to the Participant directing his or her investments pursuant to ERISA Section 404(c) if elected by the Employer on the Adoption Agreement.
- (g) The Employer may, in a uniform and nondiscriminatory manner limit the available investments in the Plan. The Employer shall specify that Participants are permitted to direct investments in accordance with the alternatives specified at paragraph 14.3 hereof or may restrict investments to specific investment alternatives selected by the Employer (including but not limited to, certain mutual funds, investment contracts, collective funds or deposit accounts). If investments outside the alternatives selected by the Employer are permitted, Participants may not direct that investments be made in collectibles, other than U.S. Government or state issued gold and silver

coins. The Employer may permit, in a uniform and nondiscriminatory manner, a beneficiary of a deceased Participant or alternate payee under a Qualified Domestic Relations Order [as defined in Code Section 414(p)] to individually direct their account in accordance with this section.

(h) Investment directions will be processed as soon as administratively practicable after proper investment directions are received from the Participant. The Employer, Plan Administrator, Trustee and/or Custodian cannot provide any guarantee that investment directions will be processed on a daily basis (if a daily valuation Plan), or provide a guarantee in any respect as to the processing time of an investment direction. The Employer, Plan Administrator, Trustee and/or Custodian reserves the right not to value an investment alternative or a Participant's account on any given Valuation Date for any reason deemed appropriate by the Employer or Plan Administrator. The Employer, Plan Administrator and Trustee and/or Custodian further reserves the right to delay the processing of any investment transaction for any legitimate business reason (including but not limited to, failure of systems or computer programs, failure of the means of the transmission of data, force majeure, the failure of a service provider to timely receive values or prices, to correct its errors or omissions or the errors or omissions of any service provider).

÷

ARTICLE XV

TOP-HEAVY PROVISIONS

15.1 *Applicability of Rules* If the Plan is or becomes Top-Heavy in any Plan Year beginning after 1983, the provisions of this Article will supersede any conflicting provisions in the Plan or Adoption Agreement.

15.2 *Minimum Contribution* Notwithstanding any other provision in the Employer's Plan, for any Plan Year in which the Plan is Top-Heavy or Super Top-Heavy, the aggregate Employer contributions and forfeitures allocated on behalf of any Participant (without regard to any Social Security contribution) under this Plan and any other Defined Contribution Plan of the Employer shall be determined as follows:

- (a) When the Employer maintains one Plan or a combination of Paired or non-paired Defined Contribution Plans and no Defined Benefit Plans which are Top-Heavy or Super Top-Heavy, the Employer will contribute the lesser of 3% of such Participant's Compensation or the largest percentage of the Employer contributions and forfeitures, as a percentage of the Key Employee's Compensation, up to a maximum permitted under Code Section 401(a)(17), as indexed, allocated on behalf of any Key Employee for that year.
- (b) If the Employer maintains or maintained a Defined Benefit Plan, the provisions of the "Limitations on Allocations" section of the Adoption Agreement shall apply.
- (c) Each Participant who is employed by the Employer on the last day of the Plan Year shall be entitled to receive an allocation of the Employer's minimum contribution for such Plan Year. The minimum allocation applies even though under other Plan provisions the Participant would not otherwise be entitled to receive an allocation, or would have received a lesser allocation for the year because the Participant fails to make required contributions to the Plan, the Participant's Compensation is less than a stated amount, or the Participant fails to complete 1,000 Hours of Service (or such lesser number designated by the Employer in the Adoption Agreement) during the Plan Year. A Paired profit-sharing Plan designated to provide the minimum Top-Heavy contribution must do so regardless of profits. An Employer may make the minimum Top-Heavy contribution available to all Participants or just non-Key Employees.

The Top-Heavy minimum contribution does not apply to any Participant to the extent the Participant is covered under any other plan(s) of the Employer and the Employer has provided in the Adoption Agreement that the minimum allocation or benefit requirements applicable to this Plan will be satisfied in the other plan(s).

If a Key Employee makes an Elective Deferral or has an allocation of Matching Contributions credited to his or her account, a Top-Heavy minimum will be required for non-Key Employees who are Participants. For purposes of satisfying the minimum contribution requirement, Elective Deferrals and Matching Contributions are not taken into account.

15.3 Minimum Vesting For any Plan Year during which this Plan is Top-Heavy, the minimum vesting schedule elected by the Employer in the Adoption Agreement will automatically apply to the Plan. If the vesting schedule selected by the Employer in the Adoption Agreement is less liberal than the allowable schedule, the schedule will automatically shift to one which satisfies the Top-Heavy minimum. If the vesting schedule under the Employer's Plan shifts in or out of the Top-Heavy schedule for any Plan Year, such shift is an amendment to the vesting schedule and the election in paragraph 9.8 of the Plan applies. The minimum vesting schedule applies to all accrued benefits within the meaning of Code Section 411(a)(7) except those attributable to Employee contributions, including benefits accrued before the effective date of Code Section 416 and benefits accrued before the Plan became Top-Heavy. No reduction in vested benefits may occur in the event the Plan's status as Top-Heavy changes for any Plan Year. This paragraph does not apply to the account balances of any Employee who does not have one Hour of Service after the Plan initially becomes Top-Heavy and such Employee's account balance attributable to Employer contributions and forfeitures will be determined without regard to this paragraph. 15.4 *Limitations on Allocations* In any Plan Year beginning prior to January 1, 2000, in which the Top-Heavy Ratio exceeds 90% (i.e., the Plan becomes Super Top-Heavy), the denominators of the Defined Benefit Fraction and Defined Contribution Fraction shall be computed using 100% of the dollar limitation instead of 125%.

₹,

÷

.

ARTICLE XVI

AMENDMENT AND TERMINATION

16.1 Amendment The Employer may amend any or all provisions of the Plan from time to time.

16.2 **Protected Benefits** An amendment (including the adoption of this Plan as a restatement of an existing plan) may not decrease a Participant's accrued benefit, except to the extent permitted under Code Section 412(c)(8), and may not reduce or eliminate a Code Section 411(d)(6) protected benefit determined immediately prior to the adoption date, or if later, the Effective Date of the amendment. Where this Plan is being adopted to amend another plan that contains a protected benefit not provided for in this document, the Employer may attach a schedule to the Adoption Agreement that describes such protected benefit which shall become part of the Plan.

Plan Termination The Employer shall have the right to terminate its Plan upon 60 days notice in writing 16.3 to the Trustee. If the Plan is terminated, partially terminated, or if there is a complete discontinuance of contributions under a profit-sharing plan maintained by the Employer, all amounts credited to the accounts of Participants shall vest and become nonforfeitable. In the event of a partial termination, only those who are affected by such partial termination shall be fully vested. In the event of termination, the Plan Administrator shall direct the Trustee with respect to the distribution of accounts to or for the exclusive benefit of Participants or their beneficiaries. Such distribution shall be made directly to Participants or, at the direction of the Participant, may be transferred directly to another Eligible Retirement Plan or Individual Retirement Account. In the absence of an election by a Participant who has received notice from the Plan Administrator under paragraph 6.11, the Plan Administrator may direct the Trustee to transfer the Participant's benefit to another Defined Contribution Plan maintained by the Employer, other than an employee stock ownership plan. If the Employer does not maintain another Defined Contribution Plan, the Plan Administrator may direct the Trustee to transfer the Participant's benefit to an Individual Retirement Account with an institution selected by the Plan Administrator, or make a distribution pursuant to paragraph 7.15. Prior to making any distribution, the Plan Administrator shall establish in a manner acceptable to the Trustee, that the Plan has received a favorable determination letter from the Internal Revenue Service approving the Plan termination and authorizing the distribution of benefits to Plan Participants. In the absence of such determination letter, the Trustee may agree to make distributions to Participants if the applicable requirements, if any, of the Employee Retirement Income Security Act of 1974 and the Internal Revenue Code governing the termination of employee benefit plans, have been or are being, complied with, or that appropriate authorizations, waivers, exemptions, or variances have been, or are being obtained.

16.4 Distribution Restrictions The portion of the Participant's nonforfeitable Accrued Benefit attributable to Elective Deferrals (or to amounts treated under the Plan as Elective Deferrals) is not distributable on account of Plan termination, as described in this paragraph, unless:

- (a) the Participant otherwise is entitled under the Plan to a distribution of that portion of this nonforfeitable Accrued Benefit, or,
- (b) the Plan termination occurs without the establishment of a successor Plan. A successor Plan under clause (b) is a Defined Contribution Plan (other than an employee stock ownership plan) maintained by the Employer (or by a related Employer) at the time of the termination of the Plan or within the period ending twelve months after the final distribution of assets. A distribution made after March 31, 1988, pursuant to clause (b), must be part of a lump sum distribution to the Participant of his nonforfeitable Accrued Benefit.
- 16.5 Mergers and Consolidations
 - (a) In the case of any merger or consolidation of the Employer's Plan with, or transfer of assets or liabilities of the Employer's Plan to, any other plan, Participants in the Employer's Plan shall be entitled to receive benefits immediately after the merger, consolidation, or transfer which are equal to or greater than the benefits they would have been entitled to receive immediately before the merger, consolidation, or transfer if the Plan had then terminated.

(b) Any corporation into which the Trustee or any successor thereto may be merged or with which it may be consolidated, or any corporation resulting from any merger or consolidation to which the Trustee or any successor thereto may be a party, or any corporation to which all or substantially all the business of the Trustee or any successor thereto may be transferred, shall automatically be the successor without the filing of any instrument or performance of any further act, before any court.

16.6 Resignation and Removal The Trustee may resign by written notice to the Employer which shall be effective 90 days after delivery. The Trustee shall deliver the Trust to its successor on the effective date of the resignation or removal, or as soon thereafter as practicable, provided that this shall not waive any lien the Trustee may have upon the Trust for its compensation or expenses. Following the effective date of the notice of termination, the Trustee shall have no further responsibility for providing services to the Employer or the Plan. If the Employer fails to appoint a successor trustee within 60 days, or such longer period as the Trustee may specify in writing, the Employer shall be deemed the successor trustee. In such event, the Trustee may continue to hold custody of the assets of the Plan until such time as appropriate arrangements have been made for the security of the Plan assets, but upon notification thereof to Plan Participants, shall no longer have any responsibility for the investment of Plan assets.

ARTICLE XVII

GOVERNING LAW

17.1 Governing Law Construction, validity and administration of this Plan and Trust adopted under the terms of this document and accompanying Adoption Agreement, shall be governed by Federal law to the extent applicable and to the extent not applicable by the laws of the State of Delaware.

_

. 1

EXHIBIT B

.

RSP ADOPTION AGREEMENT

Chesapeake Utilities Corporation Retirement Savings Plan

The Employer named below hereby establishes a Plan for eligible Employees as provided in this Adoption Agreement and the accompanying Plan Document.

I. EMPLOYER INFORMATION

A. Name And Address:

Chesapeake Utilities Corporation PO Box 615 Dover, DE 19903

- B. Telephone Number: (302) 734-6744
- C. Employer's Tax ID Number: 51-0064146
- D. Form of Business: Corporation
- E. Name of Plan: Chesapeake Utilities Corporation Retirement Savings Plan

002

52-1124134

Plan Number:

F

۰.

۰.

- G. Trust Tax ID Number:
- H. Employer's SIC Code: 4920
- I. Employer's Tax Year End: December 31
- J. Employer's State of Incorporation: Delaware

II. EFFECTIVE DATE

ţ

1

This is an Amended Plan.. The Effective Date of the original Plan was February 1, 1977. The most recent Amendment has an Effective Date of January 1, 1999.

III. **DEFINITIONS**

A.. "Compensation"

Employer Contribution Type	Compensation Definition	Computation Period	Dollar Limitation	Exclusions
Discretionary	a	а	\$	a, c
Match	a	a	\$	a, c
QNEC/QMAC	a	а	\$	a, c

Employee Contribution Type	Compensation Definition	Computation Period	Dollar Limitation	Exclusions
Elective Deferrals	a	а	\$	a, c
After-Tax Voluntary			\$	
Required After-Tax			\$	

Antidiscrimination Tests	Compensation Definition	Computation Period	Dollar Limitation	
ADP/ACP		· · · · · · · · · · · · · · · · · · ·	\$	

1. Compensation Definition:

- a. Code §3401(a) W-2 Compensation subject to income tax withholding at the source.
- b. Code §3401(a) W-2 Compensation subject to income tax withholding at the source, with all pre-tax contributions added in.
- c. Code §6041/6051 Income reportable on Form W-2.
- d. Code §6041/6051 Income reportable on Form W-2, with all pre-tax contributions added.
- e. Code §415 All income received for services performed for the Employer.
- 2. Compensation Computation Period:
 - a. Compensation paid during a Plan Year while a Participant.
 - b. Compensation paid during entire Plan Year.
 - c. Compensation paid during the Employer's fiscal year.
 - d. Compensation paid during the Calendar year.
- 3. Exclusions from Compensation:
 - a. No exclusions from Compensation under the Plan.
 - b. Any amount deferred under a salary reduction agreement and which is not includible in the gross income of a Participant under Code Section 125 in connection with a cafeteria plan, Code Section 402(e)(3) in connection with a cash or deferred plan, Code Section 402(h)(1)(B) in connection with a simplified employee pension plan, and Code Section 403(b) in connection with a tax-sheltered annuity plan will be excluded from the definition of Compensation under the Plan.
 - c. Compensation for purposes of the Plan shall not be greater than \$160,000 (as indexed) or as elected above.

- d. Overtime
- e. Cash Bonuses
- f. Commissions
- g. Exclusion applies only to Highly Compensated Employees
- h. Other:

B. "Disability"

- [X] 1. As defined in paragraph 1.23 of the Plan Document.
- [] 2. Disability will be defined as:
- C. "Entry Date"

Contribution Type	Entry Date
Elective Deferrals	с
Employer Match	C
Discretionary Contributions	С
QNECs	C

Entry Date Options:

- a. The first day of the month coinciding with or next following the date on which an Employee meets the eligibility requirements.
- b. The first day of the payroll period coinciding with or next following the date on which an Employee meets the eligibility requirements.
- c. The first day of the Plan Year, or the first day of the fourth, seventh or tenth month of the Plan Year coinciding with or next following the date on which an Employee meets the eligibility requirements.
- d. The earlier of the first day of the Plan Year or the first day of the seventh month of the Plan Year coinciding with or next following the date on which an Employee meets the eligibility requirements.
- e. The first day of the Plan Year following the date on which the Employee meets the eligibility requirements. If this election is made, the Service waiting period cannot be greater than one-half year and the minimum age requirement may not be greater than age 20%.
- f. The first day of the Plan Year nearest the date on which an Employee meets the eligibility requirements. This option can only be selected for Employer related contributions.
- g. The first day of the Plan Year during which the Employee meets the eligibility requirements. This option can only be selected for Employer related contributions.
- h. The Employee's date of hire.
- D. "Highly Compensated Employees Top-Paid Group Election"

If the following election is made, such election shall apply to all Plans maintained by the Employer for Plan Years beginning in 1998 and thereafter.

- [] Employees must be among the Top-Paid Group and have earned more than \$80,000, as indexed.
 - [] Election is applicable for the 1997 Plan Year.
 - [] Election is applicable for 1998 and subsequent Plan Years.

E. "Limitation Year"

The Limitation Year shall be the Plan Year.

F. "Net Profit"

E

ŧ

Not applicable. Employer Contributions to the Plan are not conditioned on profits.

G. "Plan Year"

The 12-consecutive month period commencing on January 1 and ending on December 31.

- H. "QDRO Payment Date"
 - [X] 1. The Plan's earliest QDRO payment date with regard to the Participant against whom the order is entered is the date the QDRO is determined to be qualified.
 - [] 2. The statutory age 50 requirement applies for purposes of making distribution to an alternate payee under the provisions of a QDRO.
- I. "Qualified Joint and Survivor Annuity"
 - [X] 1. Not applicable. The safe-harbor provisions of paragraph 8.7 of the Plan Document are applicable. The normal form of payment is a lump sum.
 - [] 2. The Joint and Survivor Annuity rules are applicable and the survivor annuity shall be ____% (50%, 66-2/3%, 75% or 100%) of the annuity payable during the lives of the Participant and his or her Spouse. If no answer is specified, 50% shall be used.
- J. "Year of Service"

. .

. .

For eligibility, allocation and vesting purposes, a Year of Service will be calculated by:

- [] 1. Elapsed Time Method.
- [X] 2. Hours of Service Method. A Year of Service will be credited upon completion of (choose number of hours for each item below):

а.	Eligibility	1,000 hours
Ъ.	Allocation of Employer Matching Contributions (including QMACs)	1,000 hours
c.	Allocation of QNEC Contributions	1,000 hours
d.	Allocation of Employer discretionary contributions	1,000 hours
e.	Vesting	1,000 hours

Hours shall be determined on the basis of the method selected below. Only one method may be selected. The method selected shall be applied to all Employees covered under the Plan as follows:

- [X] i. On the basis of actual hours for which an Employee is paid or entitled to payment.
- [] ii. On the basis of days worked. An Employee shall be credited with ten (10) Hours of Service if such Employee would be credited with at least one (1) Hour of Service during the day.
- [] iii. On the basis of weeks worked. An Employee shall be credited with forty-five (45) Hours of Service if such Employee would be credited with at least one (1) Hour of Service during the week.

- [] iv. On the basis of semi-monthly payroll periods. An Employee shall be credited with ninety-five (95) Hours of Service if such Employee would be credited with at least one (1) Hour of Service during the semi-monthly payroll period.
- [] v. On the basis of months worked. An Employee shall be credited with one-hundred-ninety (190) Hours of Service if such Employee would be credited with at least one (1) Hour of Service during the month.

K. "Valuation of Trust Fund"

£

The Trust Fund shall be valued on the following schedule:

[X]	a.	Daily	[] d.	Semi-Annually
ĺĺ	ь.	Monthly	[] e.	
[]	c.	Quarterly	[] . f.	Other:

IV. ELIGIBILITY REQUIREMENTS

Complete the following charts using the eligibility requirements outlined below.

A. New Employees:

Contribution Type	Minimum Age	Service Requirement	Class Exclusions
All Contributions	none	3 months	a,b,f
Elective Deferrals & Top-Heavy	[
Voluntary After-Tax			
Required Contributions	1		-
Employer Match			g
QNECs & QMACs			g
Other Employer Contributions			g

Class Exclusions:

- a. Employees included in a unit of Employees covered by a collective bargaining agreement between the Employer and Employee Representatives, if benefits were the subject of good faith bargaining and if two percent or less of the Employees who are covered pursuant to that agreement are professionals as defined in Section 1.410(b)-9 of the Internal Revenue Regulations. For this purpose, the term "Employee Representatives" does not include any organization more than half of whose members are Employees who are owners, officers, or executives of the Employer.
- b. Employees who are non-resident aliens [within the meaning of Code Section 7701(b)(1)(B)] who receive no Earned Income [within the meaning of Code Section 911(d)(2)] from the Employer which constitutes income from sources within the United States [within the meaning of Code Section 861(a)(3)].
- c. Employees compensated on an hourly basis.
- d. Employees compensated on a salaried basis.
- e. Commissioned salesmen.
- f. The Plan shall exclude from participation any nondiscriminatory classification of Employees determined as follows:

Effective January 1, 1995, any individual who makes a written one-time irrevocable election (within the meaning of Section 401(k) and 401(m) and the regulations thereunder) not to participate in the Plan upon his or her commencement of employment with the Employer or upon his or her first becoming eligible under any employee benefit or fringe benefit plan of the Employer.

g. The Plan shall exclude from participation any nondiscriminatory classification of Employees determined as follows:

Effective October 1, 1997, Employee Contractors will be permitted to participate in the Plan, subject to the Plan Eligibility and Service requirements, solely for purposes of contributing Elective Deferrals to the Plan. No Matching Contributions shall be made on behalf of such Employee Contractors.

V. **RETIREMENT AGE**

£

- A. Normal Retirement Age:
 - [X] 1. Normal Retirement Age shall be 65.
 - [] 2. Normal Retirement Age shall be the later of attaining age ____ (not to exceed age 65) or the ____ (not to exceed the 5th) anniversary of the first day of the first Plan Year in which the Participant commenced participation in the Plan.
- B. Early Retirement Age:
 - [] 1. Not Applicable.
 - [X] 2. The Plan shall have an Early Retirement Age of 55.

VI. EMPLOYEE CONTRIBUTIONS

[X] A. Elective Deferrals:

Participants shall be permitted to make Elective Deferrals in whole percentages from a minimum of 1% to a maximum of 15% of their Compensation.

[] B. Automatic Enrollment:

Participants shall automatically, upon first becoming eligible to participate in the Plan, have elective Deferral withheld from their pay in the amount of _____% of Compensation. Participants shall have the right, on proper and timely notice to the Employer, to terminate or amend this Elective Deferral percentage.

[] C. ·· After-tax Voluntary Contributions:

Participants shall be permitted to make after-tax Voluntary Contributions in any amount from a minimum of ____% to a maximum of ____% of their Compensation or a flat dollar amount from a minimum of \$_____ to a maximum of \$_____.

[X] D. Rollover Contributions:

. .

- [X] Participants may make Rollover Contributions.
- [X] Employees may make Rollover Contributions prior to meeting the eligibility requirements for participation in the Plan.

- [] E.. Transfer Contributions:
 - [] Participants may make Transfer Contributions.
 - [] Employees may make Transfer Contributions prior to meeting the eligibility requirements for participation in the Plan.

VII. EMPLOYER CONTRIBUTIONS

The Employer shall make contributions to the Plan in accordance with the formula or formulas selected below. The Employer's contribution shall be subject to the limitations contained in Articles III and X. For this purpose, a contribution for a Plan Year shall be limited for the Limitation Year which ends with or within such Plan Year.

[X] A. Matching Employer Contribution Formulas:

1. For employees listed in Schedule A, the following Employer Matching Contribution Formula will be in effect:

For Employees with 9 Years of Service or less with Chesapeake Utilities Corporation, the Employer shall contribute and allocate to each participant's account, an amount equal to 60% of the first 6% of the Participant's Compensation, to the extent deferred.

For Employees with 10 or more Years of Service with Chesapeake Utilities Corporation, the Employer shall contribute and allocate to each participant's account an amount equal to 100% of the first 6% of the Participant's Compensation, to the extent deferred.

2. For all other Employees, the following Matching Contribution Formula will be in effect:

Years of Service + Age	Matching Percentage
Less than 40	100%
40 to less than 65	150%
65 to less than 80	175%
80 or more	200%

The Employer Matching Percentage will be based on the Participant's Elective Deferrals up to 6% of Compensation. The employer Match up to 100% will be invested exclusively in employer stock. The Employer Match of any amount exceeding 100% may be directed by the Participant in any of the then available investment options offered by the Trustee.

An Employee's Years of Service with the Employer and an Employee's Age will be calculated on the first day of each calendar quarter effective with the Plan year beginning January 1, 1999.

An Employee's Years of Service plus his or her age shall determine the Employer Matching Contribution Percentage in effect for the Employee for the Plan Year. The Employer Matching Contribution percentage will be re-determined for each eligible Employee on the first day of each calendar quarter.

2. Matching Contribution Computation Period: Matching Contributions will be calculated on the following basis:

	Weekly	. Quarterly	
	Bi-weekly	Semi-annually	
	Semi-monthly	Annually	
·	Monthly	X Payroll Based	

The calculation of Matching Contributions based on the Computation Period selected above has no applicability as to when Employer Matching Contributions are remitted by the Employer to the Trust Fund.

7

B. Qualified Non-Elective Employer Contribution Formulas:

- [X] 1. Discretionary Percentage Contribution Formula: The Employer shall have the right to make an additional discretionary contribution which shall be allocated to each eligible Employee in proportion to his or her Compensation as a percentage of the Compensation of all eligible Participants. This part of the Employer's contribution and the allocation thereof shall be unrelated to any other Employer contribution made hereunder and shall be fully vested. This contribution will be allocated to:
 - [] a. All Participants.
 - [X] b. Only Participants who are Non-Highly Compensated Employees.
- [X] C.. Minimum Employer Contribution Formula under Top-Heavy Plans: For any Plan Year during which the Plan is Top-Heavy, the sum of the contributions and forfeitures (excluding Elective Deferrals) allocated to non-Key Employees shall not be less than the amount required under paragraph 15.2 of the Basic Plan Document #01. Top-Heavy minimums will be allocated to:

I. all eligible Participants.] 1.	all eligible Participants.
-------------------------------	------	----------------------------

[X] 2. only eligible non-Key Employees who are Participants.

IX. ALLOCATIONS TO PARTICIPANTS

[X] A.. Allocation Dates:

Contributions listed below will be allocated to Participants as of the Valuation Date and as of the following additional Allocation Date(s) [indicate Allocation Date(s) from the list below to be used for each contribution type]:

1.	Daily	б.	Semi-Annually
2.	Weekly	7.	Annually
3.	Semi-Monthly	8.	Payroll Based
4.	Monthly	9.	Other:
5.	Quarterly		· · · ·

Type of Contribution(s)

Allocation Date

1

All Contributions

[] B. Employer contributions for a Plan Year will be allocated to all Participants employed by the Employer as of the last day of the Plan Year and who have completed a Year of Service. Contributions will also be allocated to Participants who have terminated employment during the Plan Year who satisfy the following eligibility requirements (check all applicable boxes):

			Match	<u>ONEC</u>	<u>OMAC</u>	Other
1.	Comple	eted a Year of Service	[]	[]	[]	[]
2.	No Yea	r of Service will be required for En	ployees who	terminate	due to:	
	a.	Retirement	[X]	[]	[]	[]
	b.	Disability	[X]	[]	[]	[]
	c.	Death	[X]	[]	[]	[]
	d.	Other Termination	[X]	[]	[]	[]

[X]

[] C. Leased Employees:

If a Leased Employee of the Employer is a Participant in the Plan and also participates in a plan maintained by the leasing organization (select one):

- [] 1. The Plan Administrator will determined the Leased Employee's allocation of Employer contributions without taking into account the Leased Employee's allocation, if any, under the leasing organization's plan.
- [] 2. The Plan Administrator will reduce a Leased Employee's allocation of Employer Non-Elective contribution (other than designated Qualified Non-Elective Contributions) under this Plan by the Leased Employee's allocation under the leasing organization's plan, but only to the extent that allocation is attributable to the Leased Employee's Service provided to the Employer.
 - a. The leasing organization's plan must be a money purchase plan which would satisfy the definition of a safe-harbor plan defined at 2.6 in the Basic Plan Document #01, irrespective of whether the safe harbor exception applies.
 - b. The leasing organization's plan must satisfy the features and, if a defined benefit plan, the method of reduction described in an addendum to this Adoption Agreement.

X. **DISPOSITION OF FORFEITURES**

A. Allocation Alternatives:

Disposition Method		Employer Contribution Type				
		<u>Match</u>	Other Contributions	Excess Aggregate		
1.	Reduce next Matching Contribution	[]	[]	N/A		
2.	Offset Plan expenses then reduce next Matching Contribution	[]	[]	[]		
3.	Allocate to all Participants as an additional Match	[]	[]	N/A		
4.	Allocate to Participants who are NHCEs as an additional Match	[]	[]	[]		
5.	Allocate to all Participants in in proportion to Compensation	[]	[]	N/A		
6.	Allocate to Participants who are NHCEs in proportion to Compensation	[]	[]	[]		
7.	Reduce future Employer contributions	[X]	[X]	[X]		
8.	Offset Plan expenses then reduce future Employer contributions	[]	[]	[]		

B. Eligibility for Allocation if Disposition Methods in X(A)(3)-(6) are Elected:

Participants eligible to share in the allocation of other Employer contributions under Section VIII shall be eligible to share in the allocation of forfeitures except where allocations are only to Non-Highly Compensated Employees.

- C. Date for Reallocation of Forfeitures:
 - 1. If no distribution or deemed distribution has been made to a former Participant, nonvested portions shall be forfeited at the end of the Plan Year during which the former Participant incurs his or her fifth consecutive one-year Break in Service.
 - 2. If a former Participant has received the full amount of his or her vested interest, the nonvested portion of his or her account shall be forfeited and shall be disposed of:
 - [] a. at the end of the Plan Year during which the former Participant incurs his or her fifth consecutive one-year Break in Service.
 - [] b. as of the next Allocation Date following the date on which the former Participant receives payment of his or her vested benefit.
 - [X] c. at the end of the Plan Year during which the former Employee incurs his or her 1st one-year Break in Service.
 - [] d. as of the end of the Plan Year during which the former Participant received full payment of his or her vested benefit.
 - [] e. as of the Plan quarter in which the former Participant has received full payment of his or her vested benefit.
- D. Restoration of Forfeitures Upon Rehire:

If amounts are forfeited prior to five consecutive one-year Breaks in Service, the Funds for restoration of account balances to Participants will be obtained from the following sources in the order indicated (number each item accordingly):

- <u>1</u> Current year's forfeitures.
- _2____ Additional Employer contribution.
- Earnings on investments for the applicable Plan Year (not available in a Daily Valuation Plan).

XI. BONUS OPTION

[]

A. If cash bonuses paid by the Employer <u>are</u> included in the definition of Compensation, the Employer may permit a Participant to amend their deferral election, on a one-time basis, to defer to the Plan, an amount not to exceed ___% or \$_____ of any bonus received by the Participant for any Plan Year.

NOTE: If this option is not elected, the Participant's normal deferral election percentage will be automatically withheld from the bonus.

[] B. If cash bonuses paid by the Employer <u>are not</u> included in the definition of Compensation, the Employer may permit a Participant, by completing an appropriate form, to elect to defer from any bonus received by the Participant an amount not to exceed <u>%</u> or <u>%</u> of any bonus received for any Plan Year.

NOTE: If this option is not elected, the Participant will be unable to defer any percentage from the bonus and the entire bonus will be paid to the Participant in cash.

XII. LIMITATIONS ON ALLOCATIONS AND TOP-HEAVY CONTRIBUTIONS

[] A. This is the only Plan the Employer maintains or ever maintained.

Allocation of Excess Annual Additions: In the event that the allocation formula results in an Excess Amount, such excess, after distribution of Employee related contributions pursuant to paragraph 10.2 of the Basic Plan Document #01 shall be:

- [] 1. Placed in a suspense account for the benefit of the Participant without the crediting of gains or losses for the benefit of the Participant (cannot be used in a Daily Valuation Plan).
- [] 2. Reallocated as additional Employer contributions to all other Participants to the extent that they do not have any Excess Amount.

If no method is selected, the suspense account method will be used.

- B. The Employer does maintain or has maintained another Plan [including a Welfare Benefit Fund or an individual medical account as defined in Code §415(1)(2)], under which amounts are treated as Annual Additions and has completed the proper sections below.
 - 1. If the Participant is covered under another qualified Defined Contribution Plan maintained by the Employer, other than a Master or Prototype Plan:
 - [X] a. the provisions of Article X of the Plan Document will apply, as if the other plan were a Master or Prototype Plan.
 - [] b. The Employer has attached provisions stating the method under which the plans will limit total Annual Additions to the Maximum Permissible Amount, and will properly reduce any Excess Amounts, in a manner that precludes Employer discretion.
 - 2. Allocation of Excess Annual Additions:

In the event that the allocation formula results in an Excess Amount, such excess, after distribution of Employee related contributions pursuant to paragraph 10.2 of the Plan Document shall be:

- [X] a. Placed in a suspense account for the benefit of the Participant without the crediting of gains or losses for the benefit of the Participant.
- [] b. Reallocated as additional Employer contributions to all other Participants to the extent that they do not have any Excess Amount.
- 3. If a Participant is or ever has been a participant in a Defined Benefit Plan maintained by the Employer, the Employer must attach provisions which will satisfy the 1.0 limitation of Code §415(e). Such language must preclude Employer discretion and is applicable for Limitation Years beginning before January 1, 2000.
- C. Top-Heavy Plans:

In the event the Plan is or becomes Top-Heavy, the minimum contribution or benefit required under Code Section 416 relating to Top-Heavy Plans shall be satisfied in the elected manner:

- [] 1. The Employer does maintain another Defined Contribution Plan. The minimum contribution will be satisfied by:
 - [] a. this Plan.

b.

[]

(Name of other Qualified Plan)

[X]

. 1

[] 2. The Employer's other Qualified Plan is a Defined Benefit Plan. Provisions are attached to this Plan stating the method under which the minimum contribution and benefit provisions of Code §416 will be satisfied. Such language will preclude Employer discretion and is applicable for Limitation Years beginning before January 1, 2000. If a Defined Benefit Plan is or was maintained, an attachment must be provided showing interest and mortality assumptions used in the Top-Heavy Ratio.

XIII. ANTIDISCRIMINATION TESTING

[X] A. Prior Year Computation Period:

If the Plan Year is not the calendar year, the prior year computation period for purposes of determining if an Employee earned more than \$80,000, as indexed, is the calendar year beginning in the prior Plan Year.

- [] 1. Election is applicable for the 1997 Plan Year.
- [X] 2. Election is applicable for 1998 and subsequent Plan Years.

Note: If the election above is made, such election will apply to all Plans maintained by the Employer for Plan Years beginning in 1998 and thereafter.

- B. Testing Elections:
 - 1. ADP Testing (select one):
 - [X] a. Current year data for all Participants will be used.
 - [] b. Prior year data for Non-Highly Compensated Employees will be used.
 - 2. ACP Testing (select one):
 - [X] a. Current year data for all Participants will be used.
 - [] b. Prior year data for Non-Highly Compensated Employees will be used.
- C. Special Testing Elections For the First Plan Year:
 - 1. ADP Testing (select one):
 - [] a. Current year data for all Participants will be used.
 - [] b. Current year data for Highly Compensated Employees will be used. The ADP for Non-Highly Compensated Employees is assumed to be 3% or the actual ADP if greater.
 - 2. ACP Testing (select one):
 - [] a. Current year data for all Participants will be used.
 - [] b. Current year data for Highly Compensated Employees will be used. The ACP for Non-Highly Compensated Employees is assumed to be 3% or the actual ACP if greater.
- [] D. Elective Deferrals may be recharacterized as Voluntary After-tax Contributions to satisfy the ADP test. Voluntary After-tax Contributions must be permitted in the Plan, for this section to be operable.

XIV. VESTING

. -

Employees shall always have a fully vested and nonforfeitable interest in any Employee (including Elective Deferrals) contributions and their investment earnings. Any Employer contributions made pursuant to the Safe-Harbor Plan provisions must be fully vested when made.

Each Participant shall acquire a vested and nonforfeitable percentage in his or her account balance attributable to Employer contributions and their earnings under the schedules selected below except in any Plan Year during which the Plan is determined to be Top-Heavy. In any Plan Year in which the Plan is Top-Heavy, Two-twenty vesting schedule [option (B)(4)] or the three-year cliff schedule [option (B)(3)] shall automatically apply unless the Employer has elected a faster vesting schedule. If the Plan is switched to option (B)(4), because of its Top-Heavy status, that vesting schedule will remain in effect even if the Plan later becomes non-Top-Heavy until the Employer executes an amendment of this Adoption Agreement.

A. Vesting Computation Period:

1 1 -

The computation period for purposes of determining Years of Service and Breaks in Service for purposes of computing a Participant's nonforfeitable right to his or her account balance derived from Employer contributions:

- [] 1. shall not be applicable since Participants are always fully vested,
- [X] 2. shall commence on the date on which an Employee first performs an Hour of Service for the Employer and each subsequent 12-consecutive month period shall commence on the anniversary thereof, or
- [] 3. shall commence on the first day of the Plan Year during which an Employee first performs an Hour of Service for the Employer and each subsequent 12-consecutive month period shall commence on the anniversary thereof.

A Participant shall receive credit for a Year of Service if he or she completes at least 1,000 Hours of Service, or if lesser, the number of hours specified in Section III(M) of this Adoption Agreement, at any time during the 12-consecutive month computation period. A Year of Service may be earned prior to the end of the 12-consecutive month computation period and the Participant need not be employed at the end of the 12-consecutive month computation period to receive credit for a Year of Service.

B. Vesting Schedules:

. •

Years of Service							
	1	2	3	4	_5	6	
1.	Full and	d immedi	iate Vesti	ng			
2.	%	100%					
3.	_%	%	100%				
4	%	20%	40%	60%	80%	100%	
5.	%	%	20%	40%	60%	80%	100%
6.	10%	20%	30%	40%	60%	80%	100%
7.	_0_%	<u> </u>	_0_%	0%	100%		
8.	_%	%	%	%	%	%	100%

Vesting Schedule Chart

Employer Contribution Type

	Match on Elective Deferrals Match on Voluntary Contributions
	Match on Required Employee Contributions Non-Integrated Contribution Formula
	Integrated Contribution Formula Alternative Integrated Contribution Formula
	Top-Heavy Minimum Contributions
7	All Employer Contributions

C. Service Disregarded for Vesting:

1- F - ~

- [X] 1. Not Applicable. All Service is considered.
- [] 2. Service prior to the [] Original Effective Date [] Amended Effective Date of this Plan or a predecessor plan is disregarded when computing a Participant's vested and nonforfeitable interest.
- [] 3. Service prior to a Participant having attained age 18 is disregarded when computing a Participant's vested and nonforfeitable interest.

XV. SERVICE WITH PREDECESSOR ORGANIZATION

Service with the following organizations will be considered for the Plan purpose indicated:

	<u>Eligibility</u>	Vesting
See Attached Schedule B	[]	[]
	[]	[]

XVI. IN-SERVICE WITHDRAWALS

[]	А.	Elective options)		wals are permitted in the Plan as follows (select one or more of the following
		[]	1.	Participants having completed five or more Years of Service and who are fully vested may elect to withdraw all or any part of their account balance.
		[]	2.	Participants having completed five or more Years of Service may elect to withdraw all or any part of their Vested Account Balance.
		[]	3.	Participants who are fully vested may elect to withdraw all or any part of their account balance which has been credited to their account for a period in excess of two years.
		[]	4.	Participants may elect to withdraw all or any part of their Vested Account Balance which has been credited to their account for a period in excess of two years.
		[]	5.	Participants who are fully vested may elect to withdraw all or any part of their account balance after having attained age (not less than 59½).
		[]	6.	Participants may elect to withdraw all or any part of their Vested Account Balance after having attained age (not less than 59 ¹ / ₂).
. •	- ·	[]	7.	Participants may withdraw all or any part of their account balance after having attained the Plan's Normal Retirement Age.
[X]	B.	Hardshij	o withdra	wals are permitted in the Plan (select one or more of these options):
		[]	1.	Participants may withdraw Elective deferrals only.
		[]	2.	Participants may withdraw fully vested Employer contributions plus their earnings.
		[X]	3.	Participants may withdraw Elective deferrals and fully vested Employer contributions (and earnings on the Employer contributions).
		[]	4.	Qualified Matching Contributions, Qualified Non-Elective Contributions plus

4. Qualified Matching Contributions, Qualified Non-Elective Contributions plus their investment earnings plus the earnings on elective deferrals to the extent that they were credited to the Participant's account as of the last day of the Plan Year ending prior to July 1, 1989 may be withdrawn.

XVII. ANCILLARY BENEFITS

- [X] A. Participant loans as provided for in paragraph 14.5 of the Basic Plan Document #01 are permitted in accordance with the Employer's established loan procedures. Loan payments [X] will [] will not be suspended under the Plan as permitted under Code Section 414(u) in compliance with the Uniformed Service Employment and Reemployment Rights Act of 1994.
- [] B. The insurance provisions of paragraph 14.6 of the Basic Plan Document #01 are applicable.

XVIII. INVESTMENT DIRECTION

A. Investment Management Responsibility:

The Employer shall appoint the Trustee to manage the assets of the Plan unless otherwise indicated below. By selecting a box, the Employer is making a designation as to whom will have investment authority over that specified contribution type.

T----

4

B. Type of Contribution (check all applicable boxes):

		Trustee	Employer	Participant	Investment <u>Manager</u>
1,	All contributions	[]	[]	[]	[]
2.	Elective Deferrals	[]	[]	[X]	[]
3,	Voluntary Contributions	[]	[]	[]	[]
4.	Required Contributions	[]	[]	[]	[]
5.	QMACs	[]	[]	[]	[]
6.	Matching Contributions	[]	[X]	[]	[]
7.	QNECs	[]	[]	[]	[]
8.	Other Employer contributions	s []	[]	[]	[]
9.	Rollover Contributions	[]	[]	[X]	[]
10.	Transfer Contributions	[]	[]	[]	[]

- C. Limitations on Participant Directed Investments:
- [X] 1. Participants are only permitted to select from among investment alternatives made available by the Employer under the Plan.
- [] 2. Participants are permitted to invest in any investment permitted at paragraph 14.3 of the . . . Plan Document.
- [X] D. ERISA Section 404(c):

The Employer has elected to be covered by the fiduciary liability provisions with respect to Participant directed investments under ERISA Section 404(c).

XIX. DISTRIBUTION OPTIONS

- A. Timing of Distributions
 - 1. Distributions payable as a result of termination for death, Disability or retirement shall be paid as soon as administratively feasible, following the date on which a distribution is requested or is otherwise payable.
 - 2. Distributions payable as a result of termination for reasons other than death, Disability or retirement shall be paid as soon as administratively feasible, following the date on which a distribution is requested or is otherwise payable.
- B. Required Beginning Date:
 - 1. The Required Beginning Date of a Participant with respect to a Plan is (select one from below):
 - [] a. The April 1st of the calendar year following the calendar year in which the Participant attains age 70½.
 - [] b. The April 1st of the calendar year following the calendar year in which the Participant attains age 70½ except that distributions to a Participant (other than a 5% owner) with respect to benefits accrued after the later of the adoption or Effective Date of the amendment of the Plan must commence by the later of April 1st of the calendar year following the calendar year in which the Participant attains age 70½ or retires.
 - [X] c. The later of the April 1st of the calendar year following the calendar year in which the Participant attains age 70½ or retires except that distributions to a 5% owner must commence by the April 1st following the calendar year in which the Participant attains age 70½.
 - 2. Alternatives Available for Participants who have attained age 70¹/₂:

Select (a), (b), and/or (c) whichever is applicable. Subsection (c) must be selected to the extent that there would otherwise be an elimination of a pre-retirement age 70½ distribution option for Employees other than those listed above.

- a. Any Participant attaining age 70½ in years after 1995 may elect by April 1st of the calendar year following the year in which the Participant attained age70½ (or by December 31, 1997 in the case of a Participant attaining age 70½ in 1996) to defer distributions until the calendar year in which the Participant retires. If no such election is made, the Participant will begin receiving distributions by the April 1st of the calendar year following the year in which the Participant attained age 70½ or by December 31, 1997 in the case of a Participant attaining age 70½ in 1996).
- [] b. Any Participant attaining age70½ in the year prior to 1997 may elect to stop distributions and recommence by the April 1ⁿ of the calendar year in which the Participant retires. There is either *(select one)*:
 - [] i. a new Annuity Starting Date upon recommencement, or

[] ii. no Annuity Starting Date upon recommencement.

[X]

c.

The pre-retirement distribution option is only eliminated with respect to Employees who reach age 70½ in or after a calendar year that begins after the later of December 31, 1998, or the adoption of the amendment. The pre-retirement age 70½ distribution option is an optional form of benefit under which benefits are payable in a particular distribution form (including any modifications that may be elected after benefit commencement) commence at a time during the period that begins on or after January 1 of the calendar year in which an Employee attains age 70½ and ends April 1st of the immediately following calendar year.

- C. Forms of Payment (select all that apply):
 - Lump Sum [X] 1.
 - **Installment Payments** [] 2.
 - Life Annuity. This option is not available under a safe-harbor Plan described at [] 3. paragraph 8.7 of the Basic Plan Document #01.
 - Term Certain Annuity with payments guaranteed for ____ years (not to exceed [] 4. 20). This option is not available under a safe-harbor Plan described at paragraph 8.7 of the Basic Plan Document #01.
 - Joint and [] 50%, [] 66-2/3%, [] 75% or [] 100% survivor annuity. This option is not available under a safe-harbor Plan described at paragraph 8.7 of the [] 5. Basic Plan Document #01.
 - 6. Other: []

[] D. Recalculation of Life Expectancy:

> When determining installment payments in satisfying the minimum distribution requirements under the Plan, Participants and/or their Spouse (surviving Spouse) shall have the right to have their life expectancy recalculated annually. If recalculation is elected:

- only the Participant shall be recalculated. [] 1.
- 2. both the Participant and Spouse shall be recalculated. []
- the Participant will determine who is recalculated. [X] 3.

XX. SIGNATURES

1 1

The Sponsor recommends that the Employer consult with legal counsel and/or tax advisor before executing this Adoption Agreement.

This Agreement and the corresponding provisions of the Plan Document were adopted by the Employer the _____ day of _ _____, _____,

Executed by the Employer:

William C. Boyles

Title:

Vice President

Signature: . .

TRUSTEE: B.

> **PNC Bank**, Delaware **PO Box 791** Wilmington, DE 19899-0791

The Employer's Plan as contained herein was accepted by the Trustee the ____ day of

Accepted on behalf of the Trustee by:

Title:

Signature:

SCHEDULE A

Employees listed in this Schedule A are those Employees who elected retirement option #2 ("Retirement Option 2") under the new retirement program adopted by the Employer, to be effective January 1, 1999 (the "new Retirement Program").

•

. .

. .

. .

SCHEDULE B

<u>Eligibility</u>	Vesting	
 []	[]	

Service with the following organizations will be considered for the Plan purpose indicated:

•

 $c \in c$

٠

.